# THE EFFECT OF HYPOTHYROIDISM ON HEPATIC ADRENERGIC RECEPTORS IN THE RAT

by

**c**) Harold G. Preiksaitis

Ø

A thesis submitted to the

Faculty of Graduate Studies and Researchin partial fulfillment of the requirements

for the degree of

Doctor of Philosophy.

Department of Pharmacology and Therapeutics

McGill University

Montreal, Canada

March, 1983

i

53

٩.

1,1

Some of the work of this thesis has been published.

H.G. Preiksaitis and G. Kunos (1979) Adrenoceptor-mediated activation of liver glycogen phosphorylase: effects of thyroid state. Life Sci 24:35-42.

H.G. Preiksaitis, W.H. Kan and G. Kunos (1982) Decreased  $\alpha_1$ -adrenoceptor responsiveness and density in liver cells of thyroidectomized rats. J Biol Chem 257:4321-4327.

ii

#### ABSTRA.CT

Hepatic adrenoceptors were examined in different thyroid states. Activation of glycogen phosphorylase in hepatocytes from euthyroid and hyperthyroid rats was mediated by alpha adrenoceptors. In hypothyroid rats there was a significant beta receptor contribution to this response and enhanced beta receptor-mediated accumulation of cAMP. ' Responses to glucagon, a hormone which also activates phosphorylase via cAMP, were not similarly affected. The increase in beta receptor-mediated responses in the hypothyroid rat was accompanied by decreased alpha receptor activity as indicated by a marked supression of phenylephrine-stimulated phosphorylase activation and calcium efflux. Similar effects produced by vasopressin and the calcium ionophore, A23187, were changed only slightly or not at all. Alpha<sub>1</sub> receptors in liver plasma membranes were identified by [<sup>3</sup>H] prazosin binding. Hypothyroidism did not influence the binding affinity of agonists or [<sup>3</sup>H] prazosin, but reduced the density of alpha, receptors. This change was opposite to the increase in density of beta receptors in hypothyroid rat liver reported by others. All the changes observed in hypothyroidism were partially reversed by in vivo thyroid hormone replacement. It is concluded that hypothyroidism produces a selective enhancement of beta and suppression of alpha receptor responses in rat liver. These changes are most likely the result of corresponding alterations in the density of membrane receptor sites.

iii

Ł

RĚSUME

Les récepteurs adrénergiques hépatiques furent examinés sous différents états thyroidiens. L'activation de la phosphorylase du glycogène dans les hépatocytes de rats euthyroidiens et hyperthyroidiens s'effectuait par-devers les récepteurs adrénergiques alpha. Il y avait, chez les rats hypothyroidiens, une participation significative du récepteur bêta a cette réponse. Ce dernier provoquait en outre une accumulation accrue de l'AMP cyclique. Les réponses au glucagon, une hormone qui active également la phosphorylase via l'AMP cyclique, n'étaient pas affectées de façon similaire. L'augmentation des réponses obtenues par l'intermédiaire des récepteurs bêta chez les rats hypothyroidiens était accompagnée d'une diminution de l'activité des récepteurs alpha, telle que démontrée par la suppression marquée de l'activation de la phosphorylase et de la sortie du calcium en réponse a la phényléphrine. Des effets comparables, produits par la vasopressine et l'ionophore calcique A23187, ne furent que légèrement ou nullement changés. Les récepteurs de type alpha, dans les membranes-plasmatiques du foie furent identifiés par la liaison de prazosine tritiée. L'hypothyroidisme n'influencait pas l'affinité de la liaison des agonistes adrénergiques ou de la prazosine tritiee, mais reduisait toutefois la densite des recepteurs de type alpha, Ce changement était en opposition à l'augmentation de la densité des récepteurs bêta au niveau du foie du rat hypothyroidien, telle que notée par d'autres chercheurs. Tous les changements observés lors de Inypothyroidisme étaient partiellement renversés, in vivo, par un apport exogène d'hormone thyroidienne. Il est de ce fait conclu que l'hypothyroidisme produit au niveau du foie une augmentation sélective des réponses reliées a l'activité des récepteurs bêta et une suppression de celles relevant des récepteurs alpha. Ces chahgements sont vraisemblablement le fruit d'une modification au niveau de la densité des sites récepteurs membranaires.

iv

TABLE OF CONTENTS
TABLE OF CONTENTS
LIST OF FIGURES
LIST OF TABLES
ACKNOWLEDGEMENTS
ABBREVIATIONS USED XIV
SECTION ONE: INTRODUCTION
1.1 Historical Perspective and Preamble
1.2 Sympathetic Nervous System-Thyroid Interactions 2
1.2.1 General effects 2
1.2.2 Catecholamine disposition: effects of thyroid state 3
1.2.3 Thyroid effects on end-organ responsiveness to
<b>G</b> catecholamines 5
1.2.4 Thyroid effects on adrenergic radioligand binding 12
1.3 Hepatic Adrenergic Receptors 15
1.3.1 Classification of hepatic adrenergic receptors 15
1.3.2 The mechanism of beta adrenergic activation of hepatic
glycogen phosphorylase 21
1.3.3 The mechanism of alpha adrenergic activation of hepatic
glycogen phosphorylase 27
1.3.4 Identification of hepatic adrenergic receptors by
radioligand binding
1.4 Role of the Autonomic Nervous System in the Control of Hepatic
Metabolism
1.4.1 Anatomical considerations
1.4.2 The effects of splanchnic nerve stimulation
1.4.3 The effects of stimulation of the hepatic parasympathetic
nerves

C

ν

. 1.5 Effects of	Thyroid State on Mepatic Carbohydrate Metabolism	49
1.5.1 Glycog	en levels and phosphorylase activity	49
1.5.2 Hepati	c enzyme activities	5 <b>0</b>
1.5.3 Effect	s on redox state and energy charge	50
1.5.4 Summar	у	51
1.6 Formulatio	n of the Problem	51
SECTION TWO: EXPE	RIMENTAL PROCEDURES	
2.1 Animals	, 	54
2.2 Drugs and	Chemicals	54
2.3 Isolation	of Hepatocytes	55
2.4 Assay of G	lycogen Phosphorylase	56
2.5 Determinat	ion of cAMP	60
2.6 Calcium Ef	flux Electrode Studies	60
2.7 <sup>45</sup> Ca Efflu	x Studies	61
2.8 Thyroid Ho	rmone Assays	62
2.9 Preparation	n of Liver Plasma Membranes	62
2.10 [ <sup>3</sup> H] Prazos:	in Binding Studies	62
2.11 Statistics	and Expression of Results	63
SECTION THREE: RE	SULTS	,
3.1 Activation	of glycogen phosphorylase by adrenergic agonists in	`
hepatocytes	from euthyroid, hypothyroid and hyperthyroid rats	65
3.2 Effects of	adrenergic antagonists on adrenaline activation of	
glycogen	phosphorylase in hepatocytes from euthyroid,	
, hypothyroid	and hyperthyroid rats	70
3.3 The effect	of cortisol-treatment on glycogen phosphorylase	
activation	by adrenaline, isoproterenol and phenylephrine in	
hepatocytes	from thyroidectomized rats	72
3.4 Accumulatio	n of cAMP in response to stimulation by adrenergic	*

vi,

agonists in hepatocytes from normal and thyroidectomized rats .. 74

3.5	Glucagon-induced activation of glycogen phosphorvlase and cAMP	
	accumulation in hepatocytes from thyroidectomized and normal	r.
	rats	82
3.6	Effect of triiodothyronine replacement on isoproterenol-induced	
	CAMP accumulation and glycogen phosphorylase activation in	
	hepatocytes from thyroidectomized rats	85
3.7	Effect of phenylephrine on calcium efflux in hepatocytes from	-
	normal, thyroidectomized and thyroidectomized, T <sub>3</sub> -treated rats.	88
3.8	The effect of vasopressin on calcium efflux from hepatocytes	
e •	of normal and thyroidectomized rats $\overset{\flat}{\ldots}$	93
3.9	Effect of calcium ionophore, A23187, on calcium efflux and	
• •	glycogen phosphorylase activation in hepatocytes from normal	,
	and thyroidectomized rats	95
3.10	Phenylephrine-induced activation of glycogen phosphorylase and	
	<sup>45</sup> Ca efflux in hepatocytes from normal, thyroidectomized and	
	thyroidectomized, triiodothyronine-treated rats	98
3.11	Vasopressin-induced activation of glycogen phosphorylase and	
	<sup>45</sup> Ca efflux in hepatocytes from normal and thyroidectomized	
	rats	99 <sub>,</sub>
3.12	Results of [ <sup>3</sup> H]prazosin binding to plasma membranes isolated	
	from normal, thyroidectomized and thyroidectomized, T3-treated	
	rat livers	103
SECTION	N FOUR: DISCUSSION	
4.1	General Considerations: The effect of altered thyroid state on	
	hepatic adrenergic receptors	112
4.2	The effect of hypothyroidism on hepatic beta adrenergic	
,	receptors. Comparison with glucagon	117
4.3	The effect of hypothyroidism' on hepatic alpha adrenergic	
	receptors. Comparison with vasopressin and A23187	125

C

こ

4	4.4	Similar changes in hepatic adrenoceptors produced by	
		non-thyroidal influences	136
4	4.5	Physiological implications of thyroid-induced changes in	
		hepatic adrenoceptor properties	142
4	4,6	Possible mechanisms of thyroid induced alterations in hepatic	
4		adrenoceptor properties	148
SEC	CTIC	N FIVE: SUMMARY AND CONCLUSIONS	153
REF		NTRS .	156

C

viii

LIST OF FIGURES

	┸╸┶	Calcium- and CAMP-dependent pathways for the activation of liver
		glycogen phosphorylase
	2.1	Time- and protein-dependent increase in [U-14C]glucose
	a.	incorporation into glycogen in the assay of hepatic glycogen
>.		phosphorylase <u>a</u>
	3.1	Time-course of activation of glycogen phosphorylase $\underline{a}$ by
		adrenaline in hepatocytes from euthyroid, hypothyroid and .
	e	hyperthyroid rats
	3.2	Activation of glycogen phosphorylase <u>a</u> by adrenaline,
		isoproterenol and phenylephrine in hepatocytes from euthyroid,
0	J	hypothyroid and hyperthyroid rats
	3.3	Effects of adrenergic antagonists propranolol, phenoxybenzamine
		and dihydroerqocryptine on activation of glycogen phosphorylase
	đ	a by adrenaline in hepatocytes from euthyroid, hypothyroid and
		hyperthyroid rats
	3.4	The effect of cortisol treatment on the activation of glycogen
	•	phosphorylase a by adrenaline, isoproterenol and phenylephrine
		in hepatocytes from hypothyroid rats
•	3.5	Time-course of cAMP accumulation in response to isoproterenol in
r		hepatocytes from normal and thyroidectomized rats
	3.6	Accumulation of cAMP in response to stimulation by adrenaline,
•		isoproterenol and phenylephrine in hepatocytes from normal and
		thyroidectomized rats
	3.7	Accumulation of cAMP accumulation in response to stimulation by
		adrenaline and noradrenaline in hepatocytes from a
,		thyroidectomized rat
	3.8	Isoproterenol-stimulated cAMP accumulation as a function of
		s*

ix

, °e ,	
	tissue amount in hepatocytes from normal and thyroidectomized
ن م	rats
3.9	Activation of glycogen phosphorylase by glucagon in hepatocytes
, ,	from normal and thyroidecomized rats
, ´3, 1	Accumulation of cAMP in response to stimulation by glucagon in
¢	hepatocytes from normal and thyroidectomized rats
3.1	Effect of triiodothyronine replacement on isoproterenol
,	stimulated accumulation of cAMP in hepatocytes from
d a	thyroidectomized rats. Comparison with response in normal and
, , <b>, ,</b>	untreated thyroidectomized rats
3.12	Effect of triiodothyronine replacement on activation of
a - 1 - 1 - 1	phosphorylase by isoproterenol in hepatocytes from
ŭ	thyroidectamized rats. Comparison with response in normal and
	untreated, thyroidectomized rats
3.13	Effects of various glycogenolytic agents on calcium efflux from
	hepatocytes isolated from a normal and thyroidectomized rat as -
0 	measured by a calcium-sensitive electrode 90
3.14	Phenylephrine-stimulated calcium efflux in hepatocytes from
, , , , , , , , , , , , , , , , , , ,	normal, thyroidectomized and thyroidectomized, $T_3$ -treated rats 91
3.15	Vascoreșsin-stimulated calcium efflux in hepatocytes from normal
,	and thyroidectomized rats
3.16	Activation of glycogen phosphorylase by calcium inophore, A23187,
- 0 B	in hepatocytes from normal and thyroidecomized rats
° 3.,17	Stimulation of calcium efflux in hepatocytes from normal and
þ	thyroidectomized rats by calcium ionophore, A23187, 97
, <b>3. 1</b> 8	Effect of extracellular calcium concentration on vasopressin and
ŕ	phenylephrine activation of glycogen phosphorylase in hepatocytes
- a	from a normal rat
3.19	from a normal rat

¢.....

5.00

1

۴,

LIST OF TABLES .

 $\mathbb{C}$ 

(

1.1 Agents that Stimulate Hepatic Glycogenolysis	0
3.1 The influence of thyroid state on the potency of adrenergic	1
agonists in activating hepatic glycogen phosphorylase	9
3.2 Effect of phosphodiesterase inhibitor (IBM) on isoproterenol-	
and glucagon-stimulated cAMP accumulation in hepatocytes from	0
normal and thyroidectomized rats	a O
3.3 Effect of phenoxybenzamine on adrenaline-stimulated cAMP	
accumulation in hepatocytes from a normal and a thyroidectomized	8 0
rat	1
3.4 Potency of phenylephrine and vasopressin in stimulation of	1
3.4 Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from	1
3.4 Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from normal, thyroidectomized and thyroidectomized,	
3.4 Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from normal, thvroidectomized and thyroidectomized, triiodothyronine-treated rats	2
<ul> <li>3.4 Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from normal, thvroidectomized and thyroidectomized, triiodothyronine-treated rats</li></ul>	2
<ul> <li>3.4 Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from normal, thvroidectomized and thyroidectomized, triiodothyronine-treated rats</li></ul>	2

8

8

ŷ

1 (

R

1.

ŝ,

#### ACKNOWLEDGEMENTS

The author wishes to acknowledge the following.

Dr. John Ruedy and Dr. Richard Ogilvie for the opportunity of studying pharmacology at McGill University.

Dr. George Kunos for his enthusiastic and indulgent supervision of me and this work.

Drs. Brian Collier, Alan Tenenhouse and Mark Nickerson for their teaching, advice and encouragement.

Julio (W.H.) Kan and Charles Peterfy for their helpful collaboration in some of the experimental work.

Lucia Fallavollita for her skilled technical assistance.

Royal Raymond for the photography and manufacture of some of the equipment.

Andre Côté and Yves Théoret for translating the abstract into French. Jennifer Kelso and JoAnne Perreault for typing.

The Faculty of Medicine, McGill University for allowing me to continue this work while studying medicine.

The National Research Council of Canada, la Direction Générale du "
I'Enseignment Supérior du Québec and the Canadian Heart Foundation for financial support.

Lastly, Charlie Price, Heimo Scheer, Charles Peterfy, Jane Batten, Peter Grosser, George Colpitts, Tom Dikens, Seana O'Regan, Hermani DeMatos, Kenny Kwok and others who made the too many years it took to prepare this thesis seem too short.

xiii

### ABBREVIATIONS USED

Ð

Ø

C

COMT	Catechol-O-methyl Transferase
MAO	Monoamine oxidase
cAMP, cyclic AMP	Adenosine-3',5'-monophosphoric acid, cyclic
cCMP	Guanosine-3',5'-monophosphoric acid, cyclic
ATP, ADP	Adenosine triphosphate, diphosphate
AMP, 5'-AMP	Adenosine-5'-monophosphate
Tx ,	Thyroidectomized
T <sub>3</sub>	3,3',5'-Triiodothyronine
T <sub>4</sub>	Thyroxine
IBVX	Isobutylmethylxanthine
DHA	Dihydroalprenolol
DHE	Dihydroergocryptine
IHBP	<b>Iodo</b> hydroxybenzylpindolol
POB	Phenoxybenzamine
Gpp(NH)p	Guany 1-5'-y limidodiphosphate
EDTA	Ethylenediaminetetracetate /
BGTA	Ethylene glycol bis( -amino ethyl ether)- N,N,N',N'-tetra acetic acid
NAD, NADH	Nicotinamide adenine dinucleotide (oxidized, reduced)
TRIS	Tris(hydroxymethyl)-aminomethane
B	Maximum binding capacity
· K <sub>d</sub>	Dissociation constant
EC	Concentration of agonist causing half- maximal activation
pD <sub>2</sub>	$-\log_{10}(EC_{50})$
n <sub>H</sub>	Hill coefficienț

## SECTION ONE

C

# INTRODUCTION

Ð

0 --

Ü

#### **1.1** Historical Perspective and Preamble.

Many of the manifestations of thyrotoxicosis in man resemble hyperactivity of the sympathetic nervous system. That some relationship between the thyroid and adrenergic activity exists was first realized in the mid-nineteenth century (Harrison 1964). The earliest concept that alterations in the activity of the cervical sympathetic nerves resulted in enhanced glandular function gave way to the more widely held contemporary opinion that thyroid-induced changes in end-organ responses to sympathetic stimulation were important in the pathogenesis of thyroid disease (Harrison 1964). Indeed the effect of altered thyroid state on adrenergic responsiveness has been extensively studied especially in cardiac and adipose tissue (Kunos 1981). The recent introduction of the use of radiolabelled ligands in the study of adrenergic receptor properties has added an exciting dimension to these investigations (Hoffman and Lefkowitz 1980); it has been shown that, at least in some instances, thyroid-induced alterations in adrenergic responses can be attributed to changes at the level of the catecholamine receptor (Kunos 1981). However, in spite of the many excellent studies of this phenomenon, the underlying mechanism of these changes in adrenoceptor properties remains an enigma.

The concept of multiple types of adrenergic receptors was first proposed by Dale (1906) and later modified in the dual receptor theory of Alquist (1948). Subsequently, most adrenergic responses were classified as being mediated by either alpha or beta receptors. However, the glycogenolytic response of hepatic tissue to catecholamines could not easily be classified as belonging to the alpha or beta group of responses; the reasons for this are manifold (Hornbrook 1970). Recently it has become clear that both alpha and beta receptors can contribute to the glycogenolytic response of the liver. In the heart, where force and rate responses to catecholamines similarly involve both

alpha and beta receptors, hyperthyroidism was shown to increase beta and decrease alpha receptor activity. Therefore, it seemed logical to postulate that hepatic alpha and beta receptors would be modulated by thyroid hormones in a similar manner. This thesis describes the testing of this hypothesis and confirms an effect of thyroid state on hepatic adrenergic receptors, although the observed change is not as predicted.

The introduction which follows briefly reviews thyroid effects on the sympathetic nervous system (section 1.2), hepatic advenergic receptors (section 1.3), the role of the sympathetic nervous system in the control of hepatic metabolism (section 1.4) and the metabolic effects of altered thyroid state with special emphasis on the liver and carbohydrate metabolism (section 1.5). No attempt has been made to comprehensively review here the huge literature that exists in any of these subject areas. Many controversial points arise, however, in the interest of clarity and brevity, only those that are felt to pertain directly to the present work are discussed at any length; most are taken up in Section IV: Discussion.

#### 1.2 Sympathetic Nervous System-Thyroid Interactions.

#### 1.2.1 General Effects

As might be expected for a subject that has been investigated for approximately 100 years, the vast amount of accumulated literature on sympathetic nervous system-thyroid interactions is often contradictory and confusing. Several excellent attempts have been made at organizing this information in recent reviews (Harrison 1964; Waldstein 1966; Leak 1970; Spaulding and Noth 1975; Landsberg 1977; Kunos 1981; Fain 1981).

Many clinical observations suggest that some manifestations of hyperthyroidism are due to over-activity of the sympathetic nervous system. Increased heart rate and contractility, sweating, muscle tremor and weight loss

are seen in patients with thyrotoxicosis. Similar effects can be produced by the administration of sympathomimetic agents to normal subjects and are also seen in patients with pheochromocytoma; indeed, the latter must be considered in the differential diagnosis of hyperthyroidism. Moreover, these manifestations may be ameliorated by the administration of adrenergic blocking agents. However, when sympathetic activity was determined directly by measuring noradrenaline turnover in hearts of hyperthyroid animals, it was shown to be normal or slightly decreased (Landsberg 1977). On the other hand, sympathetic activity is enhanced in hypothyroidism (Landsberg and Axelrod 1968).

Plasma and urinary noradrenaline levels are normal or low, and plasma dopamine-beta-hydroxylase activity is decreased in hyperthyroidism, whereas in hypothyroidism, plasma noradrenaline and dopamine-beta-hydroxylase are increased, and urinary noradrenaline levels are either increased or normal. Neither hypothyroidism nor hyperthyroidism appear to have a significant effect on urinary levels, plasma levels or the secretion of adrenaline. These findings (Landsberg 1977) suggest that sympathetic nervous activity is either decreased or unchanged in hyperthyroidism and increased in hypothyroidism. Neither state appears to produce a significant alteration in secretion of adrenaline by the adrenal medulla.

Therefore, the manifestations of thyrotoxicosis cannot be attributed to overactivity of the sympatho-adrenal axis and one must consider changes occurring at other sites.

#### 1.2.2 Catecholamine Disposition: Effects of Thyroid State.

Alterations in tissue response to catecholamines could result either from changes in tissue sensitivity or from altered disposition of catecholamines causing local increases or decreases at receptor sites. Circulating catecholamines including adrenaline released from the adrenal medulla are destroyed mainly by catechol-O-methyl transferase (COMT) in the liver and

kidney (Mayer 1980). Since gross alterations in circulating adrenaline or adrenaline released from the adrenal medulla are not produced by changes in thyroid hormone levels (see above), it is unlikely that deranged metabolism could account for altered sympathetic activity in the intact animal.

Of perhaps greater significance, particularly with respect to thyroid-induced changes identified in isolated tissues, is the disposition of catecholamines in the vicinity of the adrenergic receptor. Normally, approximately 90% of noradrenaline released into the synapse is rapidly taken up again into the sympathetic nerve terminal by a specific transport system (Mayer 1980). Adrenaline, which finds its way into the synaptic space, is also a substrate for uptake via this mechanism. Some evidence exists to suggest that the activity of this transport system is decreased in hyperthyroidism and unchanged in hypothyroidism (Wurtman et al. 1963; Landsberg and Axelrod 1968). Wurtman et al. (1963) suggested that this could be the basis for apparently increased sympathetic activity in thyrotoxicosis. However, additional factors must be involved since tissue sensitivity to isoproterenol, an adrenergic agonist which is not a substrate for neuronal uptake, still showed thyroid-dependent changes (McNeill and Brody 1968; Kunos 1977). Moreover, inhibition of the extraneuronal uptake of isoproterenol in the heart by the addition of steroids did not modify the reduced sensitivity of the inotropic response in the hypophysectomized rat (Kunos et al. 1980). Thus at least in this example, alterations in extraneuronal uptake of the sympathomimetic could not account for alterations in adrenergic sensitivity.

Uptake into the noradrenergic nerve terminal is followed by storage of catecholamine in vesicles for subsequent re-release and, to a lesser extent, metabolism by monoamine oxidase (MAO), whereas extra-neuronal uptake results in metabolism by COMT (Mayer 1980). Effects of thyroid hormone levels on both MAO and COMT have been reported for many tissues and the findings are summarized in several recent publications (Youdim et al. 1976; Sourkes et al.

1977; Kunos 1981). These enzyme activities vary with species, age, sex and tissue examined. In general, the changes are small and while they may explain some aspects of altered tissue sensitivity to catecholamine, they are unlikely to be of major importance (Kunos 1981).

#### 1.2.3 Thyroid Effects on End-organ Responsiveness to Catecholamines.

Present-day understanding of adrenergic receptors had its foundation in the dual receptor theory of Ahlquist (1948). He proposed that adrenergic receptors be divided into two groups based on the order of potency of agonists for producing the response: isoproterenol > adrenaline > noradrenaline for beta receptor-mediated responses and adrenaline > noradrenaline > isoproterenol for alpha receptor-mediated responses. This theory was later refined to include two subtypes of beta receptors: beta<sub>l</sub> for which adrenaline = noradrenaline and beta<sub>2</sub> for which adrenaline > noradrenaline (Lands et al. 1967). Alpha receptors have also been sub-divided, originally on the basis of anatomical location as pre- and post-synaptic receptors (Langer 1974). The more recent terminology of alpha, and alpha, receptors reflects pharmacological rather than anatomical differences (Berthelson and Pettinger 1977). This modification in theory is justified since both receptor subtypes have been identified by pharmacological and ligand binding studies at post-synaptic sites (Fain and García-Sáinz 1980; Jard et al. 1981). The recent development of highly selective agonists and antagonists has proved useful in the differential study of these receptor types. It now appears clear that these receptors also differ in . the mechanism by which they affect a tissue response. While both beta, and beta $_2$  receptors are coupled to adenylate cyclase and stimulation results in increased tissue cAMP levels (Ross and Gilman 1980; Limbird 1981), alpha receptors mediate their effects through the mobilization of calcium and alpha<sub>2</sub> receptors inhibit adenylate cyclase activity (Fain and García-Sáinz 1980).

This concept of multiple subtypes of adrenergic receptors has been

- 5

emerging during recent years and has proven to be much more complex than one might have anticipated only a few years ago. Therefore, it is easy to appreciate the difficulties encountered in early attempts to understand the mechanism of thyroid effects on adrenergic responses. More recent studies of this problem attempt to analyze thyroid effects within this framework by examining specific receptor types (Kunos 1981; Fain 1981). Although controversies still exist, a clearer picture is emerging. Reasons for apparent discrepancies include sex; age; species differences, both with respect to catecholamine actions and sensitivity to thyroid hormones; the lack of complete dose-response studies and inadequate assessment of sensitivity changes, as well as differences in treatments used to produce hypothyroidism or hyperthyroidism in experimental animals (for a more complete discussion see Kunos 1981). Effects of thyroid hormones on adrenergic responsiveness of cardiac and adipose tissue have been examined extensively and will be briefly summarized here.

#### Cardiac Tissue.

Most studies have shown that hypothyroidism produced an increased alpha and decreased beta receptor contribution to adrenergic stimulation of the rate and force of contraction in isolated cardiac tissues (Nakashima et al. 1973; Kunos 1977; Kunos et al. 1980). The opposite was observed in hyperthyroidism (Kunos 1977; Hashimoto and Nakashima 1978). These results are supported by the finding of enhanced catecholamine-stimulated adenylate cyclase in cardiac preparations from hyperthyroid animals (Tse et al. 1980) and an attenuated activity in preparations from hypothyroid animals (Levey et al. 1969; Kunos et al. 1976; Brodde et al. 1980). Not all studies, however, are in agreement with these findings since some authors have found catecholamine-stimulated adenylate cyclase activity unaltered in hyperthyroid myocardium (McNeill et al. 1969; Young and McNeill 1974). The cardiac glycogenolytic effect of adrenergic agonists was also reported to be enhanced in hyperthyroidism, although these effects were not analyzed in detail with respect to receptor

type (Hornbrook et al. 1965; McNeill and Brody 1968; Hornbrook and Cabral 1972). In all of the above-noted studies thyroid state was altered in the intact animal. Tsai and Chen (1978) have demonstrated a  $T_3$ -mediated enhancement of beta receptor-stimulated cAMP accumulation in cultured fetal heart cells. This finding indicates that thyroid hormone could produce an increase in beta receptor activity in <u>vitro</u>.

Adipose Tissue.

Catecholamine-mediated activation of lipolysis in adipose tissue occurs via activation of the beta receptor/adenylate cyclase system (Himms-Hagen 1972; Young and Landsberg 1977). In addition, adipocytes also possess alpha<sub>2</sub> receptors that mediate inhibition of the adenylate cyclase and alpha<sub>1</sub> receptors that affect changes in glycogen metabolism analogous to those in liver (see section 1.3; García-Sáinz and Fain 1980). While the presence of all three types of adrenergic receptors have been shown in preparations of human and hamster adipose tissue, there is no evidence for alpha<sub>2</sub> receptors in rat fat cells (García-Sáinz and Fain 1980).

Debons and Schwartz (1961) showed that in adipose tissue from the rat the in vitro lipolytic response to catecholamine was enhanced in hyperthyroidism and suppressed in hypothyriodism. The decreased response in hypothyroidism was thought to be due to an attenuated cAMP accumulation during stimulation (Malbon et al. 1978b). Some investigators did show a decrease in adenylate cyclase response to catecholamine in hypothyroidism (Correze et al. 1974; Malbon et al. 1978b), while others found no change compared with euthyroid controls (Armstrong et al. 1974). Fain (1981) attributed these variable results to the use of different preparations and the inability to precisely control experimental conditions. Moreover, on the basis of unchanged beta receptor binding found in some studies (see section 1.2.4), he suggested that the impaired beta receptor responsiveness in hypothyroid adipocytes was due to a defective coupling of hormone receptor complexes to

adenylate cyclase. However, other effects cannot be discounted and probably contribute to the impaired response.

The presence of phosphodiesterase inhibitor potentiated adrenaline-induced lipolysis to a greater extent in adipocytes from hypothyroid than euthyroid rats (Goswami and Rosenberg 1978). Accordingly, others (Armstrong et al. 1974; Van Inwegen et al. 1975; Correze et al. 1976) reported increased phosphodiesterase activities in preparations from hypothyroid rat adipose tissue. These findings suggest that at least a portion of the impaired responsiveness of hypothyroid adipocytes may be due to increased metabolism of cAMP. However, Malbon et al. (1978b) were unable to find any change in the cAMP phosphodiesterase activity of fat cell ghosts. This may have been due to; differences in the preparation used in that study compared with others described above.

Ohisalo and Stouffer (1979) observed that the impaired responsiveness of adipocytes from hypothyroid rats could be reversed by use of more dilute cell suspensions or the inclusion of adenosine deaminase in the incubation medium. This suggested that increased production or sensitivity to adenosine by hypothyroid adipocytes could contribute to the impaired responsiveness. On the other hand, Fain and Malbon (1979) found a decreased release and also impaired oxidative degradation of adenosine in hypothyroid adipocytes which tends to refute the importance of adenosine in this phenomenon.

Catecholamine-stimulated cAMP accumulation and lipolysis were enhanced in fat cells from  $T_3$ -treated rats, although this effect was not reflected in an enhanced adenylate cyclase response (Caldwell and Fain 1971; Malbon et al. 1978b). Thyroid state had an opposite effect on cAMP accumulation in hamster adipocytes. Isoproterenol-stimulated cAMP accumulation was enhanced in hypothyroidism and impaired in hyperthyroidism, the latter being attributed to a toxic effect of high doses of thyroid hormone (Fain 1981). Conversely, Giudicelli et al. (1980) also working with hamster adipocytes, inferred that the

decreased cAMP response in hyperthyroid animals was due to an alteration in alpha receptor activity. Alpha<sub>2</sub> receptor stimulation results in inhibition of adenylate cyclase (Fain and García-Sáinz, 1980) and Rosenqvist et al. (1971) suggested that the impaired lipolytic response in human adipocytes could in part be due to a greater alpha<sub>2</sub> suppression of cAMP accumulation. However, it was later shown that neither hypothyroidism nor hyperthyroidism altered the alpha<sub>2</sub> receptor-medited inhibition of adenylate cyclase in hamster adipocytes (García-Sáinz et al. 1981). As mentioned before, thyroid effects on adrenergic receptors can show marked species variation and hence cannot readily be 'extrapolated. Therefore, the question of thyroid effects on alpha<sub>2</sub> receptors in humans and their role in altered sensitivity of the lipolytic response to catecholamines remains unanswered.

Alpha<sub>1</sub> receptor stimulation increases phosphatidylinositol turnover which is thought to be linked to an elevation of intracellular calcium (Fain and García-Sáinz 1980). Both alpha and beta receptor stimulation can produce activation of glycogen phosphorylase and inactivation of glycogen synthase in adipocytes (Lawrence and Larner 1977) just as in the liver (section 1.3). García-Sáinz and Fain (1980) have reported no change in alpha, receptor-mediated stimulation of phosphatidylinositol turnover or glycogen synthase inhibition in hypothyroid rat fat cells. However, Kunos (1981) has suggested that the lack of change in inactivation of glycogen synthase by the mixed agonist adrenaline in the face of markedly impaired inhibition by the beta agonist isoproterenol in the above study must suggest the presence of an increased alphan receptor activity in the hypothyroid state. This point requires clarification. García-Sáinz et al. (1981) also reported a lack of effect of thyroid hormone on alpha, stimulation of phosphatidylinositol turnover in hamster adipocytes.

#### Hepatic Tissue.

Early studies demonstrated that the glycogenolytic effects of adrenaline

are potentiated in hyperthyroidism (Harrison 1964). In recent years it has been shown that both alpha and beta receptors can, influence hepatic carbohydrate metabolism (see section 1.3). However, the glycogenolytic response in the intact animal is complex (Hornbrook 1970) and interpretation in terms of receptor effects is very difficult. At the outset of the present study there were only a few reports dealing with thyroid hormone effects on hepatic adrenergic responses. Experiments with isolated perfused rat liver were interpreted to suggest that beta receptor-mediated stimulation of gluconeogenesis in the normal rat<sup>v</sup> was suppressed in hypothyroidism, while the contribution of alpha receptors was enhanced, (Hagino & Nakashima 1973, 1974). However, these findings do not agree with more recent studies which have shown that gluconeogenesis in the normal rat is mediated predominantly by alpha adrenergic receptors (for references see section 1.3). Nor are they in agreement with the findings to be presented here or those df Malbon and co-workers (Malbon et al. 1978a, 1980; Malbon 1980a). A more complete discussion of this is deferred to section 4.1.

Other Tissues.

Effects of thyroid hormones on catecholamine-mediated responses in a number of other tissues have been reported. These relatively few studies have been recently reviewed by Kunos (1981) and will not be described in detail here. However, of some relevance to the present studies, in as much as they deal with carbohydrate metabolism, are the effects of thyroid hormones on insulin secretion by the pancreas. Beta receptor stimulation enhances insulin secretion while alpha stimulation is inhibitory (Young and Landsberg 1977). Beta receptor activity dominates in hyperthyroidism and hence the mixed alpha/beta agonist adrenaline potentiates glucose-induced insulin secretion (Okajima and Ui 1978). In contrast, alpha receptors dominate in the hypothyroid pancreas, which results in an adrenaline-induced suppression of insulin release (Okajima and Ui 1978). Precisely how these changes relate to

the overall altered metabolic state of hypothyroidism or hyperthyroidism is unknown.

#### Summary.

From the foregoing it is clear that generalizations about the effects of thyroid hormones on adrenergic receptor-mediated responses must be made with caution. Most evidence suggests that hypothyroidism results in a decreased beta receptor activity in the heart and adipose tissue. Adipocytes from hamsters appear to represent a noteworthy exception (Fain 1981). Similar effects have been noted in pancreas (Okajima and Ui 1978). Alpha receptor activity is apparently enhanced in cardiac tissue from hypothyroid animals although this effect has not been analyzed in detail with respect to selective The question of how hypothyroidism changes in alpha receptor subtypes. affects alpha receptors in adipose tissue is unresolved. One group of investigators claim there is no effect on either alpha, or alpha, receptors (Fain 1981; Garçía-Saínz et al. 1981; García-Sáinz and Fain 1982), although the suggestion has been made that such an effect has been overlooked (Kunos 1981).

Hyperthyroidism produces changes opposite to those found in hypothyroidism, although these effects are in general less clear. This is probably due to variations in species sensitivity as well as the use of different treatment regimens for the administration of thyroid hormone. The effects of hypothyroidism are in most cases reversible by thyroid hormone replacement (Kunos 1981; Fain 1981) and the effect of thyroid hormone level appears to follow a continuum. Thus, treatment of euthyroid animals with thyroid hormone produced opposite changes in cardiac adrenoceptor properties to those seen in hypothyroidism (Kunos 1977; Hashimoto and Nakashima 1978).

#### 1.2.4 5 Thyroid Effects on Adrenergic Radioligand Binding.

Direct radioligand binding studies of adrenergic receptors have been of great value in precisely defining the role of receptors in thyroid-induced changes in tissue responsiveness. While providing a powerful tool to aid in the understanding of mechanisms which underlie these changes, studies using this approach appear to have introduced even more controversial and apparently contradictory findings into the literature. These investigations are subject to the same pitfalls and problems of variations in species, age, sex, etc. (see section 1.2.3) that have hampered the development of a clear understanding of thyroid effects on adrenergic responses. Even so, results of ligand binding studies have not in all cases supported the existence of changes in receptor properties that could explain the observed alterations in responsiveness even when the same tissue was examined for both (see below). There are two possible explanations for this: (1) thyroid hormones affect changes in responsiveness by altering components of the tissue response and have no effect on receptors per se or (2) the ligand binding study does not adequately assess receptor properties. Clearly the existence of a change in receptor properties which could explain the observed change in tissue response does not preclude the possibility of additional thyroid hormone effects on post-receptor events. Even in those situations where biochemical changes, other than at the receptor site itself, are not found, one cannot rule out possible additional effects on the coupling of receptor to the effector system. In the case of the beta receptor, for which the coupling to adenylate cyclase is relatively well understood (Ross and Gilman 1981; Limbird 1981), one can make assertions with greater confidence.

The first successful adrenergic Binding studies were made in 1974 for beta receptors and in 1976 for alpha receptors (Hoffman and Lefkowitz 1980). Although this field has grown rapidly in a short period, the methodology is still being developed, as is an appreciation for the complexity of the receptor

systems being studied. Hence the over-zealous interpretation of earlier studies has undoubtedly contributed to incorrect and misleading conclusions. Moreover, the common practice of restating in print many times the results of a single study has led to the premature acceptance of data without adequate reinvestigation and confirmation. Hence, any dogmatic statements about adrenergic receptors based on binding studies must be viewed with skepticism.

Thyroid hormone effects on radioligand binding properties of adrenergic receptors has been reviewed and some of the common problems encountered discussed (Hoffman and Lefkowitz 1980; Kunos 1981; Fain 1981). The results of some of these studies in cardiac and adipose tissue will be summarized here. Cardiac Tissue.

Most studies have shown an increase in beta receptor numbers in cardiac tissue from thyroid hormone-treated rats and a decrease in hypothyroid preparations (Hoffman and Lefkowitz 1980). In general there was no or little change in receptor affinity (Kunos 1981). This is the expected result on the basis of observed alterations in tissue responsiveness (section 1.2.3). Of particular interest are studies which have demonstrated increased binding sites for [<sup>3</sup>H] DHA, a beta antagonist, when nanomolar amounts of thyroid hormone were added in vitro to either cultured rat myocardial cells (Tsai and Chen 1978), or ventricle slices of hypothyroid (Chang and Kunos 1981) or euthyroid rats (Kempson et al. 1978). The latter study showed an early (1.5-2 h) increase in binding sites that was not affected by protein synthesis inhibitors as well as a slower increase (15 h) which was. Although most investigations have demonstrated increased alpha adrenergic responsiveness of hypothyroid hearts (section 1.2.3), results of alpha receptor radioligand binding assays have been variable. Thus, studies using [<sup>3</sup>H] DHE to evaluate alpha receptor changes in hypothyroid rats have shown either a decrease (Ciaraldi and Marinetti 1977, 1978) or no change (Williams and Lefkowitz 1979; McConnaughey et al. 1979) in alpha receptor binding site density. Reasons for these discrepant findings

are not clear, although it is possible that the presence of presynaptic  $alpha_2$  receptor binding sites or differences in the physical-biochemical properties of preparations from normal and hypothyroid rats may have been factors. However, Noguchi and Whitsett (1983) using  $[^{3}H]$  prazosin, a selective  $alpha_1$  antagonist, also found a decrease in  $alpha_1$  receptor binding sites in hypothyroid rats hearts. On the other hand, Sharma and Banerjee (1978) had shown that  $T_3$  treatment of thyroidectomized rats resulted in a decrease in  $[^{3}H]$  DHE binding site density. This finding was later confirmed by Chang and Kunos (1981) who found that the binding of  $[^{3}H]$  prazosin was decreased in hypothyroid rats after treatment with  $T_3$  for 36 h, while  $[^{3}H]$  DHA binding was increased. Similarly,  $T_4$  treatment of hypophysectomized rats for two days produced an increase in  $[^{3}H]$  DHA binding sites and a decrease in alpha\_1 receptor binding sites identified as prazosin-suppressible  $[^{3}H]$  WB 4101 binding (Kunos et al. 1980).

Most studies on tissue responsiveness to sympathomimetics have shown that hyperthyroidism decreases the alpha receptor component in cardiac tissue (section 1.2.3). Those findings are supported by the results of receptor binding studies. Thus, Ciaraldi and Marinetti (1977, 1978), Williams and Lefkowitz (1979) and McConnaughey et al. (1979) found evidence of a decrease in alpha receptor binding in cardiac tissue from hyperthyroid rats.

Adipose Tissue.

 $[^{3}H]$ DHA has been used to assess thyroid-induced changes in beta receptors of adipocytes. Malbon et al. (1978b) and Goswami and Rosenberge (1978) reported no change in beta receptor binding sites with thyroid state. However, beta binding site numbers have been reported to be decreased in hypothyroid (Giudicelli 1978) and increased in hyperthyroid rats (Ciaraldi and Marinetti 1978). Non-linear Scatchard plots (Malbon et al. 1978b) in the absence of negative cooperativity (Malbon and Cabelli 1978) suggest that  $[^{3}H]$ DHA binding may be to heterogeneous sites and hence these studies must

be interpreted with caution. Moreover, Malbon (1980b) found that the well known effect of guanyl nucleotides on agonist affinity for the beta receptor (Ross and Gilman 1980; Limbird 1981) was lost in preparations from hypothyroid or hyperthyroid adipose tissue. This finding suggests that there may be a complex regulatory role for guanyl nucleotides in thyroid hormone effects on fat cells.

Alpha receptors have been little studied in adipose tissue. Giudicelli et al. (1980) have reported a decrease in binding site density for  $[^{3}H]$  DHE in adipose tissue from hyperthyroid hamsters and no change in hypothyroid hamster adipocytes. García-Sáinz et al. (1981) on the other hand, found no change in  $[^{3}H]$  DHE binding in adipocyte membranes from hyperthyroid hamsters. Neither of these studies differentiated alpha<sub>1</sub> and alpha<sub>2</sub> receptors, and a definitive statement about possible changes in alpha receptors in adipocytes must await studies with selective alpha<sub>1</sub> and alpha<sub>2</sub> ligands.

Findings of work on adrenergic receptor binding properties in hepatic tissue will be discussed later (section 1.3.4).

#### 1.3 Hepatic Adrenergic Receptors.

#### 1.3.1 Classification of hepatic adrenergic receptors.

The body stores glucose in the form of glycogen which can be rapidly mobilized, such as during strenuous muscular exercise, to supply fuel for glycolysis when local glucose or oxygen are in short supply. Hepatic glycogenolysis together with gluconeogenesis supply glucose for tissues such as brain and erythrocytes which utilize glucose as their basic metabolic fuel. Hence, the liver has the important role of maintaining the euglycemic state during short periods of fasting.

The control of liver glycogen metabolism has recently been reviewed (Hers 1976; Hems and Whitton 1980). In 1940, Soskin (1940) put forward the concept

that circulating glucose itself is the primary stimulus for glycogen breakdown or deposition in the liver. However, glucose storage and output is additionally modulated by hormonal and nervous influences, principally those of insulin and glucagon. These hormones act to control the activation or inactivation of glycogen phosphorylase, the rate-limiting enzyme for glycogen breakdown and glycogen synthase, the enzyme that catalyzes glycogen synthesis (Hems and Whitton 1980).

<u>.</u>

A variety of hormones in addition to glucagon can activate glycogen phosphorylase and hence rapidly stimulate hepatic glycogenolysis (table 1.1), however, the physiological importance of these is unknown. Adrenergic agonists are included in this list. Although the effects of catecholamines on hepatic carbohydrate metabolism have been studied for many years (Hornbrook 1970), it is only recently that the mechanism of adrenergic activation of hepatic phosphorylase has been clarified. This is particularly ironic in view of the important role the study of adrenaline effects on phosphorylase activation played in the discovery of cAMP by Sutherland and coworkers in the late 1950's (Robison et al. 1971). The historical development of our understanding of adrenergic receptors in the liver has been reviewed by Haylett (1979) and will be dealt with only briefly here.

Hepatic adrenergic receptors could not, until recently, be easily classified as alpha or beta (see section 1.2.3). At least part of this difficulty arose because of attempts to analyze the effect of catecholamines on the hyperglycemic response in the intact animal; catecholamine-induced hyperglycemia is a complex response involving direct effects on the liver, effects on insulin and glucagon secretion by the pancreas, glucose uptake and lactate production by muscle, hpolysis in adipose tissue, as well as a number of other minor effects (Hornbrook 1970; Young and Landsberg 1977; Himms-Hagen 1972). In addition, the response shows considerable variation with species and route of catecholamine administration (Ellis 1967). Some of the

difficulties with classification also arose because of the apparent refractoriness of the hyperglycemic response to conventionally accepted doses of adrenergic antagonists (Hornbrook 1970). However, studies from this laboratory (Kan et al. 1979) as well as others (Jenkinson et al. 1978) demonstrated the need for very high antagonist concentrations in order to block the response even in <u>in vitro</u> preparations. This is most likely due to the high capacity of the liver to take up antagonists thus reducing the effective concentration in the vicinity of the receptor (Jenkinson et al. 1978; Kan et al. 1979).

Although it was well accepted on the basis of the studies of Sutherland and co-workers (Robison et al. 1971) that adrenaline could activate phosphorylase via the beta receptor-adenylate cyclase pathway (see section 1.3.2), the role of alpha receptors in contributing to this response was questioned (Hornbrook 1970). Earlier evidence that, at least in the fed rat, adrenaline-induced hyperglycemia had the characteristics of an alpha receptor-mediated effect (Fleming and Kenny 1964) did not gain acceptance until the work of Sherline et al. (1972). These authors demonstrated that in the perfused rat liver catecholamines activate glycogenolysis by a beta receptor-mediated effect resulting in a rise in cAMP, but also by a However, they cautioned cAMP-independent alpha receptor-mediated effect. that phosphorylase activation by alpha-receptors could be secondary to hypoxia resulting from alpha-mediated constriction of the liver vasculature (Sherline et al. 1972). At about the same time Haylett and Jenkinson (1972b) showed. that catecholamine-stimulated glucose release from ginuea pig liver slices involved both beta and alpha receptors. Later, Exton and co-workers confirmed the existence of alpha receptor-mediated activation of glycogen phosphorylase and glucose output in suspensions of isolated rat hepatocytes, thus ruling out the possible effects of tissue hypoxia (Exton and Harper 1975; Hutson et al. 1976). In addition, studies showed evidence of alpha receptor-mediated activation of gluconeogenesis in isolated rat liver cells (Tolbert et al. 1973;

17

ł

Kneer et al. 1974; Hutson et al. 1976) that was also independent of changes in cellular cAMP (Tolbert et al. 1973; Hutson et al. 1976).

1

It appears therefore, that both alpha and beta receptors are present in the liver. Adrenaline effects on carbohydrate metabolism in the rat liver are mediated predominantly via the alpha receptor with only a minimal contribution by the beta receptor (Exton and Harper 1975). However, it has been shown that the relative importance' of these two receptor types varies with the species examined, which also accounts for some of the confusion in earlier attempts to classify the hepatic receptor. At present there is good evidence that (at least in vitro) both alpha and beta components contribute significantly in guinea pig (Haylett and Jenkinson 1972b; Osborn 1975) and rabbit liver (Haylett 1976). In the mouse, alpha receptors appear to play a relatively more important role (Seydoux et al. 1979). The dog and cat, long considered to mobilize hepatic glycogen stores via a beta receptor-mediated effect (Haylett 1979), have not been studied in detail since the importance of hepatic alpha receptors became accepted. However, Kuo et al. (1977) have demonstrated in vivo that the cat has both alpha and beta receptors that contribute to the glycogenolytic response.

In man, earlier studies comparing the effects of adrenergic agents on the hyperglycemic response in vivo indicated a major role for alpha receptors (see Exton 1979 for references). More recently, Rizza et al. (1980) analyzed this response in greater detail. These authors infused adrenaline into healthy volunteers after "clamping" glucose, insulin and glucagon by a continuous infusion of somatostatin. Under these conditions they could analyze the effects of selective alpha and beta blockade and concluded that hepatic glucose output was directly increased by a beta adrenergic mechanism. This effect could be augmented by alpha receptor-mediated inhibition of pancreatic insulin secretion and beta-stimulation of glucagon release. Alpha blockade was judged to be relatively ineffective in inhibiting adrenaline's direct hepatic effect. The

relative contribution of alpha and beta receptors to catecholamine effects in the liver also appears to be influenced by age (Hornbrook 1978; Blair et al. 1979a; Kawai and Arinze 1981), sex (Bitensky et al. 1970; Studer and Borle 1982), hormonal state (Wolfe et al. 1976; Chan et al. 1979a) and the presence of pathological changes in liver (section 4.5). These effects are discussed more fully in a later part of this thesis.

## Table 1.1

Agents that Stimulate Hepatic Glycogenolysis.\*

Cyclic AMP-dependent Agents

beta adrenergic agonists

**gluc**agon

parathyroid hormone

Cyclic AMP-independent Agents

alpha adrenergic agonists vasopressin

angiotensin II

A23187

\* for references see text and Hems (1977).

# 1.3.2 The mechanism of beta adrenergic activation of hepatic glycogen phosphorylase.

Our contemporary understanding of the mechanism of catecholamine activation of glycogen phosphorylase via the beta receptor is based on the classic studies of Sutherland and co-workers (Robison et al. 1971). Much of this information has been reviewed recently (Exton et al. 1981; Exton 1982) and will be briefly summarized here.

The presently accepted scheme as depicted in figure 1.1 is based on the contributions of a great many investigators working with various tissues and experimental systems. Interaction of catecholamine with the beta adrenergic receptor 'on the plasma membrane of the liver cell leads to the activation of adenylate cyclase. This interaction involves at least three separate proteinaceous membrane components: the receptor, a catalytic moiety and a guanine nucleotide-sensitive regulatory protein which is absolutely required for hormone-activation of the cyclase (Ross and Gilman 1980; Limbird 1981). The cyclase catalyzes the formation of cAMP, the so-called second messenger, from ATP. Cyclic AMP in turn activates protein kinase. Protein kinase exists in the cell as an inactive tetrameric structure composed of two catalytic and two regulatory subunits. Cyclic AMP binds to the regulatory sites and causes the complex to dissociate producing two active catalytic subunits. This reaction is reversible and activation is halted when the rate of cAMP extrusion from the cell or metabolism to 5'AMP by phosphodiesterase returns intracellular cAMP levels to their basal value.

Axelrod and coworkers (Hirata and Axelrod 1980) have recently proposed that coupling of the beta adrenergic receptor and adenylate cyclase involves local methylation of membrane phospholipids. They demonstrated that in several tissues stimulation of the beta receptor (as well as other cAMP-linked receptors) activated a membrane bound methyltransferase II which catalyzes methylation of membrane phospholipids. This effect is not cAMP-mediated and
results in a localized increase in membrane fluidity. They postulated that this increase in fluidity permits the coupling of receptor to adenylate cyclase and in so doing leads to cAMP generation. If indeed this hypothesis proves correct, it adds a vital link in the understanding of receptor mechanisms. Recent reports suggest, however, that this mechanism may not be generalized. In particular, it may not apply to the rat liver, where nearly complete inhibition of methyltransferase activity left hormone-induced stimulation of adenylate cyclase virtually intact (Colard and Breton 1981; Sanche et al. 1982).

Hormonal stimulation of adenylate cyclase results ultimately in the phosphorylation of a number of cellular components (Garrison et al. 1979). Two isoenzy mes of cAMP-dependent protein kinase have been shown to exist. in mammalian tissues and these appear to differ only in their regulatory subunits (Walsh 1978). Thus, the catalytic subunit possesses rather broad substrate specificity. Three substrates for the ATP-dependent phosphorylation by this protein kinase have thus far been identified (Exton 1979). As depicted in figure 1.1. phosphorylase kinase is phosphorylated to the active form by the catalytic subunit of protein kinase. This enzyme, in turn, activates glycogen phosphorylase b to the a or active form by a second ATP-requiring phosphorylation. Phosphorylase a catalyzes the breakdown of glycogen leading to increased glucose output. The second substrate for cAMP-dependent protein kinase is glycogen synthase. Phosphorylation of this enzyme results in inactivation and hence decreased glycogen synthesis. Most recently 'L-type pyruvate kinase has also been shown to be inactivated through phosphorylation by cAMP-dependent protein kinase. This enzyme represents an important control point in the Embden-Meyerhoff pathway. Its inactivation probably plays an' important role in the stimulation of gluconeogenesis by cAMP-dependent A number of other hepatic enzyme activities are thought to be hormones. regulated by cAMP-dependent protein kinase phosphorylation, but these have not been as well studied (Exton et al. 1981). Also, there exist a number of

hepatic elements that undergo cAMP-dependent phosphorylation for which the function of this regulation is not yet known (Garrison et al. 1979).

In addition to beta receptor agonists, several other hormones are able to activate, hepatic glycogenolysis via the adenylate cyclase-cAMP-dependent protein kinase pathway (Hems 1977; Hems and Whitton 1980). Those so far identified include glucagon and are listed in Table 1.1. Interaction of glucagon with its membrane-bound receptor activates adenylate cyclase in a manner exactly analogous to that described above for the beta receptor (Exton 1982). Hence post-receptor events that lead to specific protein phosphorylations are likely to be identical for these hormones. However, one cannot ignore the possible existence of additional effects of these hormones that might not involve the adenviate cyclase system. For example, high concentrations of glucagon have been shown to mobilize intracellular calcium, an effect which is undetectable at physiologic concentrations (Blackmore et al. 1978, 1979c). While most evidence suggests that this effect does not contribute significantly to the stimulation of glycogenolysis (Exton et al. 1981), one cannot rule out a role in regulating other metabolic processes in the hepatocyte. Côté and Epand (1979) have described the synthesis of a glucagon analogue which antagonizes glucagon stimulation of adenylate cyclase and yet is capable of activating glycogenolysis in isolated rat hepatocytes. Cardenas-Tanus in et al. (1982) found that angiotensin II inhibited glucagon-stimulated cAMP accumulation in isolated hepatocytes, but did not alter the hormone's metabolic effects in the cell. The mechanisms underlying these findings are not yet established and until more is known about these effects it is reasonable to accept that cAMP is the major mediator of the effects of glucagon.

Insulin is able to counteract the effects of adrenaline and glucagon on hepatic glucose output (Soderling and Park 1974). Although the mechanism of this effect is not fully understood, insulin lowers glucagon-stimulated cAMP accumulation and activation of protein kinase, phosphorylase kinase as well as

glycogen phosphorylase by low concentrations of glucagon (for references see Blackmore et al. 1979a). Loten et al. (1978) have demonstrated that insulin will increase the "low  $K_m$ " cAMP phosphodiesterase of liver suggesting that this is the basis for the hormone's effects on cAMP-mediated processes. However, possible additional effects on adenylate cyclase itself in the intact cell or the regulation of phosphoprotein phosphatase activity may be important (Soderling and Park 1974). The effects on beta receptor-mediated glycogenolysis are likely to be same, but direct experimental evidence for this is lacking. (Van de Werve et al. 1977). Inhibition of alpha receptor activation of phosphorylase by insulin will be discussed in section 1.3.3. There is also some evidence that stimulation of parasympathetic nerves to the liver can have effects similar to insulin infusion on cAMP accumulation (Shimazu 1968).

The properties and regulation of liver phosphoprotein phosphatases represent somewhat of a grey area in our understanding of hepatic metabolic regulation. This has been the subject of recent reviews (Lee et al. 1980; Li 1982; Krebs and Beavo 1979). Since the activity level of phosphoprotein enzymes is dependent on the net result of phosphorylation and dephosphorylation, phosphoprotein phosphatase activities may also play an important role in metabolic regulation by hormones. Numerous forms of phosphoprotein phosphatases varying in their physicochemical properties and substrate specificities have been described in many tissues including the rat liver (Li 1982). Early findings which suggested that each phosphoprotein possessed a corresponding specific phosphatase gave way to the concept of a single multifunctional phosphoprotein phosphatase with broad substrate specificity; the previously identified forms representing degradation products that retained some activity (Li 1982). However, most current evidence would support the existence of multiple forms of phosphoprotein phosphatase with distinct, but overlapping substrate specificities. For example, glycogen synthase phophatase and phosphorylase phosphatase have different properties and can be

separated by ion exchange chromatography (Lee et al. 1981). Phosphoprotein phosphatase activity can be regulated in a number of ways some of which may be important mechanisms in vivo (Krebs and Beavo 1979). Consideration of the regulatory phenomena described for glycogen phosphorylase phosphatase will serve to illustrate this point. The activity of phosphorylase a is inhibited by glucose which, by binding to phosphorylase, increases its affinity for deactivation by the phosphatase (Li 1982). Other metabolites such as glucose-6-phosphate have a similar so-called substrate-directed modulating influence (Krebs and Beavo 1979). Direct protein-protein interactions may also play an important role. Thus, phosphorylase kinase can interact with a phosphatase and inhibit the dephosphorylation of phosphorylase a (Krebs and Beavo 1979). Phosphorylase a can inhibit the phosphatase-mediated activation of glycogen synthase (Hers 1976). Hers and coworkers<sup>6</sup> (Hers 1976) have proposed that these interactions play an important role in intracellular regulation. They point out that high levels of phosphorylase a may inhibit the concomitant activation of glycogen synthase. Their scheme, however, is not generally accepted (Krebs and Beavo 1979; Lee et al. 1981; Li 1982).

Specific proteinaceous phosphatase inhibitors have been isolated from muscle and liver as well as other tissues (Lee et al. 1981). These have been most extensively examined in muscle where two different protein inhibitors have been characterized. The finding that one of these inhibitors must be phosphorylated by a cAMP-dependent protein kinase to be able to inhibit a specific phasphatase activity, led Cohen (1978) to propose a central role for these inhibitors in the regulation of glycogen metabolism. In his scheme (Cohen 1978), the cAMP-dependent phosphorylation of inhibitor-1 would lead to inhibition of the phosphoprotein phosphatase responsible for the activation of glycogen synthase and inactivation of glycogen phosphorylase, assuming these are the same enzymes. While this scheme still lacks much supporting evidence, Foulkes and Cohen (1979) have demonstrated the <u>in vivo</u> phosphorylation of

muscle inhibitor-1 in response to adrenaline injection.

A similar scheme could be hypothesized to play a role in regulation of hepatic glycogen metabolism. Some sparce evidence for the possible hormonal regulation of phosphorylase phosphatase in liver does exist. Goris et al. (1978) have reported finding a phosphorylatable phosphoprotein phosphatase inhibitor ine dog liver. Shimazu and Amakawa (1975) demonstrated that stimulation of sympathetic nerves to rabbit liver enhanced glycogenolysis and attributed this to inhibition of phosphorylase phosphatase they observed. Under those conditions, Shimazu and Amakawa (1975) found no increase in cAMP and suggested the presence of an unidentified neurotransmitter to account for the phosphatase inhibition. It is convenient to digress at this point from cAMP-mediated effects and deal briefly with phosphoprotein phosphatases in relation to hepatic alpha adrenergic actions which are discussed more fully in the next section. Hepatic alpha receptor-mediated activation of phosphorylase is independent of cAMP and proceeds via the release of intracellular calcium ions (section 1.3.3). Although phosphoprotein phosphatase activity is dependent on divalent cations (Li 1982), a role for calcium in regulating this activity has not been established (Khoo and Steinberg 1975). Furthermore, Proost et al. (1979) in experiments very similar to those of Shimazu and Amakawa (1975) found no evidence for a role of phosphatase in mediating the activation of phosphorylase by nerve stimulation. Blackmore and Exton (1981)<sup>2</sup> found that alpha adrenergic stimulation of phosphorylase was impaired in hepatocytes from rats genetically deficient in phosphorylase b kinase activity. This finding suggests that it is modulation of the kinase activity that controls the level of activation of phosphorylase, but a possible additional role for the phosphatase cannot be ruled out. This area requires further study.

Before concluding this brief discussion of hormonal effects occurring via the cAMP system it is worth drawing attention to the fact that a number of cellular activities not directly related to carbohydrate metabolism are regulated

in a similar manner. Beavo and Krebs (1979) have compiled a list of 22 enzymes that have been reported to undergo phosporylation-dephosphorylation.

<del>1</del>

# 1.3.3 The mechanism of alpha adrenergic activation of hepatic glycogen phosphorylase.

Agents that are capable of stimulating hepatic glycogenolysis by a cAMP-independent mechanism are listed in table 1.1. While most studies have concentrated on alpha adrenergic effects in the liver, vasopressin and angiotensin II actions have also been examined in some detail. These latter two agents have effects similar to alpha adrenergic agonists and evidence for this will be summarized later in this section. Since the first clear demonstration by Exton and co-workers (Exton and Harper 1975) that alpha adrenergic agonists can activate glycogen phosphorylase without a concomitant increase in cellular cAMP levels, much effort has been expended in establishing a mechanism for this action. Considerable advances have been made and this work has been reviewed recently (Exton 1979, 1980, 1981; Exton et al.

Goldberg and co-workers (Goldberg et al. 1973) first suggested that alpha adrenergic effects were mediated by cGMP in a manner analogous to the involvement of cAMP in beta receptor-mediated events. Although, alpha stimulation does result in an increase in cGMP levels in the liver (Pointer et al. 1976), this effect can be produced by a number of other agents that do not activate glycogen phosphorylase (Pointer et al. 1976; Hems et al. 1978). Hence, this idea was discarded.

Many lines of evidence pointed to the involvement of calcium ions in the cAMP-independent activation of glycogen phosphorylase. Alpha adrenergic-activation of hepatic glycogenolysis is inhibited when extracellular calcium is removed (Keppens et al. 1977; Assimacopoulos-Jeannet et al. 1977;

27 /

Van de Werve et al. 1977; Chen et al. 1978). The divalent cation ionophore. A23187, mimicked alpha adrenergic effects on hepatic glycogenolysis in a calcium-dependent fashion (Keppens et al. 1977, Assimacopoulps-Jeannet et al. 1977). Blackmore et al. (1978) then demonstrated that the rapid chelation of extracellular calcium ion by the stochiometric addition of EGTA with the agonist did not significantly impair the alpha adrenergic activation of phosphorylase in isolated rat hepatocytes. The rapid addition of EGTA in this manner would presumably prevent the influx of extracellular calcium during stimulation without depleting intracellular calcium stores. In fact, Blackmore et al. (1978) were able to identify a net loss of cellular calcium occurring during stimulation that was even greater when EGTA was added as described. Soon afterward it was confirmed that alpha stimulation induced a loss of calcium from isolated hepatocytes (Chen et al. 1978; Burgess et al. 1979) pr perfused liver (Blackmore et al. 1979b,c; Althaus-Salzman et al. 1980; Reinhart et al. 1982a). However, not all groups were able to identify a net loss of calcium after alpha stimulation (Foden and Laudle 1978; Murphy et al. 1980). The reason for this difference is not clear. Nonetheless, there was sufficient evidence to support the postulation of Exton and coworkers (Exton 1979) that the transient increase in cytosolic calcium was responsible for the activation of glycogen phosphorylase by phosphorylase kinase, an enzyme known to be sensitive to calcium (see below). This was supported by the findings of Murphy et al. (1980) who showed that alpha adrenergic stimulation of isolated hepatocytes produced a two- to three-fold increase in free cytosolic calcium that correlated well, with respect to time-course and dose-response, with phosphorylase activation.

While it is now well accepted that alpha receptor-mediated activation of phosphorylase occurs via the production of a transient increase-in cytosolic calcium, there is still some disagreement about the source of this calcium. Studies using atomic absorbtion spectrometry (Blackmore et al. 1979b,c) or

chlortetracycline° flourescence (Babcock et al. 1979) to measure mitochondria-associated calcium suggest that most of the calcium released during alpha receptor stimulation is mitochondrial in origin. On the other hand, studies using  $^{45}$ Ca showed an increased mitochondrial uptake of  $^{45}$ Ca after stimulation (Foden and Randle 1978; Althaus-Salzman et al. 1980; Poggioli et al. 1980). Unfortunately in these latter studies total mitochondrial calcium was not determined and it is unclear whether the increased <sup>45</sup>Ca reflects a true change in calcium content or merely increased exchange. One of these groups (Poggioli et al. 1980) in a subsequent study (Berthon et al. 1981) suggested that these discrepancies were most likely the result of differences between experimental protocols. In the later study, they demonstrated an initial, transient increase in mitochondrial calcium (measured by atomic absorbtion spectrometry) followed by a loss of mitochondrial calcium. They concluded that the transient increase in cytosolic calcium was the sum of sequential calcium release from endoplasmic reticulum and then mitochondria. However, these results were not confirmed in a detailed kinetic analysis of the effects of adrenaline on calcium distribution in isolated rat hepatocytes by Barritt et al. (1981). These authors demonstrated that adrenaline caused a loss of calcium from an intracellular compartment which includes mitochondria and also an early increase of calcium transport from the extracellular medium. Attempts to analyze a possible role for extracellular calcium in mediating the effects of alpha stimulation by the use of verapamil, a calcium channel blocker, met with difficulties owing to a direct antagonistic effect of this drug on the alpha receptor site (Blackmore et al. 1979d).

Blackmore et al. (1982) re-examined the role of extracellular calcium in the alpha receptor-mediated activation of phosphorylase in isolated hepatocytes. From the results of this study they concluded that net calcium influx does not occur during the initial response to alpha agonists since they could find no increase in cellular calcium measured by atomic absorbtion spectrometry as

early as 5s after stimulation. Also, rapid chelation of extracellular calcium, as described above, had no detectable effect on phosphorylase activation. They surmized that uptake of chemical calcium or  ${}^{45}$ Ca must represent a recovery In support of these findings, Reinhart et al. (1982a) found that net process. efflux of calcium from perfused livers occurred as early as 7.1 + 0.5 s after phenylephrine-stimulation. Based on measurements of chemical calcium after subcellular fractionation of perfused livers, these workers concluded that most calcium released by stimulation is mitochondrial and, to a lesser extent, from endoplasmic reticulum. A large portion of the calcium released is extruded from the cell by a calcium-sensitive ATPase located in the hepatic plasma membrane (Lotersztajn et al. 1981). Calcium is quickly reaccumulated which accounts for the influx of extracellular calcium noted by others. Based on existing evidence, this scheme provides the most reasonable mechanism for alpha receptor-mediated effects on hepatic calcium movements.

٥

The locus of calcium's effect in the glycogenolytic cascade is most likely at the level of phosphorylase kinase (see figure 1.1). This enzyme is calcium sensitive, its activity being enhanced without undergoing phosphorylation (Khoo and Steinberg 1975; Shimazu and Amakawa 1975; Vandenheede et al. 1979). Blackmore and Exton (1981) examined the alpha adrenergic activation of phosphorylase in hepatocytes from rats genetically deficient in phosphorylase b kinase activity. They showed that stimulation of alpha receptors in hepatocytes isolated from these rats produced the same degree of calcium mobilization as in normal rats, but phosphorylase activation was impaired. Glucagon, vasopressin and the calcium ionophore, A23187, had similar effects. Calcium-dependent modulator protein (calmodulin) has been identified as the delta subunit of phosphorylase kinase from skeletal muscle (Cohen et al. 1978). A similar component has been identified in purified rat liver phosphorylase kinase (Chrisman et al. 1982) indicating the locus for the calcium-sensitive regulation of this enzyme. Taken together, these results suggest that

phosphorylase kinase is the most likely site of calcium's stimulation of the glycogenolytic cascade.

Alpha adrenergic stimulation has also been shown to inhibit glycogen synthase by a stable phosphorylation (Hutson et al. 1976; Murphy et al. 1980) and to stimulate gluconeogenesis (Pointer et al. 1976; Hutson et al. 1976; Chan and Exton 1978) probably in part as a result of phosphorylation and inactivation of pyruvate kinase (Chan and Exton 1978; Garrison and Borland 1979). Garrison and co-workers (Garrison et al. 1979) found that stimulation of hepatocytes in the presence of  ${}^{32}\text{PO}_{\text{A}}$  with either noradrenaline plus a beta antagonist or with glucagon resulted in phosphorylation of the same 11 to 12 cytosolic proteins, three of which were identified as phosphorylase, glycogen synthase and pyruvate kinase (Garrison et al. 1979). Therefore, Exton (1980) has suggested that the liver possesses a calcium-sensitive protein kinase which has a substrate specificity similar to cAMP-dependent protein kinase. The possibility that phosphorylase kinase may be the multifunctional calcium-sensitive kinase has not been examined extensively. However, Payne and Soderling (1980) have isolated from rabbit liver a calmodulin-dependent glycogen synthase kinase that did not phosphorylate glycogen phosphorylase. This finding suggests the existence of at least two and possibly more calcium sensitive protein kinases in liver.

There are a number of alpha adrenergic effects in liver for which the role of calcium ions and protein phosphorylation is less clear. Haylett and Jenkinson (1972a,b) demonstrated that alpha stimulation caused a rapid efflux of potassium from guinea pig liver slices that was associated with a reversible 10 mV hyperpolarization; the resting membrane potential was approximately -30 to -40 mV. Activation of beta receptors had little effect on either of these responses although it was equally as effective as activation of alpha receptors in stimulating glucose output (Haylett and Jenkinson 1972a) or activating phosphorylase in the guinea pig (Osborn 1979). In a later study with

isolated guinea pig hepatocytes, Burgess et al. (1979) found that the calcium ionophore, A23187, was capable of producing a similar potassium efflux to that seen with alpha agonists, suggesting the presence of a calcium-sensitive potassium channel. Another group working with guinea pig liver slices (Weiss and Putney 1978) could not demonstrate a similar effect of A23187 on the potassium release response (measured as  $^{86}$ Rb efflux), although these workers also identified a role for calcium in the response. The reason for this difference 15 not clear.

Not all species show a similar effect of alpha stimulation on ion movements in the liver (Jenkinson et al. 1978). Thus, rat liver cells do not respond to either alpha agonists or A23187 by potassium loss as seen in the guinea pig and it has been suggested that a calcium-sensitive potassium channel may be lacking in the rat (Burgess et al. 1979). Blackmore et al. (1979c) found that the alpha agonist phenylephrine as well as glucagon caused a rapid uptake of potassium in perfused rat liver. In addition, the hyperpolarizing effect of alpha stimulation observed in guinea pig (Haylett and Jenkinson 1972a,b) does not occur in all species; depolarization has been reported in the dog (Jenkinson et al. 1978). The reason for these interesting species differences is not known and is worthy of further study.

Alpha stimulating effects on amino acid transport (LeCam and Freychet 1978), oxygen consumption and lactate release<sub>o</sub> (Jakob and Diem 1975; Blair et al. 1979b), ureagenesis (Corvera and García-Sáinz 1981), glycolysis and Na<sup>+</sup>K<sup>+</sup>transport (Becker and Jacob 1982), cGMP accumulation (Pointer et al. 1976) and the turnover of phosphatidyl inositol (Kirk et al. 1977; Billah and Mitchell 1978; Tolbert et al. 1980) have also been reported.

Although many of the effects of alpha receptor stimulation in liver appear clearly to be linked to calcium mobilization, the pathway whereby interaction of agonist with receptor leads to the release of intracellular calcium is unknown. Whiting and Barritt (1982) have listed some of the mechanisms that

have been proposed to account for this. Discussion of all of these is beyond , the scope of this introduction. However, the possible role of phosphatidylinositol hydrolysis has received a great deal of attention and will be discussed briefly. Michell and co-workers put forward the hypothesis that receptor activation results in stimulation of phosphatidylinositol breakdown which is somehow coupled to calcium mobilization (Michell 1975). This hypothesis was based on observations that alpha receptor stimulation in liver (Kirk et al. 1977; Billah and Michell 1979; Tolbert et al. 1980) as well as a number of other tissues (for references see Exton 1980; Fain and García-Sáinz 1980; Michell and Kirk 1981) resulted in increased incorporation of  ${}^{32}PO_{4}$  into phosphatidulinositol as a result of breakdown followed by resynthesis of this phospholipid (Michell and Kirk 1981). This response is thought to precede calcium mobilization, since it was still apparent afer removal of medium calcium and could not be reproduced by the calcium ionophore, A23187 (Billah and Michell 1978; Tolbert et al. 1980). Tolbert et al. (1980) have demonstrated that like the activation of hepatic phosphorylase (Hoffman et al. 1980b; El-Refai and Exton 1980; Aggerbeck et al. 1980a), alpha adrenergic stimulation of phosphotidylinositol breakdown is mediated by an alpha, receptor. While this hypothesis provides an attractive possible explanation for alpha effects on calcium homeostasis, it may have serious shortcomings. The most contrary evidence comes from a study by Exton's group. These authors (Prpić et al. 1982) carefully examined phosphatidylinositol breakdown induced by vasopressin, angiotensin II and adrenaline in isolated hepatocytes. Based on their study they concluded that the breakdown of phosphatidylinositol induced by these agents was too slow to play a causative role in mobilization of calcium; breakdown of the phospholipid was not seen before two min whereas a previous study (Blackmore et al. 1982) had shown that calcium movement and phosphorylase activation were at maximum by 10's. Furthermore, the effect of these hormones on phosphatidylinositol hydrolysis was abolished in

calcium depleted hepatocytes indicating that the effect is calcium dependent (Prpić et al. 1982). Previous studies (see above) which found the opposite did not provide evidence of adequate calcium depletion (Prpić et al. 1982). If these findings are substantiated by further study, the phosphatidylinositol hypothesis (at least in the case of the liver) will have to be discarded and the search for the mechanism of calcium mobilization continued.

Insuln inhibits not only cAMP-mediated activation of glycogenolysis and gluconeogenesis (section 1.3.2), but also alpha adrenergic effects in both isolated hepatocytes and perfused liver (Massaque and Guinovart 1978; Blackmore et al. 1979a; Dehaye et al. 1981). Insulin inhibited the alpha receptor-mediated release of calcium suggesting that its effect is at or proximal to the production of the signal for intracellular calcium mobilization (Blackmore et al. 1979a; Dehaye et al. 1981). Interestingly, calcium release or phosphorylase activation by vasopressin, angiotensin II or A23187 was not inhibited in the presence of insulin. These observations would indicate that insulin's effects are not on intracellular calcium stores, but rather uniquely on the mechanism whereby alpha adrenergic agonists mobilize intracellular calcium, possibly on the alpha receptor itself.

Recent evidence from radioligand binding studies utilizing purified liver plasma membranes (see section 1.3.4) has suggested the existence of both  $alpha_1$  and  $alpha_2$  receptors in the rat liver (Hoffman et al. 1980a,b). The regulation of endogenous noradrenaline release by presynaptic alpha receptors has been demonstrated in the dog liver (Yamaguchi 1982); such receptors are usually of the  $alpha_2$  type. However,  $alpha_2$  receptors at post-synaptic sites have been described for a number of tissues and their stimulation has been shown to have an inhibitory effect on adenylate cyclase (Fain and García-Sáinz 1980). Hence, Jard et al. (1981) were able to demonstrate  $alpha_2$  inhibition of adenylate cyclase in rat liver plasma membranes. Angiotensin II and vasopressin were also tested in this system, but only angiotensin II inhibited

34 <sup>~</sup>

adenylate cyclase activity.

There exists in the literature a number of earlier studies that also; showed evidence of an inhibitory alpha effect on adenylate cyclase (Boyd and Martin 1976), cAMP accumulation in isolated hepatocytes (Chan et al. 1979a), as well' as the perfused rat liver (Sherline et al. 1972). In none of these studies was the effect characterized with respect to alpha receptor subtype. "More recently, Exton and coworkers (Assimacopoulos-Jeannet et al. 1982) demonstrated that alphan receptor activation could inhibit glucagon-stimulated gluconéogenesis, phosphorylase activation and glucose output in isolated rat hepatocytes. These effects appeared to be due in part to inhibition of cAMP accumulation through  $alpha_l$  receptor stimulation, although additional effects (possibly involving redistribution of intracellular calcium) are implied by the "observed suppression of exogenous cAMP- or dibutyryl cAMP-mediated Vasopressin and angiotensin II had inhibitory effects similar gluconeogenesis. to catecholamine. This inhibitory alpha, effect is different from that described by Jard et al. (1981) for alpha, receptors in isolated membranes. One might expect that both phenomena may play a role in hepatic regulation. However, the rather precise conditions required to observe the alpha, effect (5 mM EDTA, 50-200 mM Na and 1-200 µM GTP) bring into question its physiological significance,

Under certain conditions alpha receptor stimulation has been shown to result in increases in hepatic cAMP accumulation. Chan and Exton (1977) have obtained evidence that alpha receptor stimulation resulted in cAMP accumulation in isolated hepatocytes that had been depleted of calcium by incubation in the presence of 1 mM EGTA. The increased levels of cAMP were associated with increased activation of glycogen phosphorylase. This response in calcium depleted hepatocytes was found to be somewhat less sensitive to agonists, but more sensitive to antagonists than the alpha mediated activation of phosphorylase in normal cells, but still appeared to be of the

alpha, subtype. It was suggested that the removal of calcium somehow alters the properties of the alpha receptor to allow coupling with adenylate cyclase. An alternate explanation would be that calcium depletion somehow unmasks the activity of the third type of alpha receptor which is positively coupled to adenylate cyclase. More recently, Jackowsky et al. (1982) demonstrated alpha, adrenergic stimulation of adenylate cyclase in calcium depleted liver plasma membranes. Vasopressin and angiotensin II did not have similar effects under identical conditions. Whether or not this could represent the same alpha receptor-mediated accumulation of cAMP in calcium depleted hepatocytes described above is not known. The descrepancy between these findings and those of Jard et al. (1981) is most likely due to differences in the preparations used and conditions of the assay; stimulation by alpha, receptors disappears in the presence of high GTP, magnesium or calcium ion concentrations (Jackowski et al. 1982), whereas inhibition by alpha, receptors required high GTP and Furthermore, Jackowski et al. (1982) claim sodium ion (Jard et al. 1981). that in spite of the presence of 5 mM EDTA, the conditions employed by Jard et al. (1981) do not represent significant calcium depletion. These considerations aside, since rather artificial conditions are required to demonstrate these actions, their physiological importance is dubious, but they might provide some insight into the biochemistry of the receptor-adenylate cyclase system. It is noteworthy that Blair et al. (1980b) have demonstrated alpha mediated cAMP accumulation in normal rat hepatocytes without calcium depletion and this finding was confirmed by Okajima and Ui (1982). This surprising result will require further investigation.

Hence, it appears that the liver possesses both  $alpha_1$  and  $alpha_2$  receptors which mediate different responses. Alpha<sub>1</sub> receptors cause the mobilization of calcium by a mechanism that is not yet established. The increased cytosolic calcium so produced can activate or deactivate a number of enzymes some of which are involved in hepatic carbohydrate metabolism.

Alpha<sub>2</sub> receptors inhibit or stimulate adenylate cyclase depending on the conditions of the assay. An effect of alpha<sub>2</sub> receptor stimulation on glucose output or phosphorylase activation has not as yet been described and the physiological role, if any, of hepatic alpha<sub>2</sub> receptors is still open to question. Vasopressin and angiotensin II are thought to activate glycogen phosphorylase and hence stimulate glycogenolysis by a mechanism similar to that described for alpha receptors. The evidence for this is the following: (1) The effect of the hormones on glycogenolysis is independent of cAMP accumulation (Kirk and Hems 1974; Keppens et al. 1977; Garrison et al. 1979).

 $\Delta g$ 

(2) Their action in liver is critically dependent on extracellular calcium (Stubbs et al. 1976; Keppens et al. 1977; Garrison et al. 1979).

(3) Stimulation with angiotensin II or vasopressin causes marked efflux of calcium from either perfused liver (Blackmore et al. 1979c) or hepatocyte suspensions (Blackmore et al. 1978; Chen et al. 1979), and release appears to be from the same intracellular stores as are affected by alpha agonists (Chen et al. 1978; Blackmore et al. 1979b,c; Reinhart et al. 1982a).

(4) Most<sup>21</sup> of this releasable calcium is of mitochondrial origin (Blackmore et al. 1979c; Murphy et al. 1980; Reinhart et al. 1982a).

(5) The pattern of phosphorylation of cytosolic protein in response to vasopressin or angiotensin II stimulation of isolated hepatocytes is the same as that observed with alpha receptor activation (Garrison et al. 1979; Garrison and Wagner 1982).

(6) Vasopressin or angiotensin II also stimulate phosphatidylinositol breakdown (Tolbert et al. 1980; Billah and Mitchell 1978; Kirk et al. 1977),

(7) potassium uptake by rat liver (Blackmore et al. 1979c), and

ړ.\*

(8) gluconeogenesis in hepatocyte suspensions (Whitton et al. 1978).

• A final point of evidence (9) comes from the studies of Bréant et al. (1981) who demonstrated that alpha adrenergic agonists and vasopressin

produced a dose-dependent heterologous desensitization of the phosphorylase response in isolated hepatocytes. However, others (Morgan et al. 1982) found that no desensitization of the alpha<sub>1</sub> glycogenolytic response was seen in perfused rat liver provided adequate calcium reaccumulation occurred before repeating the stimulus. These latter authors suggested that the heterologous desensitization found by Bréant et al. (1981) was due to depletion of an intracellular calcium pool(s) that is common for vasopressin, angiotensin II and alpha<sub>1</sub> agonists. Later, Keppens and DeWulf (1982) showed that desensitization occurred at low temperatures, while calcium fluxes could only be produced at higher temperatures. Hence they have proposed that this desensitization is a membrane phenomenon whereby an hypothetical hormone-effector complex is inactivated by a desensitizing stimulus. In any case, all of these findings support the existence of a common pathway for vasopressin, angiotensin II and alpha<sub>1</sub> agonists, although the issue of the mechanism of the heterologous desensitization remains unresolved.

6

Just as for alpha agonists (Dehaye et al. 1980), vasopressin (Cantau et al. 1980) and angiotensin II (Campanile et al. 1982) effects are mediated by interaction with a plasma membrane receptor. Hence, it seems reasonable to postulate that the common pathway for the hepatic action of these hormones may be established as early as the coupling of receptors to the effector system, just as it is for the beta adrenergic and glucagon receptor-adenylate cyclase systems (section 1.3.2). However, as discussed earlier in this section, insulin inhibits the hepatic actions of alpha<sub>1</sub> receptor agonists, but not angiotensin II or vasopressin suggesting some early divergence of the activating mechanism or selective action of insulin at the alpha<sub>1</sub> receptor site itself.

The calcium ionophore, A23187, has been much used in the study of the role of calcium in the actions of glycogenolytic hormones in the liver and a brief comment about its action seems in order before concluding this section.

A23187 produces a dose-dependent increase in phosphorylase <u>a</u> as a consequence of its ability to mobilize calcium (Blackmore et al. 1978). It has a biphasic effect on calcium content: at low concentrations the drug releases calcium from intracellular stores and produces marked calcium efflux from the liver cell (Blackmore et al. 1978). At higher concentration, influx of calcium from the extracellular medium is also produced (Blackmore et al. 1978). Hence, when extracellular calcium is low, only the efflux response is seen (Blackmore et al. 1978; Chen et al. 1978). The biphasic nature of this response may explain the inability of Friedman et al. (1979) to detect calcium efflux in response to A23187. These workers tested only relatively high doses of the drug and did not examine its effects in calcium-free medium.

Maximal stimulation with A23187 appears to deplete calcium from the hormone sensitive intracellular pool (Chen et al. 1978). After stimulation, there is a marked reduction in hormone-releasable mitochondrial calcium (Babcock et al. 1979), supporting the concept that this represents a significant, portion of the hormone-sensitive intracellular calcium pool.

1.3.4 Identification of hepatic adrenergic receptors by radioligand binding.

Radiolabelled agonists and antagonists have been used to identify putative adrenergic receptor sites in various tissues including the liver (Exton 1980; Hoffman and Lefkowitz 1980). While most of these studies have met with / considerable success, some controversy regarding the most suitable ligands to measure adrenergic receptors in the liver still exists (Hoffman et al. 1981). This is particularly true of the hepatic alpha receptor, which, owing to the great interest in the mechanism of alpha adrenergic effects, is being extensively studied. The task of measuring changes in the properties of this receptor system is made more complicated by the existence of alpha<sub>1</sub> and alpha<sub>2</sub> receptors in liver, as well as other receptor-like binding sites (see below). The results of these studies will be summarized here.

Early attempts to identify adrenergic receptors with <sup>3</sup>H-catecholamines have been reviewed by Cuatrecasas et al. (1974). These authors concluded that <sup>3</sup>H-catecholamine binding in intact tissues and microsomal preparations was predominantly to non-specific catechol binding sites and did not reflect receptor binding. U'Prichard and Snyder (U'Prichard 1981) found in several tissues that reliable receptor binding with <sup>3</sup>H-catecholamines could only be obtained when precautions were taken to minimize catechol-directed non-specific binding and auto-oxidation of the ligand. Hence, El-Refai et al. (1979) used  ${}^{3}$ H-adrenaline and  ${}^{3}$ H-noradrenaline in the presence of 10<sup>-5</sup> M propranolol to study alpha receptor binding sites in rat liver plasma membranes. Catechol and ascorbate were included in the incubation mixture to reduce non-specific binding and oxidation of the catecholamines respectively. Using this approach, these authors were able to identify two classes of specific alpha receptor-like binding sites, suppressible by 10-5 M phentolamine. One class (120 to 160 fmol/mg protein) had a high affinity for agonists and sensitivity to guanine nucleotides, and was thought to represent the physiological alpha adrenergic receptor. The second site showed a high affinity for antagonists, a markedly low affinity for agonists and was readily labelled by [<sup>3</sup>H] DHE (1275 fmol/mg protein, Kd=7.2 nM). Both receptors were found to be of the alpha, subtype (El-Refai and Exton 1980a). Later these same authors demonstrated that trypsin treatment of isolated membranes resulted in a time-dependent increase in the high affinity agonist site and a concomitant decrease in low affinity sites (El-Refai and Extor 1980b). Geynet et al. (1980) also found that limited proteolysis of rat liver membranes increased the affinity and number of [<sup>3</sup>H] noradrenaline binding sites. However, guanine nucleotide sensitivity was lost with treatment. It was suggested that trypsin treatment converted low affinity to high affinity sites and that trypsin may mimic an in vivo regulation of alpha receptors (El-Refai and Exton 1980b).

Similar [<sup>3</sup>H] DHE binding results were obtained by another group

40

>

(1.7 pmol/mg protein, Kd=1.8 nM; Clarke et al. 1978); however, these authors implied that all of this binding was to the physiological receptor sites. Indeed, Aggerbeck et al. (1980b), have correlated [<sup>3</sup>H] DHE binding to plasma membranes and activation of glycogen phosphorylase in isolated hepatocytes. While the relative potencies of both agonists and antagonists for inhibition of ligand binding and phosphorylase activation could be correlated (Aggerback et al. 1980b), agonists were approximately ten-fold more potent in activating phosphorylase than antagonizing [<sup>3</sup>H] DHE binding. Antagonists were approximately ten-fold less potent in antagonizing phosphorylase activation than <sup>[3</sup>H] DHE binding. Methodological problems complicate interpretation of these The low apparent inhibitory potency of antagonists was probably results. artifactual; these authors used preparations with high tissue to medium ratios at which the great capacity of liver cells to take up drugs results in a reduction of free drug concentration in the vicinity of the adrenergic receptor (Jenkinson et al. 1978; Kan et al. 1979). This leads to an overestimation of the  $K_d$  of the receptor based on phosphorylase activation. On the other hand, the difference between  $EC_{50}$ 's and binding  $K_d$ 's of agonists was probably greater than estimated, as the  $EC_{50}$ 's were unusually high, suggesting that the hepatocytes used may have been of poor viability. This difference cannot be explained by the existence of spare receptors, since these are absent in rat liver cells (Kunos et al. 1983).

Hoffman et al. (1979) have quantified the relative number of  $alpha_1$  and  $alpha_2$  receptors in several tissues including the liver. Although  $[{}^{3}H]$  DHE binds both these receptor subtypes, analysis of displacement curves produced with prazosin, an  $alpha_1$  receptor antagonist, yohimbine, an  $alpha_2$  receptor antagonist and phentolamine, a mixed  $alpha_1/alpha_2$  antagonist, could yield information about each subtype. Using this approach, these authors could not identify significant  $alpha_2$  receptors in rat liver plasma membranes. In a subsequent later study (Hoffman et al. 1980a), it was found that guanine

nucleotides had a regulatory effect on agonist binding to  $alpha_2$  but not  $alpha_1$ receptors. A small, but significant effect of guanine nucleotide on [<sup>3</sup>H] DHE agonist displacement curves (Hoffman et al. 1980a) prompted a re-examination of liver membranes for the presence of alpha, receptors. Approximately 20% of alpha receptor sites labelled by [<sup>3</sup>H] DHE were alpha, (Hoffman et al. Guanine nucleotides are apparently able to transform alpha2 1980a,b). receptors from a state with high affinity for agonists to a low affinity state (Hoffman et al. 1980a,b). [<sup>3</sup>H] Adrenaline, at low concentrations labels mainly the high-affinity alpha, site, and it was suggested that binding of <sup>3</sup>H-catecholamines described by El-Refai et al. (1979) was to these sites and not to  $alpha_1$  receptors; however, at higher  $[^{3}H]$  adrenaline concentrations alpha, sites are also labelled (Hoffman et al. 1980b). The ability of trypsin to mediate the transformation of these sites (El-Refai and Exton 1980b) remains unexplained. Hoffman et al. (1981) compared the findings of  $[^{3}H]$  DHE binding in liver plasma membranes with  $[^{3}H]$  prazosin and  $[^{3}H]$  yohimbine as selective probes for  $alpha_1$  and  $alpha_2$  receptors respectively. Athese authors found this to be a usable approach to delineate alpha receptor subtypes.

Geynet et al. (1981) compared  $[{}^{3}H]$  noradrenaline,  $[{}^{3}H]$  prazosin and  $[{}^{3}H]$  DHE binding in hepatic plasma membranes. All three ligands displayed characteristics expected for binding to alpha<sub>1</sub> receptors. Antagonist competition studies suggested that  $[{}^{3}H]$  noradrenaline and  $[{}^{3}H]$  prazosin labelled different sites, while  $[{}^{3}H]$  DHE bound to both. Moreover,  $[{}^{3}H]$  prazosin sites could be converted to  $[{}^{3}H]$  noradrenaline binding sites by limited proteolysis with alpha-chymotrypsin. Based on these results these authors speculated that  $[{}^{3}H]$  noradrenaline binding sites could be converted to  $[{}^{3}H]$  noradrenaline binding sites these authors are alpha\_1 receptor, while  $[{}^{3}H]$  prazosin binding is to a precursor form of the active receptor and/or an alpha<sub>1</sub> binding site not coupled to the response.

Thus, radiolabelled ligand binding studies directed towards identifying hepatic alpha receptors are complicated by the existence of several forms of

alpha receptor-like binding sites. The functional relevance of these forms and hence the most suitable ligand to measure the physiologically important alpha receptors remain debatable issues.

Another approach to the study of hepatic alpha adrenergic, receptors involved the use of radiolabelled phenoxybenzamine ([<sup>3</sup>H] POB), an irreversible alpha antagonist. Using this ligand it was feasible to label sites with the expected properties of alpha receptors in intact, viable hepatocytes (Kan et al. 1979) or purified liver plasma membranes (Guellaen and Hanoune 1979; Guellaen However, the "specific" binding of [<sup>3</sup>H] POB in these latter et al. 1979). studies lacked stereoselectivity, which raises doubts about its relevance to the physiological alpha receptor. Guellaen et al. (1979) went on to solubilize [<sup>3</sup>H] POB binding site and estimated the molecular weight under non-denaturing conditions at 96,000 daltons by gel filtration. In a subsequent study by the same group SDS-PAGE vielded three peaks labelled by [<sup>3</sup>H] POB; one with a subunit molecular weight of 44,800 daltons, was identified by competition experiments as the alpha receptor binding protein (Guellaen et al. 1982). In a more recent study, Kunos et al. (1983) demonstrated that at nanomolar concentrations [<sup>3</sup>H] POB bound to membrane sites with the characteristics of alpha<sub>1</sub> receptors including marked stereoselectivity for agonists. Micromolar concentrations of antagonists and millimolar concentrations of agonists protected a second set of sites from binding  $[^{3}H]POB$  similar to those identified by Guellaen et al. (1982). However, these latter sites lacked stereoselectivity and thus their physiological function is not known. Kunos et al. (1983) solubilized an hepatic membrane component labelled by [<sup>3</sup>H] POB with properties of alpha, receptors including stereoselectivity and found a subunit molecular weight of approximately 80,000 daltons with a smaller 58,000 dalton protein thought to be a proteolytic fragment of the receptor site. Interestingly, a 58,000 dalton protein (SDS-PAGE) was also identified by Graham et al. (1982) in rat liver using affinity chromatography with an agarose-linked

analogue of prazosin. They also proposed that this protein was the ligand-binding subunit of the alpha<sub>1</sub> receptor. The above discrepancies in molecular weight may be due to variable success in inhibiting proteolytic degradation of a larger (approximately 80,000 dalton) membrane component that represents the native alpha<sub>1</sub> receptor.

The study of hepatic beta adrenergic receptors by the use of radiolabelled ligands is also not without difficulties. The beta antagonist,  $[{}^{3}\text{H}]$  DHA has been shown to bind specifically to sites on the liver plasma membrane with characteristics of beta<sub>2</sub> receptors (Munnich et al. 1981). The density of binding sites was low: 60 fmol/mg membrane protein as one would expect since the predominant adrenergic receptor in normal rat liver is of the alpha type. However, studies with a second potent beta antagonist,  $[{}^{125}\text{I}]$  IHBP gave values of 188 ± 27 fmol/mg protein (Wolfe et al. 1976) and 24 ± 3 fmol/mg protein (Malbon 1980) in plasma membranes purified from intact liver and isolated hepatocytes respectively. However, the unusually low affinity for  $[{}^{125}\text{I}]$  IHBP found in the study by Wolfe et al. (1976; K<sub>d</sub>=2.6 + 1.5 nM) casts doubt on the validity of the binding results from this study.

Despite these reports of successful beta receptor binding studies in the liver, experiments in our laboratory were not successful in identifying high affinity, stereoselective sites in liver membranes from adult male rats using either  $[^{3}H]$  DHA or  $[^{125}I]$  IHBP. Similar negative results have been recently published (Dax et al. 1981). Therefore, further characterization of beta receptor binding sites will be necessary before the above results can be accepted without skepticism.



FIGURE 1.1 Calcium- and cAMP-dependent pathways for the activation of liver glycogen phosphorylase.

"C" and "R" are the catalytic and regulatory subunits of protein kinase respectively. For details concerning the calcium-dependent activation pathway refer to section 1.3.3. For details concerning the cAMP-dependent pathway see section 1.3.2.

#### **1.4.1** Anatomical Considerations

Claude Bernard (1849) discovered that "piqure" of the floor of the fourth ventricle produced profound hyperglycemia and glycosuria, and concluded that this effect was mediated via nervous influences on the liver. In spite of this early demonstration, it has been only recently accepted that neuronal control of glucose output and hepatic metabolism in general may be of physiological importance. The anatomy and function of hepatic nerves has been extensively reviewed by Lautt (1980) and will not be detailed here.

Autonomic control pathways originate in the hypothalamus. Stimulation of the ventromedial hypothalamic nucleus produces hyperglycemia by neuronal activation of hepatic gluconeogenesis and glycogenolysis via the sympathetic system (Shimazu and Ogasawara 1975). Stimulation of the lateral hypothalamic nucleus produces slight hypoglycemia and can be linked to activation of hepatic glycogen synthesis via the vagus. These conclusions have also been supported by results of selective lesioning of hypothalamic sites which appear to have effects opposite to stimulation (Shimazu and Ogasawara 1975). Ventromedial hypothalamic lesioning produces a hyperglycemic-obese state and represents the modern counterpart of Claude Bernard's piqûre experiments (Hems and Whitton 1980).

#### 1.4.2 The Effects of Splanchnic Nerve Stimulation.

Direct electrical stimulation of the splanchnic nerves results in hyperglycemia due to release of pancreatic glucagon, catecholamine release from the adrenal medulla and hepatic sympathetic nerve activity (Himms-Hagen 1972). Shimazu and Amakawa (1968a) demonstrated that splanchnic stimulation could produce a marked activation of liver glycogen phosphorylase in rabbits

that was not eliminated by adrenalectomy or pancreatectomy. These authors went on to demonstrate that this activation could not be blocked by the beta receptor antagonist, dichloroisoproterenol, whereas activation by injected catecholamine could be blocked (Shimazu and Amakawa 1968b). Later studies by the same investigators revealed that nervous activation of glycogen phosphorylase occurred independently of any change of cAMP, while levels of this cyclic nucleotide were elevated when phosphorylase was activated by injection of either glucagon or adrenaline (Shimazu and Amakawa 1975). No stable change in phosphorylase kinase activity could be demonstrated after nerve stimulation, but a two to three-fold increase was noted after glucagon or epinephrine injection. The activity of phosphorylase phosphatase decreased with nerve stimulation. These authors concluded, therefore, that splanchnic nerve stimulation elevated phosphorylase activity by inhibition of phosphorylase phosphatase rather than activation of phosphorylase kinase via a cAMP-mediated mechanism analogous to that occurring after adrenaline or glucagon injection. They further speculated that some unknown factor, not noradrenaline, was released from sympathetic nerve endings to affect this response. As already discussed (section 1.3.2), other workers (Proost et al. 1979) could not confirm a neurally mediated decrease in phosphorylase phosphatase. Moreover, it has been shown that in cat, mouse and rabbit, the activation of phosphorylase by nerve stimulation can be blocked by alpha antagonists, but is only slightly or not at all affected by beta antagonists (Lautt 1979; Proost et al. 1979; Seydoux et al. 1979).

In the light of more recent information about the role of alpha and beta adrenergic receptors in mediating hepatic glucose output (section -1.3.1), the above findings are best explained as follows. Neurally-mediated activation of hepatic phosphorylase occurs via release of noradrenaline and subsequent action mainly on alpha receptors. Thus no change in cAMP levels is detectable and phosphorylase kinase is allosterically activated by calcium, not by a stable

phosphorylation as occurs with cAMP-dependent protein kinase. Conversely, injected adrenaline activates glycogen phosphorylase via the beta adrenoceptor/adenylate cyclase system and the classical cAMP-dependant protein kinase cascade (Walsh 1978). A similar proposal has been made by Proost et al. (1979) and Osborn (1978) to explain the above-described findings.

There is evidence from several studies that in other tissues  $alpha_1$  and  $beta_1$  receptors are innervated while  $alpha_2$  and  $beta_2$  are not, as proposed for alpha receptors in vascular smooth muscle (Yamaguchi and Kopin 1980) and for beta receptors in cerebral cortex (Minneman et al. 1979). The glycogenolytic response in liver is mediated by  $alpha_1$  receptors (section 1.3.3) or  $beta_2$  receptors (see section 4.4).

#### 1.4.3 The Effects of Stimulation of the Hepatic Parasympathetic Nerves.

The effects of hepatic parasympathetic stimulation have been less well studied. Shimazu and Amakawa (1975) demonstrated that simultaneous stimulation of the splanchnic nerves and vagi in the rabbit would block the activation of glycogen phosphorylase, but would have no effect on activation of this enzyme or elevated cAMP levels resulting from adrenaline or glucagon injection. Vagal stimulation by itself had no effect on these parameters, although it did produce a marked activation of glycogen synthase which could be counteracted by simultaneous splanchnic nerve stimulation (Shimazu 1971). As expected, incorporation of radioactive glucose into liver glycogen was enhanced by vagal stimulation independent of pancreatic release of insulin (Shimazu and Fujimoto 1971). Glycogen deposition is decreased in acutely vagotomized rats (Mondon and Burton 1971). In spite of the similarity of vagal stimulation and insulin action on glycogen synthase, Shimazu (1967) has shown that the neurally mediated effect is more rapid than the hormonal effect. Just as for the sympathetic nerves, an exact physiological role for the parasympathetic innervation of the liver in glucose control has not been

elucidated.

# 1.5 Effect of Thyroid State on Hepatic Carbohydrate Metabolism

# 1.5.1 Glycogen Levels and Phosphorylase Activity.

Hepatic glycogen levels tended to be normal in hypothyroid rats and markedly decreased in hyperthyroidism (Hoch 1974). These differences were maintained in the ex vivo perfused liver preparation (Laker and Mayes 1981) suggesting that hormonal or neural influences were not the only reason for this effect on glycogen levels. In spite of this difference, perfusate glucose stabilized at similar concentrations in hypothyroid, hyperthyroid and normal rats when pyruvate and lactate were present (Laker and Mayes 1981). In vivo hepatic glycogen levels appear to be dependent on nutritional state. Thus. Okajima and Ui (1979), by examining the kinetics of radiolabelled glucose in the intact rat fasted for 20 h, found that glycogen levels were decreased by 60% in hypothyroid and 90% in hyperthyroid rats compared with euthyroid controls. These authors concluded that enhanced hepatic gluconeogenesis and diversion of gluconeogenic products from liver glycogen to blood glucose helped to maintain circulating blood levels. On the other hand, a slower rate of gluconeogenesis in hypothyroid rats led to a slower release of glucose into circulation and suppressed glycogenesis.

Takahashi and Suzuki (1975) reported no effect of hypothyroidism or hyperthyroidism on total glycogen phosphorylase or synthase activities in rapidly excised and frozen rat livers. When total glycogen phosphorylase activity was assayed in isolated hepatocytes it was found to be decreased (30%) after  $T_3$ treatment for two days and slightly increased in hypothyroidism although this latter change was statistically insignificant (Malbon and LoPresti 1981).

## 1.5.2 Hepatic Enzyme Activities.

Gluconeogenesis is enhanced in hyperthyroid rat liver and the activities of gluconeogenic enzymes are increased (Bottger et al. 1970). Conversely, the rate of hepatic gluconeogenesis is lower in the thyroidectomized rat (Menehan and Weiland 1969). The profile of gluconeogenic enzymes in the thyroidectomized rat resembles that of the diabetic rat (Baquer et al. 1976). This is not surprising since hypothyroidism results in a state of relative insulin resistance (Bray and Jakobs 1974). However, consideration of changes in enzyme levels alone will not adequately explain the alterations seen in either of these states. Thyroid hormone appears to be required for the full expression of glucagon or catecholamine effects, and therefore it is likely that altered response to hormonal stimuli may be of greater importance.

# 1.5.3 Effect on Redox State and Energy Charge.

The activities of a number of critical enzymes involved in carbohydrate metabolism are modulated by the redox state (NADH/NAD) or energy charge (ATP/ADP,AMP) of the cell. These factors are profoundly influenced by alterations in thyroid state.

ۍ

Reducing equivalents produced during glycolysis are transferred from the cytoplasm to mitochondia by a cycle involving alpha-glycerophosphate dehydrogenase. The activity of this enzyme is low in thyroidectomized rats and increased several fold by thyroid hormone treatment (Bernal and DeGroot 1980). Thus, the hypothyroid liver is in a more reduced state (increased NADH/NAD) and hyperthyroid liver a less reduced state than normal. In addition to direct effects of NADH/NAD on enzyme activities there is some evidence that the redox state of the cell may have important consequences for the action of certain hormones. Clark and Jarret (1978) reported that under a oxidized conditions the ability of glucagon to produce an increase in cAMP and stimulate glucose output is decreased and have attributed this to an effect of

nicotinamide nucleotides on cAMP phosphodiesterase activity.

While the total adenylate nucleotide pool is unchanged in hypothyroid animals, the "energy charge" (ATP/ADP+AMP) is elevated (Baquer et al. 1976) over a range for which the response of many regulatory enzymes is marked (Atkinson 1968). Hems and Whitton (1980) have summarized the effects of various metabolites on glycogen phosphorylase <u>a</u> and <u>b</u>. These authors point out that AMP has an activating effect on both forms of the enzyme, while ATP and ADP are inhibitory. Hence, the increased "energy charge" of the liver cell may contribute somewhat to the reduced glucose output seen in response to glucagon and catecholamines in the hypothyroid rat. Additional, as yet undocumented, effects may also be of importance.

気

#### 1.5.4 Summary.

It appears that thyroid hormones have complex effects on carbohydrate metabolism in the intact animal. A number of studies have examined the various influences of these hormones on factors which may be important in producing these changes, but a complete explanation has not been forthcoming. This is in part due to the existence of multiple ways in which thyroid hormones might produce these effects. Of particular interest in the context of the present study is the ability of thyroid hormones to modulate the effects of other hormones.

#### 1.6 Formulation of the Problem.

Thyroid hormones can influence the response of various tissues to catecholamines, but the mechanism of these interactions is not clear. Most studies of this phenomenon have examined effects of thyroid state on the properties of adrenoceptors in cardiac and adipose tissue. Both alpha and beta receptors are present in cardiac and adipose tissue, and it appears that in certain cases the altered sensitivity to catecholamines observed in

hyperthyroidism or hypothyroidism may be attributed to changes in the balance of these receptors. In the rat heart where both receptor types mediate the positive inotropic response to catecholamines, hypothyroidism increases the alpha and decreases the beta receptor component of this response. Excess thyroid hormone has the opposite effect.

The activation of liver glycogen phosphorylase by catecholamines is also mediated by both alpha and beta receptors. If the effects of thyroid state on cardiac adrenergic receptors can be generalized, one may hypothesize that hyperthyroidism would enhance the beta receptor contribution to the activation of hepatic glycogen phosphorylase, a predominantly alpha receptor-mediated response in the normal adult rat. Unexpectedly the opposite was found: the pattern of adrenoceptor activation of phosphorylase was unchanged in hyperthyroid rats, but hypothyroidism resulted in a shift from alpha to a beta type of response.

In order to further investigate the nature of the changes observed in the hypothyroid rat, the following questions were considered.

1. Is the effect of thyroidectomy the result of changes in adrenergic receptors or some post receptor event in the glycogenglysis cascade?

2. Is the effect selective for adrenergic receptors or are other glycogenolytic hormones similarly influenced?

3. Are these changes reversible by thyroid hormone replacement?

# SECTIOŃ TWO

# EXPERIMENTAL PROCEDURES

()

Ĺ

ġ., ŧ

# 2.1 Animals

Male Sprague-Dawley rats (CD strain, Canadian Breeding Farms Ltd.) were housed in the McIntyre Animal Center under constant conditions (22°C, 12 h light-dark cycle), maintained on Purina Rat Chow and tap water ad Rats were rendered hypothyroid by surgical thyroidectomy under libitum. sodium pentobarbital anaesthesia (50 mg/kg). Success of the procedure was assessed by the absence of normal growth during the ensuing 10 to 14 weeks, after which time animals were used for this study. Hyperthyroidism was induced by daily intraperitoneal injections of  ${\rm T}_4$  (1 mg/kg) freshly dissolved in 0.01 N NaOH for seven days. T3-treated hypothyroid rats received daily intraperitoneal injections of 0.25 mg/kg freshly prepared in 0.01N NaOH for 2 or 4 days, the last injection being given approximately 12 h before the Euthyroid control animals were maintained under identical experiment. conditions. In one series of experiments hypothyroid rats were treated with cortisol (50 mg/kg) every 12 h beginning 48 h before the experiment. All animals used in the present study weighed between 250 and 350 g.

# 2.2 Drugs and Chemicals.

Adrenaline (1-epinephrine bitartrate, Sigma), d-adrenaline (d-epinephrine bitartrate, Sterling-Winthrop), phenylephrine (1-phenylephrine HCl, Sigma), isoproterenol (d,1-isopropyl arterenol HCl, K&K) and noradrenaline (1-norepinephrine bitartrate, Sigma) were dissolved just prior to use in isotonic saline containing 1 mM HCl. dl-Propranolol HCl (Ayerst) and phentolamine (phentolamine methanesulfonate, Ciba) were dissolved in isotonic saline. Dihydroergocryptine (Sandoz) and A23187 (Calbiochem) were dissolved in ethanol and kept at  $-20^{\circ}$ C. Phenoxybenzamine (phenoxybenzamine HCl, SK&F) was kept at  $-20^{\circ}$ C dissolved in acidified ethanol. Triiodothyronine (T<sub>3</sub>, 3,3',5-triiodo-L-thyroinine, Sigma) and L-thyroxine (T<sub>4</sub>, Sigma) were dissolved in

vasopressin, Sigma) and glucagon (Sigma) were dissolved in isotonic saline containing 1 g% BSA and in the case of glucagon 0.2 M glycine (pH 10).

Collagenase (type I),  $\alpha$ -D-glucose-l-phosphate (grade V), BSA, cAMP and beef heart protein kinase were from Sigma and were used directly. Difco gelatin was obtained from Fisher. Glycogen (grade V-S) was also from Sigma and was used after purification as described by Thomas et al. (1968). Gpp(NH)p was from Boehringer-Mannheim.

 $^{45}$ CaCl<sub>2</sub>,  $\alpha$ -D-[U- $^{14}$ C] glucose-1-phosphate, [<sup>3</sup>H] Prazosin (17.1 Ci/mmol) and [<sup>3</sup>H] cAMP were from New England Nuclear.

All other chemicals and reagents were from usual commercial sources.

# 2.3 Isolation of Hepatocytes.

Hepatocytes were isolated between 9:00 and 11:00 a.m. each day by the method of Berry and Friend (1962), with some modifications. Rats were anesthetized with sodium pentobarbital (50 mg/kg i.p.) and heparinized (500 I.U./kg i.v.). The abdominal cavity was opened with a wide transverse incision. The vena portae was cannulated and perfused with calcium-free Krebs-Henseleit bicarbonate buffer (Krebs and Henseleit 1932; 115 mM NaCl, 3.7 mM KCl, 1.2 mM MgSO<sub>4</sub>, 2.2° mM KH<sub>2</sub>PO<sub>4</sub>, 25 mM NaHCO<sub>3</sub>, and 20 mM glucose) for 10 min at a flow rate of 30 to 35 ml/min. During this period the liver was transferred to a perfusion-aeration apparatus. After this preperfusion, the medium was changed to 100 ml of Krebs-Henseleit buffer containing 2.5 mM calcium<sup>1</sup> and 0.05% crude collagenase and perfusion was continued for 10 to 20 min with recycling. The liver was then disrupted by gentle combing. Incubation of the collagenase-liver suspension was continued in a shaker bath for an additional period such that the liver cells were exposed to collagenase for a total period of 30 min. The perfusion and all subsequent incubations were done at  $37^{\circ}C$ , under an atmosphere of 5%  $CO_2$  in  $O_2$ , and maintained at a pH between 7.0 and 7.4 by the addition of 150 mM, NaOH as required. At

the end of the incubation, the cell suspension was filtered through nylon mesh and centrifuged for 2 min at 50 x g. The cell pellet was washed and centrifuged twice, and resuspended in Krebs-Henseleit bicarbonate buffer containing 2.5 mM calcium and 1.5% gelatin<sup>2</sup> pre-equilibrated with 5% CO<sub>2</sub> in O<sub>2</sub> at 37°C. Cells were either used immediately or kept briefly on ice. This method routinely yielded cells with 95% viability as assessed by exclusion of trypan blue<sup>3</sup>. Cell suspensions contained 30 to 40 mg wet weight of cells per ml unless otherwise specified. Wet weight was determined by centrifuging 1.0 ml aliquots of the cell suspension at 1000 rpms in a Sorval GLC-1 centrifuge for 2 min in tared 1.5 ml plastic tubes. The supernatant was removed, the tube walls carefully dried and the weight of the cell pellet determined.

<sup>1</sup>Calcium concentration was 4.2 mM in the collagenase solution in earlier experiments. The data for these experiments appears in all figures to 3.9 except 3.1. Occasionally a white precipitate would form in this solution making perfusion of the liver less efficient. Hence, for all subsequent experiments calcium content of the collagenase buffer was reduced to 2.5 mM. Viability and performance of hepatocytes was not significantly altered by this <sup>4</sup> modification.

 $^{2}$ It was found during the course of these experiments that the viability of isolated hepatocytes was enhanced and better preserved particularly through longer incubations if 1.5% gelatin was added to the Krebs-Henseleit incubation solutions. However, this necessitated a slight modification of the assay of glycogen phosphorylase <u>a</u> in hepatocytes incubated in this medium (see section 2.4). Figures showing results of experiments done without added gelatin are indicated as such in the respective figure caption.

<sup>3</sup>Viability was assessed by combining equal volumes of cell suspension and a solution of 0.4% trypan blue in isotonic saline. After one to two min at room temperature the total number of cells and the number taking up dye were counted in 3 or 4 low power (40 X) fields with a microscope for calculation of % viability.

### 2.4 Phosphorylase Assay

Freshly prepared hepatocytes were resuspended in previously equilibrated Krebs-Henseleit buffer containing 1.5 g% gelatin at 30 to 40 mg wet weight of cells per ml. Three ml aliquots of this cell suspension were gassed with 5%  $CO_2$  in  $O_2$  and incubated in a shaker bath at  $37^{\circ}C$  and 90 to 100 cycles

per min. After a 30 min preincubation, drugs were added in 10 µl volumes or as indicated in figure captions. Unless otherwise specified, incubations for glycogen phosphorylase assays were terminated 3 min after the addition of agonist by adding 1.0 ml of the cell suspension to 5.0 ml of ice-cold Krebs-Henseleit buffer not containing gelatin. After a brief period on ice, cell suspensions were rapidly centrifuged (30 s) in an IEC International Clinical Centrifuge at maximum speed, supernatant was discarded and the cell pellet was immediately homogenized in ice-cold phosphorylase assay buffer (50 mM morpholinosulfonic acid, 150 mM NaF, 2.5 mM EDTA, 2.0 mM dithiothreitol, final pH 6.5) with a Brinkman Polytron for twice 15 s seperated by 15 s of cooling. The homogenate was centrifuged at 3000 X g for 10 min and glycogen phosphorylase was assayed in duplicate aliquots of the supernatant by measuring the incorporation of [U-<sup>14</sup>C] glucose into glycogen. Total incubation volume was 0.1 ml and the reaction was started by the addition of glycogen and  $[U-^{14}C]$  glucose-l-phosphate to give a final concentration of 1.0 g% and 15 mM respectively. The assay was rendered specific for phopsphorylase a by including 0.5 mM caffeine (Stalmans and Hers 1975). Incubation (30<sup>o</sup>C, 20 min) was terminated by spotting 50  $\mu$ l aliquots of the assay mixture on Whatman 41 filter paper discs which were further processed by sequential washing in 66% ethanol and a final wash in acetone (Thomas et al. 1968). The retained radioactivity was measured by liquid scintillation counting at an efficiency of 50 to 60%. Units of enzyme activity are nanomoles of [U-<sup>14</sup>C] glucose incorporated into glycogen per mir per mg protein.

Protein concentration in the 3000 X g supernatant was determined by the method of Lowry et al. (1951) using bovine serum albumin as a standard.

In experiments in which gelatin was omitted from the Krebs-Henseleit buffer (see footnote 2, section 2.3), 1.0 ml aliquots of cell suspension were homogenized directly in 1.0 ml of ice-cold assay buffer (100 mM morpholinosulfonic acid, 300 mM NaF, 5.0 mM EDTA, 2.0 mM dithiothreitol final pH
6.5) with a Brinkman Polytron. The subsequent steps in the assay were done as described above.

Both maximally stimulated (10  $\mu$ M adrenaline) and basal phosphorylase <u>a</u> activity was linear with respect to time for at least 20 min of incubation at  $30^{\circ}$ C and also directly proportional to protein concentration to levels greater than those in homogenates routinely prepared in the present experiments (figure

2.1).



FIGURE 2.1 Time- and protein-dependent increase in  $[U^{-14}C]$  glucose incorporation into glycogen in the assay of hepatic glycogen phosphorylase <u>a</u>.

Cells were isolated from a normal rat and incubated as described in section 2.3. After 30 min of incubation 10  $\mu$ 1 of saline alone ( $\diamondsuit$ ) or adrenaline ( $\diamondsuit$ ) was added to a 3.0 ml volume of cell suspension (80 mg/ml) to give a final concentration of 10<sup>-5</sup>M adrenaline. After 3.0 min a 1.0 ml aliquot of this suspension was homogenized in an equal volume of assay buffer (section 2.4) and assayed for phosphorylase <u>a</u> activity as detailed in section 2.4. Panel A shows the amount of  $[U^{-14}C]$  glucose incorporated into glycogen when the reaction was stopped at the indicated times. Panel B shows the amount of  $[U^{-14}C]$  glucose incorporated protein concentration when the incubation was terminated after 20 min.

E 0

#### 2.5 Cyclic AMP Determination

Cells were incubated as described for the determination of glycogen phosphorylase (section 2.4). At 1.5 min after the addition of agonist (or as indicated in figure legends) 1.0 ml aliquots of cell suspension were homogenized in 5.0% truchloracetic acid for 30 s with a Brinkman Polytron. Samples were centrifuged at 3000xg for 10 min. Precipitated gelatin was removed as a layer floating above the supernatant. One ml of supernatant was purified by passage over a 0.4 cm X 4.5 cm column of Dowex  $50X8H^+$ . The column was eluted with deionized water and cAMP was recovered in the fourth to the sixth one ml fractions. Recovery was monitored with [<sup>3</sup>H] cAMP for each batch of columns prepared and was usually 95 to 98%. Fractions containing cAMP were pooled and taken to dryness in a Brinkman Sample Concentrator. These were taken up in 0.2 to 5.0 ml of 50 mM TRIS buffer (pH 7.5) containing 4mM EDTA and cAMP was assayed in duplicate 50 µl aliquots by a radiobinding assay as described by Tovey et al. (1974).

### 2.6 Calcium Efflux Measured by Calcium Electrode

Drug effects on calcium efflux from intact hepatocyte suspensions were determined with the use of a calcium-sensitive electrode as described by Chen et al. (1978) with some modifications. Cells were washed once and resuspended in a nominally calcium-free Krebs-Hensleit buffer. Fifteen ml aliquots were incubated in 50 ml plastic Ehrlenmeyer flasks at  $37^{\circ}$ C under an atmosphere of 5% CO<sub>2</sub> in O<sub>2</sub>. To determine drug effects, a calcium-sensitive electrode (Orion Model 93-20) and a reference electrode (Orion Model 90-01) were fixed into each Ehrlenmeyer. Electrodes were connected to a pH meter (Orion Model 701) and a Variqn A-25 chart recorder. A 1000 µF capacitance was included between the pH meter output and the chart recorder to reduce the noise in the recordings. This had no effect on the drug responses measured. Cells were incubated until a stable baseline calcium level was obtained: usually 30

to 60 min. Baseline free calcium concentrations were checked by comparing the pH meter output to a standard curve for calcium prepared in 150 mM KCl (approximately isotonic) at  $37^{\circ}$ C. These values ranged between 0.068 and 0.096 mM for hepatocytes prepared from normal rats, and 0.053 to 0.108 mM for hepatocytes from thyroidectomized rats. Cummulative concentration-response curves were produced by the addition of increasing amounts of drugs as detailed in figure captions for each series of experiments. At the conclusion of each experiment, changes in external calcium concentrations were determined by the addition of 3.3 mM CaCl<sub>2</sub> calibration pulses. One ml aliquots of cells were then removed for determination of wet weight and viability (section 2.3). Only results from preparations with >85% viability were included. Results were expressed as nanomoles of calcium efflux per mg wet weight of cells. Preliminary experiments to test this method were done and representative tracings are shown in figure 3.13.

# 2.7 <sup>45</sup>Ca Efflux Studies

Ŷ.

Freshly prepared hepatocytes were resuspended in Krebs-Henseleit buffer containing 1.5 g% gelatin and 2.5 mM  $^{45}$ Ca (0.5 Ci/ml), and incubated in a shaker bath at 37°C and 80 cycles per min. Throughout this  $^{45}$ Ca loading, the suspension was continuously gassed with 5% CO<sub>2</sub> in O<sub>2</sub>. After 60 min, cells were rapidly washed twice (50xg for 1 min) and resuspended in Krebs-Henseleit solution containing 2.5 mM non-radioactive calcium. Sixty min of preloading hepatocytes with  $^{45}$ Ca has been shown to label most of the exchangeable calcium in rat hepatocytes (Chen et al. 1978). Duplicate 1.0 ml aliquots were rapidly centrifuged (1500xg for 20 s) in an IEC International Clinical Centrifuge and  $^{45}$ Ca was determined in 0.7 ml aliquots of supernatant by liquid scintillation spectrometry at approximately 65% efficiency (determined by the addition of a known amount of  $^{45}$ Ca). The mean of this value was called the zero-time efflux. The remaining suspension was incubated in 1.0 ml aliquots

with varying concentrations of agonists added in 10  $\mu$ 1 volumes. After 3 min of incubation samples were centrifuged and 0.7 ml aliquots of supernatant were assayed for <sup>45</sup>Ca activity as described above. Data are expressed as % <sup>45</sup>Ca efflux, where 100% is assigned to the increase in <sup>45</sup>Ca activity from zero-time to 3 min in the absence of added drug. With care, this method was reproducible and duplicate determinations agreed to within 5%.

This procedure for the determination of  ${}^{45}$ Ca efflux is analogous to the "calcium efflux-exchange" procedure reported by Blackmore et al. (1978), except that  ${}^{45}$ Ca released from cells is measured in the present experiments rather than  ${}^{45}$ Ca cell content in the latter. The loss of  ${}^{45}$ Ca from cells is measured after 3 min of stimulation with agonists to correspond to times at which phosphorylase a activity was determined.

### 2.8 Thyroid Hormone Assays.

Serum thyroxine ( $T_4$ ) was measured in blood samples collected at the time of hepatocyte isolation by radioimmunoassay with the use of a kit commercially available from Diagnostics Biochem Canada Inc., London, Ontario.

#### 2.9 Preparation of Liver Plasma Membranes

Plasma membranes were prepared from freshly excised livers by the method of Neville (1968) as modified by Wolfe et al. (1976). Membranes were used for the binding assay on the day of preparation.

### 2.10 [<sup>3</sup>H] Prazosin Binding Assay.

Binding of  $({}^{3}\text{H})$ prazosin was assayed in triplicate at  $31^{\circ}\text{C}$ . Incubation was for 30 min in 50 mM potassium phosphate buffer at pH 7.5 containing 0.080 to 0.160 mg of protein, 20 to 1000 pM  $[{}^{3}\text{H}]$ prazosin the indicated adrenergic agonist or antagonist and 4 mM MgSO<sub>4</sub> in a final volume of 1.0 ml. The use of protein concentrations less than 0.2 mg/ml was necessary to avoid

significant (<10%) reduction of the free concentration of  $[{}^{3}$ H]prazosin in the incubation medium. Specific binding was defined as the difference between binding in the absence and presence of 2.0 µM phentolamine. Incubation was terminated by rapid vacuum filtration onto a single Whatman GF/C glass fiber <sup>6</sup> filter followed by three 5.0 ml washes with the same buffer at 22<sup>o</sup>C. The filters were air-dried, incubated with 0.3 ml of NCS tissue solubilizer (Amersham) overnight and the radioactivity determined by liquid scintillation spectrometry in a toluene based cocktail at an efficiency of 50 to 55%. Efficiency for each sample was verified by the channels ratio technique.

### 2.11 Statistics and Expression of Results.

All data shown is either the result of a single representative experiment , or the mean  $\pm$  standard error of the mean (s.e.m.) from three or more separate cell preparations as detailed in the captions to figures and tables.

Statistical comparisons were made by either a two-sample rank test: the Mann-Whitney U-test or the Kruskal-Wallis test where more than two groups are compared (Sokal and Rohlf 1969). A probability value of less than 0.05 was taken to imply a statistically significant difference between groups.

63 ·

### SECTION THREE

## RESULTS

. (

2

8

3.1 Activation of glycogen phosphorylase by adrenergic agonists in hepatocytes from euthyroid, hypothyroid and hyperthyroid rats.

Figure 3.1 shows the time-course of activation of glycogen phosphorylase by  $10^{-6}$  M adrenaline in hepatocytes from euthyroid, hypothyroid and hyperthyroid rats. Thirty minutes of preincubation resulted in stable basal phosphorylase <u>a</u> activity. Addition of agonist was followed by a rapid (less than one min) increase in activity which remained at plateau for at least five minutes. Similar results were obtained in all three groups of animals.

Complete concentration-response curves for adrenaline, isoproterenol and phenylephrine for each group of rats are shown in figure 3.2. For all experiments shown in this figure, phosphorylase a activity was determined in aliquots of cell suspension taken immediately before (zero agonist concentration) and 3.0 min after the addition of agonists. The type of adrenoceptor involved in the activation of phosphorylase was deduced from the relative potencies of these three agonists. Basal phosphorylase a activity was not changed by either thyroidectomy or thyroxine treatment. The order of potency of agonists in hepatocytes from normal rats was adrenaline > phenylephrine > isoproterenol (figure 3.2 and table 3.1) which agrees with earlier reports that the response is mediated by alpha adrenoceptors (Sherline et al. 1972; Hutson et al. 1976). Thyroxine treatment decreased somewhat the maximum phosphorylase a response to adrenaline and phenylephrine (figure 3.2), which resulted in a slight decrease in the  $EC_{50}$  to all three agonists (table 3.1), but their relative potencies remained unaltered. However, in hepatocytes from hypothyroid rats the order of potency was markedly changed to isoproterenol > adrenaline > phenylephrine This order of potency is characteristic of beta adrenoceptors; (table 3.1). Figure 3.2 shows that in normal rats isoproterenol was not only the least  $\cdot$ potent agonist, but its efficacy was-also-significantly less than that of the other two agonists. In hypothyroid rats, the efficacy of isoproterenol was greater while the efficacies of adrenaline and phenylephrine were lower than

in euthyroid controls.

The change in response pattern developed slowly. Changes in agonist potencies were minimal in two rats tested at 2 wk following thyroidectomy although serum thyroxine levels were as low as at three months (table 3.1).



FIGURE 3.1 Time-course of activation of glycogen phosphorylase  $\underline{a}$  by adrenaline in hepatocytes from euthyroid, hypothyroid and hyperthyroid rats.

Cells were isolated as described in section 2.3. After 30 min of incubation in a medium without added gelatin (time 0), 10  $\mu$ l of saline or adrenaline plus saline was added to a 10 ml volume of cell suspension to give a final concentration of 10<sup>-6</sup> M adrenaline. At the indicated times 1.0 ml aliquots of this suspension were homogenized in an equal volume of assay buffer (section 2.4) and assayed for phosphorylase <u>a</u> activity as detailed in section 2.4. Units of enzyme activity are nanomoles of [U-<sup>14</sup>C] glucose incorporated into glycogen per min per mg protein. Each set of data points are the results from a single rat from each group.

<sup>)</sup>67





Cells were isolated as described in section 2.3. After 30 min of incubation in a medium without added gelatin, adrenaline, isoproterenol or phenylephrine (10  $\mu$ l) were added to 3.0 ml of cell suspension to give the final concentrations of agonist indicated. Phosphorylase <u>a</u> activity was determined (section 2.4) in 1.0 ml aliquots of cell suspension taken immediately before (zero agonist concentration) and 3.0 min after the addition of agonist. Units of enzyme activity are as noted in section 2.4). Each point represents the mean  $\pm$  s.e.m. for 3 to 13 separate cell preparations. The EC<sub>50</sub> is also shown. pD<sub>2</sub> values for these experiments are given in table 3.1.

### Table 3.1

The influence of thyroid state on the potency of

adrenergic agonists in activating hepatic glycogen phosphorylase.

, ,	Normal	Hyperthyroid	Hypothyroid	
、 、			10-14 wk	2 wk
Serum T4 Level (ng/ml)	50.3 ± 9.9	98.3 ± 2.0*	8.2 ± 2.4*	9.9 ± 2.1
Agonist Potencies	5		,	
Isoproterenol	5.35 ± 0.38 (5)	5.75 ± 0.34 (5)	8.77 ± 0.25* (4)	6.02 ± 0.36 (2)
Adrenaline	7.73 ± 0.11 (13)	8.41 $\pm$ 0.39	7.96 ± 0.27 (5)	$7.45 \pm 0.12$ (2)
Phenylephrine	6.21 ± 0.10 (8)	6.58 ± 0.20 (5)	5.54 ± 0.15* (3)	5.77 ± 0.12 (2)

Data are from the same experiments shown in figure 3.2. Values shown are the mean  $\pm$  s.e.m. of the pD<sub>2</sub> values for the number of experiments shown in parentheses. The pD<sub>2</sub> is the negative logarithm of the agonist concentration producing half-maximal activation of hepatic glycogen phosphorylase. For details concerning the determination of phosphorylase activity, refer to the caption for figure 3.2 and section 2.4. Serum T<sub>4</sub> levels were determined as described in section 2.8. \* indicates a significant difference from corresponding values in normal rats.

3.2 Effects of adrenergic antagonists on adrenaline activation of glycogen phosphorylase in hepatocytes from euthyroid, hypothyroid and hyperthyroid rats.

The apparent change in the type of adrenoceptor mediating phosphorylase activation was further examined by comparing the effects of adrenergic antagonists on the phosphorylase response to the mixed alpha/beta receptor agonist adrenaline in hepatocytes from each group of animals.

The altered potency of agonists in hypothyroid rats was associated with a corresponding change in the effectiveness of alpha and beta receptor antagonists. Figure 3.3 illustrates that the increase in phosphorylase activity produced by adrenaline in normal hepatocytes was substantially inhibited by the alpha antagonists phenoxybenzamine (POB) and dihydroergocryptine (DHE), but was not affected by the beta antagonist propranolol. In cells from hypothyroid animals the alpha receptor antagonists did not significantly inhibit phosphorylase activation by adrenaline, whereas propranolol was an effective inhibitor at low adrenaline concentrations. Beta receptors are generally more sensitive to stimulation by adrenaline (Kunos 1977) and the absence of block by propranolol at higher adrenaline concentrations may be due to remaining alpha receptor activity.

In hepatocytes from hyperthyroid rats both DHE and POB were effective antagonists of adrenaline-induced phosphorylase activation, while the beta receptor antagonist, propranolol, was again ineffective. This pattern of drug response is the same as that seen in hepatocytes from normal rats and confirms results obtained with adrenergic agonists: the response is mediated predominantly by alpha receptors in preparations from the hyperthyroid rat liver.



FIGURE 3.3 Effects of adrenergic antagonists propranolol, phenoxybenzamine and dihydroergocryptine on activation of glycogen phosphorylase  $\underline{a}$  by adrenaline in hepatocytes from euthyroid, hypothyroid and hyperthyroid rats.

6

Experimental details are as described in the caption to figure 3.2 except that antagonists were added at the start of the 30 min incubation period to give a final concentration of  $10^{-5}$ M in each case. Antagonists were added, 1 µl per ml of cell suspension in vehicle as specified for each in section 2.2. Units of enzyme activity are as noted in section 2.4. Each point represents the mean  $\pm$  s.e.m. for 3 to 7 separate cell preparations except for panels f and i where the results of a single representative experiment are shown. Solid bars: antagonist present. Striped bars: no antagonist present.

3.3 The effect of cortisol treatment on glycogen phosphorylase activation by adrenaline, isoproternol and phenylephrine in hepatocytes from thyroidectomized råts.

Changes in hepatic adrenergic receptors similar to those observed after thyroidectomy have been described following adrenalectomy (Chan et al. 1979a). In order to test the possibility that the effect of thyroidectomy reported here was secondary to thyroid-induced changes in the adrenal-pituitary axis, the effect of cortisol treatment on the phosphorylase response in thyroidectomized rats was examined. Results are shown in figure 3.4. Treatment of thyroidectomized rats with cortisol (50 mg/kg) every 12 h beginning 48 h before the experiment did not significantly change the response to the three agonists when compared with untreated thyroidectomized rats. A similar treatment regimen has been reported to reverse the effects of adrenalectomy on hepatic beta adrenergic receptors (Wolfe et al. 1976).



-log10 (Agonist Concentration) Molar

FIGURE 3.4 The effect of cortisol treatment on the activation of glycogen phosphorylase a by adrenaline, isoproterenol and phenylephrine in hepatocytes from hypothyroid rats.

Cells were isolated from hypothyroid rats that had received intraperitoneal injections of cortisone acetate (50 mg/kg) every 12 h beginning 48 h before the experiment. Isolation, incubation of hepatocytes and assay of glycogen phosphorylase a activity were done as described in the caption to figure 3/2. Open symbols and filled symbols represent the means of 3 to 8 separate cell preparations from hypothyroid and hypothyroid, cortisol-treated rats Data for hypothyroid rats is the same, as that appearing in respectively. figure 3.2. Standard error bars have been omitted for clarity.

3.4 Accumulation of cAMP in response to stimulation by adrenergic agonists in hepatocytes from normal and thyroidectomized rats.

In order to further investigate the nature of the emergent beta receptor contribution to phosphorylase activation in thyroidectomized rats, cAMP accumulation in response to adrenaline, isoproterenol and phenylephrine was measured and compared with the response in hepatocytes from normal rats. Preliminary time-course experiments are shown in figure 3.5. Isoproterenol  $(10^{-6} \text{ M})$  produced a rapid (less than one min) increase in cAMP that remained at plateau for at least 5 min in hepatocytes from thyroidectomized rats. Similar results were obtained with normal hepatocytes except that magnitude of the response was very much less. In subsequent experiments hepatocyte suspensions were sampled for the determination of cAMP at 1.5 min after the addition of agonists.

Complete concentration response curves were produced for adrenaline, isoproterenol and phenylephrine in hepatocytes from normal and thyroidectomized rats (figure 3.6). Basal cAMP levels were not different for both groups: 1.85 + 0.76 pmol/mg protein for normal (n=3) and 2.25 + 0.93pmol/mg protein for thyroidectomized rats (n=3). All three agonists produced concentration-dependent increases in cAMP levels, however, in both groups of animals the magnitude of the response to adrenaline and isoproterenol was very much greater in hepatocytes from thyroidectomized rats; the response to isoproterenol stimulation was approximately 20-fold greater than in normal rats. The order of effectiveness in producing this response was isoproterenol >adrenaline > noradrenaline, which is consistent with a beta  $_2$  adrenergic receptor response in the thyroidectomized rat (figures 3.6 and 3.7). The difference in the maximum response to isoproterenol in the two groups could not be accounted for by a difference in metabolism of the drug since exposure was brief, and as shown in figure 3.8, the level of cAMP accumulation in response to isoproterenol was directly proportional to the density of hepatocyte

suspensions from both a thyroidectomized and normal rat.

In order to test the possibility that differential metabolism of cAMP by phosphodiesterase in the two groups was the basis for this difference, the response to isoproterenol was examined in the presence of isobutylmethylxanthine (IBMX), a phosphodiesterase inhibitor. Results of these experiments are summarized in table 3.2 and demonstrate that although cAMP accumulation in response to isoproterenol was enhanced in the presence of this inhibitor, the difference between the response in normal and thyroidectomized rat hepatocytes remained. A difference in phophodiesterase activity, therefore, cannot account for the above observations.

Comparison of the cAMP response to isoproterenol and adrenaline in the thyroidectomized rats showed that at maximal concentrations of these drugs, isoproterenol consistently produced greater cAMP accumulation than adrenaline. However, both of these drugs are are considered to be "full" beta receptor agonists. Adrenaline is also a "full" alpha receptor agonist, while isoproterenol has only a very slight effect on alpha receptors. This raised the possibility that the difference in their effectiveness could be due to an inhibitory influence of alpha receptor stimulation on cAMP accumulation as noted in the liver by others (section 1.3.3). To test this possibility the effect of phenoxybenzamine, an alpha receptor antagonist, on adrenaline-stimulated cAMP accumulation awas examined. As the results summarized in table 3.3 show, adrenaline-stimulated cAMP accumulation in hepatocytes was significantly enhanced after blockade of alpha receptors with phenoxybenzamine in the thyroidectomized rat. However, phenoxybenzamine blockade did not potentiate cAMP accumulation in response to adrenaline in the normal rat. This suggested that inhibitory alpha receptor activity was responsible for the difference in maximal cAMP accumulation induced by isoproterenol and adrenaline. Furthermore, such inhibitory alpha receptors appear to be active only in the hypothyroid and not in the normal rat.



FIGURE 3.5 Time-course of cAMP accumulation in response to isoproterenol in hepatocytes from normal and thyroidectomized rats.

Cells were isolated as described in section 2.3. After 30 min of incubation in a medium without added gelatin (time 0), 10  $\mu$ l of saline (broken line) or isoproterenol (solid line) was added to a 10 ml volume of cell suspension (approximately 40 mg/ml) to give a final concentration of  $10^{-6}$  M isoproterenol. At the indicated times, 1.0 ml aliquots of this suspension were homogenized in an equal volume of 10% trichloracetic acid and assayed for cAMP as detailed in section 2.5. Results of a single representative experiment are shown with cells from a normal and thyroidectomized rat.



FIGURE 3.6 Accumulation of cAMP in response to stimulation by adrenaline, isoproterenol and phenylephrine in hepatocytes from normal and thyroidectomized rats.

Cells were isolated as described in section 2.3 and resuspended in medium without gelatin. After 30 min of preincubation, drug was added in a 10  $\mu$ l volume to 3 ml of cell suspension to give the indicated final concentrations. After 1.5 min, 1.0 ml aliquots of this suspension were removed and assayed for cAMP as detailed in section 2.5. Results are the mean  $\pm$  s.e.m. for three separate cell preparations from thyroidectomized and normal rats (upper and right panel).





Cells were isolated as described in section 2.3. After 30 min of preincubation, adrenaline or noradrenaline was added in a 10  $\mu$ l volume of saline to 3.0 ml volume of cell suspension to give the indicated final agonist concentrations. After 1.5 min, 1.0 ml aliquots of this suspension were removed and assayed for cAMP as detailed in section 2.5. Results shown are of a single cell preparation from a thyroidectomized rat.



-<u>'</u>-

mg protein/ml cell suspension

FIGURE 3.8 Isoproterenol-stimulated cAMP accumulation as a function of tissue amount in hepatocytes from a normal and a thyroidectomized rat.

Cells were resuspended at approximately 80 mg/ml. These were then diluted wice by 50% to give three concentrations of cell material. Three ml aliquots of each of these concentrations were incubated and cAMP accumulation in response to  $10^{-6}$  M isoproterenol was determined as described in the caption to figure 3.6. Each data point reflects the value obtained from a single aliquot of cell suspension for cells isolated from a normal and a thyroidectomized rat.

Table 3.2

Effect of phosphodiesterase inhibitor (IBMX) on isoproterendiand glucagon-stimulated cAMP accumulation in hepatocytes from normal and thyroidectomized rats.

1. (	cAMP Accumulation (pmol/mg protein)		
· · ·	· · ·	· · · · · · · · · · · · · · · · · · ·	
, `` 	Normal	Thyroidectómized	
Control.	1.9±0.4	$1.6 \pm 0.3$	
+ IBMX	2.7 ± 0.4	<b>2.7</b> ± 0.5	
10 <sup>-8</sup> M Isoproterenol	$2.3 \pm 0.6$	3.8 ± 0.6	
+ IBMX	3.1 ± 0.6	8.1 ± 0.6 * ·	
10 <sup>-5</sup> M isoproterenol	$2.5 \pm 0.4$	10.0 ± 1.1 *	
+IBMX	$3.8 \pm 0.6$	42.8 ± 5.6 *	
10 <sup>-9</sup> M Glucagon	3.9±0.7	4.2 ± ].3	
+ IBMX	11.2 ± 2.9	15.2 ± 4.5	
10 <sup>-7</sup> M Glucagon	11.3 ± 1.8	24.5 ± 5.4 *	
+ IBMX	40.6±8.8	75.2 ± 12.4 *	
		Ŷ	

Cyclic AMP accumulation in response to stimulation with glucadon or isoproterenol at the indicated concentrations was determined as detailed in the caption to figure 3.10 and 3.12 except that  $10^{-4}$ M isobutylmethylxanthine (IBMX) or vehicle alone (2 µl ethanol/ml of cell suspension) was added 10 min before the addition of agonist. Values shown are the mean  $\pm$  s.e.m. for four to six cell preparations. \* indicates values significantly different from that obtained with normal hepatocytes.

Table	3.3	

Effect of phenoxybenzamine on adrenaline-stimulated cAMP ; accumulation in hepatocytes from a normal and a thyroidectomized rat.

	CAMP Accumulation	(pmol/mg protein)
· · · ·	° Control	Phenoxybenzamine
Thyroidectamized	· ·	•
Saline	2.03 ± 0.24	1.87 ± 0.28
10 <sup>-6</sup> M Adrenaline	3.75	6.50
10 <sup>-5</sup> M Adrenaline	5,25 ·± 0.40	* 7.10 ± 0.60
Normal	, ,	-
Saline	2.40	2.25
10 <sup>-6</sup> M Adrenaline	3.47	. <b>2.</b> 96
10 <sup>-5</sup> M Adrenaline	3.60	3.50
•	1	ן היין גו

Cyclic AMP accumulation in response to stimulation with adrenaline at the indicated concentrations was determined as detailed in the caption to figure 3.6 except that  $10^{-5}$ M phenoxybenzamine or vehicle alone (1 µl acidified ethanol/ml of cell suspension) was added 10 min before the addition of adrenaline (10 µl/ml). Values shown are the mean  $\pm$  s.e.m for quadruplicate incubations of a single cell preparation from a thyroidectomized rat or the results of duplicate determinations where values without standard errors are shown. Similar results were obtained in two more cell preparations from thyroidectomized and one from normal rats. \* indicates a significant difference between adjacent values.

81

- ×-

3.5 Glucagon-induced activation of glycogen phosphorylase and <u>cAMP</u> accumulation in hepatocytes from thyroidectomized and normal rats.

(--)

In order<sup>b</sup> to examine the selectivity of changes in beta adrenoceptor activity brought about after thyroidectomy, glucagon activation of phosphorylase was compared in normal and thyroidectomized rats. Figure 3.9 illustrates that glucagon produced an identical phosphorylase activation in hepatocytes from these two groups over the entire range of concentrations tested. Thus the action of a second glycogenolytic hormone, known to activate phosphorylase via a similar cAMP cascade (section 1.3.2) is not affected by thyroidectomy in the same way as beta receptor agonists are.

However, when cAMP accumulation in response to glucagon was examined, it was found to be significantly enhanced in hepatocytes from thyroidectomized rats, but only at relatively high glucagon concentrations (figure 3.10). At 10 nM glucagon, a concentration that was maximally effective in stimulating phosphorylase activity (figure 3.9), no difference in cAMP accumulation was observed (figure 3.10). This indicates that over the range of glucagon concentrations that can be considered physiologically important, there is no difference in cAMP accumulation when thyroidectomized rats are compared with normal animals. The addition of phosphodiesterase inhibitor (IBMX, table 3.2) enhanced glucagon-stimulated cAMP accumulation in both groups. The difference noted at higher glucagon concentrations in the absence of this inhibitor was even greater, suggesting that differential metabolism of cAMP in these two groups could not account for the observed difference.





Cells were isolated as described in section 2.3. After 30 min of incubation in a medium with added gelatin, glucagon (10  $\mu$ 1/ml) was added to give the indicated final concentrations. After 3 min 1 ml of cell suspension were removed, diluted into 5 ml of ice cold Krebs-Henseleit buffer, rapidly were removed and prepared for assay of phosphorylase <u>a</u> as detailed in section 2.4. Values shown are the mean <u>+</u> s.e.m. of four separate cell preparations from normal rats and five in the case of thyroidecomized rats.





Cells were isolated as described in section 2.3 and resuspended in medium containing gelatin. After 30 min of preincubation glucagon was added to give 'the indicated final concentrations. After 1.5 min, 1.0 ml aliquots of this suspension were removed and assayed for cAMP as detailed in section 2.5. Results shown are the mean  $\pm$  s.e.m. for six and eight separate cell preparations from thyroidectomized and normal rats respectively. \*indicates values significantly different from normal.

3.6 Effect of triiodothyronine replacement on isoproterenol-induced cAMP accumulation and glycogen phosphorylase activation in hepatocytes from thyroidectomized rats.

Figure 3.11 summarizes the effect of  $T_3$  replacement on isoproterenol-induced cAMP accumulation in thyroidectomized rats. Animals received 0.25 mg  $T_3$  per kg body weight by daily intraperitoneal injections for either two or four days. This dose of  $T_3$  is known to completely saturate hepatic nuclear thyroid hormone binding sites (Oppenheimer et al. 1978). As shown in figure 3.11, isoproterenol-induced cAMP accumulation was greatly enhanced in hepatocytes from thyroidectomized compared with normal rats. Thyroid hormone replacement for two days reduced the response somewhat, while after four days the effect of isoproterenol was not different from that of normal rat hepatocytes.

The effect of four days of treatment with  $T_3$  on isoprotorenol activation of glycogen phosphorylase is shown in figure 3.12. In agreement with earlier results (section 3.1), isoproterenol was significantly more effective in stimulating phosphorylase in hepatocytes from thyroidectomized rats when compared with normal. The response of hepatocytes from  $T_3$ -treated, thyroidectomized rats was returned partially, but not completely to normal levels, in contrast to the effect of four-day treatment on cAMP responses (figure 3.11). This is most likely due to the apparently greater sensitivity of o phosphorylase activation than cAMP accumulation to stimulation by glycogenolytic hormones (section 4.2).



FIGURE 3.11 Effect of triiodothyronine replacement on isoproterenolstimulated accumulation of cAMP in hepatocytes from thyroidectomized rats. Comparison with response in normal and untreated thyroidectomized rats.

Thyroidectomized rats received 0.25 mg  $T_3$  per kg by intraperitoneal injection for two or four days as indicated. Approximately 12 h after the last injection hepatocytes were isolated from these as well as normal and untreated, thyroidectomized rats as described in section 2.3. Cyclic AMP accumulation in response to the indicated isoproterenol concentrations was determined as detailed in the caption to figure 3.6. Results shown are the mean  $\pm$  s.e.m. for four cell preparations in the case of four day-treated rats and five for the others. \* indicates a significant difference from normal at a given concentration.



FIGURE 3.12 Effect of triiodothyronine replacement on activation of phosphorylase by isoproterenol in hepatocytes from thyroidectomized rats. Comparison with response in normal and untreated, thyroidectomized rats.

Cells were isolated from normal, thyroidectomized, and thyroidectomized rats treated for four days with  $T_3$  as described in the caption to figure 3.11. Cells were preincubated for 30 min after which time isoproterenol was added (10 µ1/m1) to give the indicated final concentrations. After 3 min phosphorylase was determined as described in the caption to figure 3.9. Values shown are the mean  $\pm$  s.e.m. for four cell preparations in each case.

3.7 Effect of phenylephrine on calcium efflux in hepatocytes from normal, thyrodectomized and thyrodectomized, triiodothyronine-treated rats.

As described above (section 3.1), thyroidectomy produced a marked decrease in the efficacy of both adrenaline and phenylephrine in activating glycogen phosphorylase. A likely explanation for the decreased efficacy of these two agonists is a decrease in alpha receptor activity. Stimulation of alpha receptors has been shown to lead to the release of calcium from isolated hepatocytes (section 1.3.3). Thus, to further investigate this possible decrease in alpha receptor activity, phenylephrine-stimulated calcium efflux from intact hepatocytes was examined by the use of a calcium-sensitive electrode as described by Chen et al. (1978).

In agreement with those authors, preliminary experiments (figure 3.13) showed that phenylephrine produced a concentration-dependent increase in calcium release from intact hepatocytes which could be blocked by the alpha antagonist phenoxybenzamine. The calcium inophore, A23187, induced the release of additional calcium, above that released by maximal phenylephrine concentrations (figure 3.13 A). Isoproterenol, at a concentration that is maximally effective in stimulating beta receptors, had no effect on calcium efflux in either normal or hypothyroid rats, whereas a second glycogenolytic hormone, vasopressin, produced significant calcium release even after blockade by phenoxybenzamine (figure 3.13 B and C). Figure 3.13 also shows that phenylephrine could be measured in a cumulative manner without the emergence of significant tachiphylaxis: responses to single doses were not significantly different from those to cumulative doses.

Hence, complete concentration-response curves were produced for phenylephrine in hepatocyte suspensions prepared from thyroidectomized rats and compared with triiodothyronine-treated, thyroidectomized rats and normal rats. These results are summarized in figure 3.14. As expected on the basis of results obtained for phenylephrine stimulation of glycogen phosphorylase,

thyroidectomy resulted in a marked suppression of the phenylephrine concentration-response curve for calcium efflux when compared to that for normal rats. Triiodothyronine treatment (0.25 mg/kg for four days) returned, the concentration-response curve of the thyroidectomized rats towards normal. The mean  $EC_{50}$  values for these three groups are shown in table 3.4. The decrease in efficacy of phenylephrine was accompanied by an approximately six-fold decrease in potency of this agonist after thyroidectomy.  $T_3$ -treatment also\*normalized this shift in potency. FIGURE 3.13 Effects of various glycogenolytic agents on calcium efflux from hepatocytes isolated from normal and thyroidectomized rats as measured by a calcium-sensitive electrode.

Hepatocytes were isolated (section 2.3) and prepared for determination of calcium efflux using a calcium-sensitive electrode as described in section 2.6. Tracing A shows the calcium efflux produced by incremental doses of phenylephrine in a suspension of cells from a normal rat. Adjacent are tracings showing the efflux produced by approximately equivalent single phenylephrine doses in separate suspensions of cells from the same cell preparation. A23187 caused the efflux of more calcium than a maximum dose of phenylephrine. Tracing B shows that isoproterenol, at a maximally effective glycogenolytic dose, produced no calcium efflux in a cell suspension from a second normal rat. Phenoxybenzamine completely inhibited phenylephrine-induced efflux, but had no effect on the vasopressin response. Tracing C is produced with cells from a thyroidectomized rat and shows that isoproterenol was again without effect on calcium efflux, while vasopressin produced a rapid calcium loss.

₩ ...





FIGURE 3.14—Phenylephrine-stimulated calcium efflux in hepatocytes from normal, thyroidectomized (Tx) and thyroidectomized,  $T_3$ -treated rats (Tx +  $T_3$ ).

Hepatocytes were isolated and prepared for determination of calcium efflux by calcium-sensitive electrode as described in the caption to figure 3.13. After a stable baseline was obtained, phenylephrine was added to give the cummulative concentrations shown. At the conclusion of each concentration-response curve, calibration pulses of calcium were added and the amount of calcium efflux calculated. Values shown are the mean  $\pm$  s.e., m. of five cell preparations from  $T_3$ -treated rats and six from each of the other groups. \* indicates values significantly different from those obtained with cells from normal rats. Mean  $EC_{50}$  values with standard errors are given in table 3.4.

.

Potency of phenylephrine and vasopressin in stimulation of glycogen phosphorylase and calcium efflux in hepatocytes from normal,

Table 3.4

thyroidectomized (TX). and thyroidectomized, triiodothyronine-treated

, SÞ. '	• •	ן ג ד	· · · · · · · · · · · · · · · · · · ·
v	Normal	ТХ	TX + T3
Phenvlephrine (µM)			
Phosphorylase	0.8 ± 0.3	6.0 ± 1.6* (4)	$1.3 \pm 0.7$ (4)
Calcium Efflux (Ca-free)	1.1±0.4 (6)	6.2 ± 1.7* (6)	$1.3 \pm 0.2$ (5)
<sup>45</sup> Ca-Efflux (2.5 mM Ca)	$1.6 \pm 0.7$ (4)	n.đ.	$1.6 \pm 0.3$ (4)
Vasopressin (nM)	r 0 -		1
Phosphorylase	.0.5 ± 0.2 (4)	$1.2 \pm 0.2$ (4)	n.đ. ,
Calcium Efflux (Ca-free)	1.9 ± 0.7 (5)	$5.1 \pm 1.4*$	n.đ.
45Ca-EffJux (2.5 mM Ca)	$0.5 \pm 0.1$ (4)	$3.0 \pm 0.4*$	n.d.

 $(Tx + T_2)$  rats.

Data are from the same experiments reported in figures 3.14, 3.15, 3.19 and 3.20. Each value is the mean  $\pm$  s.e.m. of the EC<sub>50</sub> value for the number of experiments shown in parentheses. Phosphorylase activities were determined with 2.5 mM extracellular calcium as was  $^{45}$ Ca efflux. Calcium efflux (measured by calcium-sensitive electrode) was determined in nominally calcium-free medium. \* indicates values significantly different from that obtained in normal rats. For details about the measurement of these responses in hepatocyte suspensions refer to the legends to the above-indicated figures.

92

A. 1
3.8 The effect of vasopressin on calcium efflux from hepatocytes of normal and thyroidectômized rats.

Vasopressin has been shown to activate hepatic glycogen phosphorylase by a cyclic AMP-independent mechanism (Kirk and Hems 1974). Similar to alpha adrenergic agonists, vasopressin stimulates calcium release from isolated hepatocytes (Blackmore et al. 1978; Chen et al. 1978) and the accompanying transient increase in cytosolic calcium is believed to mediate the activation of glycogen phosphorylase (section 1.3.3). Since the mitochondrial pool from which calcium is released and the post-receptor mechanisms for phosphorylase activation are thought to be the same for alpha adrenergic agonists and vasopressin (section 1.3.3), vasopressin was used to test whether the influence of hypothyroidism on the responses to phenylephrine is selective for alpha The results of the effect of vasopressin-induced adrenergic stimulation. calcium efflux from hepatocytes of normal and thyroidectomized rats are shown in figure 3.5. Although the response to vasopressin is somewhat suppressed in the thyroidectomized rat this difference approaches, but does not attain statistical significance. Comparison of this change with that for phenylephrine (figure 3.14) reveals a different pattern of change with a more significant suppression for the alpha receptor agonist. The  $EC_{50}$  for the vasopressin response is shifted 2.7-fold to the right (table 3.4) compared with the six-fold shift observed for phenylephrine after thyroidectomy (section 3.7).





Experimental details are as described in the caption to figure 3.14. Each data point is the mean  $\pm$  s.e.m. of five and six experiments on separate cell preparations from normal and thyroidectomized rats respectively. The differences in vasopressin-stimulated calcium efflux in the two groups of rats approach statistical significance at some points. Mean EC<sub>50</sub> values with standard errors are given in table 3.4.

3.9 Effect of calcium ionophore A23187, on calcium efflux and glycogen phosphorylase activation in hepatocytes from normal and thyroidectomized rats.

Like vasopressin, the divalent cation ionophore, A23187, has been shown to activate glycogen phosphorylase by a cAMP-independent mechanism similar to alpha agonists (Blackmore et al. 1978). Accordingly, the effect of the ionophore on glycogen phosphorylase activation was compared in thyroidectomized and normal rats. Figure 3.16 shows that the ionophore produced a dose-dependent increase in phosphorylase <u>a</u> activity in hepatocytes from normal rats that was not altered after thyroidectomy. The slight increase in phosphorylase activity at lower A23187 concentrations in the thyroidectomized group was most likely due to an increased sensitivity of this group to the vehicle, ethanol. This is supported by the fact that phosphorylase activity is significantly greater in thyroidectomized rats when alcohol alone was added (Zero A23187 concentration).

The effect of A23187 on calcium efflux was also examined in hepatocytes from normal and thyroidectomized rats. These results are summarized in figure 3.17 and show that the concentration-dependent, A23187-induced calcium efflux is identical in the two groups of animals.



A23187 Concentration (micromolar)

FIGURE 3.16 Activation of glycogen phosphorylase by calcium inophore, A23187, in hepatocytes from normal and thyroidecomized rats.

Cells were isolated as described in section 2.3. After 30 min of incubation in a medium with added gelatin, A23187 or vehicle alone (zero concentration, 2  $\mu$ l ethanol/ml) was added to give the indicated final concentrations. After 3 min 1 ml of cell suspension were removed, diluted into 5 ml of ice cold Krebs-Henseleit buffer, rapidly centrifuged and prepared for assay of phosphorylase <u>a</u> as detailed in section 2.4. Values shown are the mean <u>+</u> s.e.m. of four separate cell preparations in each group.



A23187 Concentration (micromolar)

FIGURE 3.17 Stimulation of calcium efflux in hepatocytes from normal and thyroidectomized rats by calcium ionophore, A23487.

Experimental details are as described in the caption to figure 3.14. A23187 or vehicle (2-50  $\mu$ l ethanol/15 ml of cell suspension) were added to give the indicated cumulative concentration of drug. Each data point is the mean  $\frac{+}{-1}$  s.e.m. of six experiments on separate cell preparations from normal and thyroidectomized rats.

3.10 Phenylephrine-induced activation of glycogen phosphorylase and  $^{45}$ Ca efflux in hepatocytes from normal, thyroidectomized and thyroidectomized, triiodothyronine-treated rats.

The calcium efflux experiments described above (sections 3.7, 3.8 and 3.9) were made with hepatocyte suspensions in nominally calcium-free medium. Figure 3.18 shows that phosphorylase activation by either phenylephrine or vasopressin is significantly suppressed under these conditions. Similar results have been obtained by others (Stubbs et al. 1976; Keppens et al. 1977; Van de Werve et al. 1977). Thus in order to rule out the possibility that differences in phenylephrine-stimulated calcium efflux observed when normal and thyroidectomized rats were compared (section 3.7) were due to low calcium conditions, <sup>45</sup>Ca efflux was examined at physiological calcium concentrations.

Figure 3.19 A shows that, in agreement with the results described in section 3.7, the effect of phenylephrine in activating phosphorylase was suppressed in the hypothyroid liver: the maximal response was reduced by 37% and the  $EC_{50}$  increased <sup>/</sup>8-fold (table 3.4). Euthyroid control and thyroidectomized,  $T_3$ -treated values were similar (table 3.4) indicating that the effect of hypothyroidism was reversed by  $T_3$  replacement (figure 3.19).

In support of the findings with phosphorylase and calcium electrode efflux experiments (section 3.7), figure 3.19 B shows that phenylephrine-stimulated  $^{45}$ Ca release was markedly suppressed in cells from thyroidectomized rats compared with controls: the maximal response was reduced by 82% so<sup>1</sup> that reliable estimates of EC<sub>50</sub> could not be obtained in these preparations. Nevertheless, the increase in  $^{45}$ Ca release over baseline was only significant at  $10^{-4}$ M phenylephrine, which is 100 times higher than the effective threshold concentration in controls. T<sub>3</sub> treatment of thyroidectomized rats again partially corrected this suppression (figure 3.19). Thus the effect of thyroidectomy on phenylephrine-stimulated calcium efflux described in section 3.7 was confirmed at the same ambient calcium concentrations at which

phosphorylase responses were studied.

3.11 Vasopressin-induced activation of glycogen phosphorylase and  $^{45}Ca$  efflux in hepatocytes from normal and thyroidectomized rats.

].

The data in figure 3.20A illustrate that, in contrast to phenylephrine (figure 3.19A), the maximal response to vasopressin was not influenced by hypothyroidism, but there was a small, 2.4-fold increase in  $EC_{50}$  (table 3.4) that approached statistical significance. The maximal <sup>45</sup>Ca release by vasopressin was similarly unaffected, whereas the  $EC_{50}$  increased 6-fold (table 3.4). Thus, hypothyroidism caused a moderate reduction in the potency of vasopressin, but the effects on phenylephrine were much more substantially reduced.



FIGURE 3.18 Effect of extracellular calcium concentration on vasopressin and phenylephrine activation of glycogen phosphorylase in hépatocytes from a normal rat.

57

Cells were isolated as described in section 2.3. Half of the yield from a single isolation was resuspended in nominally calcium free medium as for calcium electrode studies (section 2.6), while the other half was resuspended in medium containing normal calcium (both with gelatin). After 30 min of incubation vasopressin or phenylephrine was added to give the indicated final concentrations. After 3 min 1 ml of cell suspension was removed, diluted into 5 ml of ice cold Krebs-Henseleit buffer, rapidly centrifuged and prepared for assay of phosphorylase <u>a</u> as detailed in section 2.4. Values shown are the mean of duplicate determinations with cells from a single rat. There was no difference in viability of cells determined at the end of incubation. The **experiment** was repeated once with the same result.



11/1

FIGURE 3.19 Phenylephrine stimulation of phosphorylase (A) and  $^{45}$ Ca efflux (B) in hepatocytes from normal, thyroidectomized and thyroidectomized,  $T_3$ -treated rats.

Hepatocytes were isolated and incubated as described in section 2.3. One m1 aliquots were taken for the determination of phosphorylase <u>a</u> 3 min after the addition of phenylephrine or saline. Determination of  $^{45}$ Ca efflux was made at zero time (see section 2.5) and 3 min after addition of phenylephrine or saline. 100% represents the difference in radioactivity in the medium from zero time to 3 min in the presence of saline alone. Each data point is the mean <u>±</u> s.e.m. of four experiments on seperate cell preparations. \* indicates values significantly different from those obtained with thyroidectomized rats. Mean EC<sub>50</sub> values with standard errors are given in table 3.4.





Experimental details are as described in the caption to figure 3.19. Each data point is the mean  $\pm$  s.e.m. of four experiments on separate cell preparations. At no concentration of vasopressin are values in thyroidectomized rats significantly different from controls. Mean EC<sub>50</sub> values with standard errors are given in table 3.4.

 $10\,2$ 

3.12 Results of [<sup>3</sup>H]prazosin binding to plasma membranes isolated from normal, thyroidectomized and thyroidectomized, triiodothyronine-treated rat livers.

Activation of phosphorylase and stimulation of calcium release in the normal rat liver are known to occur via stimulation of  $alpha_1$  adrenoceptors (section 1.3.3). In view of the effects of hypothyroidism on these responses, the specific binding of  $[^{3}H]$  prazosin, a selective  $alpha_1$  receptor antagonist, to liver plasma membranes from normal, thyroidectomized and thyroidectomized,  $T_3$ -treated rats was examined. Previous studies have shown that  $[^{3}H]$  prazosin is a useful probe for  $alpha_1$  receptors in various tissues (Hornung et al. 1979; Barnes et al. 1979; Karliner et al. 1979) including the liver (Hoffman et al. 1981; Geynet et al. 1981). Figure 3.21 illustrates that the binding of  $[^{3}H]$  prazosin is saturable in membranes from all three groups. Specific binding was 80 to 85% at saturating  $[^{3}H]$  prazosin concentrations.

Binding of  $[{}^{3}$ H] prazosin to liver plasma membranes from normal rats was rapid and reversible. Figure 3.22 illustrates the kinetics of specific  $[{}^{3}$ H] prazosin binding. From the time-course of association, pseudo-first order association plots were produced and the apparent association rate constant.  $K_{ob}$ determined by linear regression analysis (Williams and Lefkowitz 1978). Dissociation initiated by the addition of excess phentolamine  $(10^{-5}$ M) was first-order (figure 3.22) and the dissociation rate constant,  $k_2$ , could be determined by linear regression analysis. From these two values the kinetically derived dissociation constant,  $K_d$ , of  $[{}^{3}$ H] prazosin could be calculated as  $K_d = k_2/k_1$ , where  $k_1 = (K_{ob} - k_2)/200$ , 200 being the concentration (pM) of  $[{}^{3}$ H] prazosin in the assay (Williams and Lefkowitz 1978). The mean  $\pm$  s.e.m. of  $k_1$  and  $k_2$  thus determined in three seperate experiments are 6.6  $\pm$  0.4 x  $10^{10}$  M<sup>-1</sup>min<sup>-1</sup> and 0.031  $\pm$  0.004 min<sup>-1</sup> respectively. The resulting kinetically derived  $K_d$  value of 41.9  $\pm$  8.2 pM was close to the equilibrium  $K_d$  obtained from saturation experiments (table 3.5). Both these values agree well with the

apparent dissociation constant for prazosin determined by Schild plot for antagonism of the phosphorylase response to adrenaline in intact hepatocytes  $(89 \pm 11 \text{ pM}^1)$ .

In order to determine the maximum binding capacity  $(b_{max})$  and the affinity of [<sup>3</sup>H] prazosin in purified liver plasma membranes from each group of animals, steady state binding data were analyzed by the method of Scatchard (1949). Typical Scatchard-plots are illustrated in figure 3.23 and the cumulated results from 'several experiments in each group are shown in table 3.5. These indicate that receptor density was reduced by more than 40% in thyroidectomized preparations, but reversed to near normal levels after  $T_3$ -treatment. There was no change in the  $K_d$  of [<sup>3</sup>H] prazosin. The Hill-plots in figure 3.23 indicate that binding of [<sup>3</sup>H] prazosin was to a single population of non-écoperative sites ( $n_H$ : 0.92 to 1.03).

In order to determine if thyroidectomy had any effect on the affinity of adrenergic agonists for the  $alpha_1$  receptor site, competition experiments were made with increasing concentrations of agonists and [<sup>3</sup>H] prazosin (0.2 nM, figure 3.24 and table 3.5). The K<sub>d</sub> of agonists was determined by the method of Cheng and Prusoff (1973):

$$K_{d} = \frac{IC_{50}}{1 + \frac{0.2}{K^{*}}}$$

where  $IC_{50}$  is the concentration of agonist producing 50% suppression of  $[^{3}H]$  prazosin and K\* is the equilibrium dissociation constant of  $[^{3}H]$  prazosin.

<sup>1</sup>Experiments done by W.H. Kan using low hepatocyte concentrations (1 mg/ml) to avoid significant uptake of prazosin by liver cells (section 1.3.1).

The data in table 3.5 show that the affinities of agonists were unaffected by thyroid state, except for a small decrease in the  $K_d$  for phenylephrine in thyroidectomized,  $T_3$ -treated rats. Gpp(NH)p, which is known to influence the affinity of adenylate cyclase coupled receptors (Limbird 1981), did not alter the  $K_d$  for adrenaline in any of the groups at either 30 or 100  $\mu$ M (table 3.5). Figure 3.24 illustrates the stereoselectivity of [<sup>3</sup>H] prazosin binding sites in preparations from normal rats, as well as the absence of effect of 100  $\mu$ M Gpp(NH)p on 1-adrenaline suppression of [<sup>3</sup>H] prazosin binding in any of the three groups.



FIGURE 3.21 Saturation curves for  $[{}^{3}H]$  prazosin binding in liver plasma membranes isolated from normal, thyroidectomized and thyroidectomized, T<sub>3</sub>-treated rats.

Plasma membranes were prepared from freshly excised livers as described in section 2.9. Binding of  $[{}^{3}H]$  prazosin in the absence (total binding, upper curves) and presence (nonspecific binding, lower lines) of 2 µM phentolamine was determined as described in section 2.10. Specific binding (middle curves) was calculated by taking the difference between total and nonspecific binding. One representative experiment from each group of rats is shown. Means of several similar experiments are given in table 3.5.

2



FIGURE 3.22 Kinetics of specific binding of  $[{}^{3}H]$  prazosin to purified plasma membranes from a normal rat liver.

Membranes were incubated with 200 pM [<sup>3</sup>H] prazosin in the absence or presence of 2  $\mu$ M phentolamine. A: time-course for specific binding of [<sup>3</sup>H] prazosin during association (**•**) and dissociation (O) after the addition of 10  $\mu$ M phentolamine (**↓**). B: a pseudo-first order association plot whose slope (r=0.98) represents the apparent association rate constant,  $K_{ob} = 0.153 \, \text{mm}^{-1}$ , C: a first-order plot of dissociation, where the slope (r=0.98) is the first order dissociation rate constant,  $k_2 = 0.033 \, \text{min}^{-1}$  The kinetically derived  $K_d$  (section 3.12) was 55 pM and the equilibrium  $K_d$  derived from a Scatchard plot was 52 pM in the same membrane preparation. The mean <u>+</u> s.e.m<sub>0</sub> of these values determined in three separate experiments are given in section 3.12.





Liver plasma membranes were prepared from freshly excised, whole livers of normal (---), thyroidectomized (----) and thyroidectomized,  $T_3$ -treated rats ( $---\Delta$ ) as described in section 2.9. Equilibrium binding of  $[^3H]$  prazosin was assayed as detailed in section 2.10. A: Scatchard plots from the same three experiments shown in figure 3.21. The K<sub>d</sub> values were 7.7.4 (pM) ( $^{\circ}$ ), 98.4 pM ( $^{\circ}$ ) and 64.6 pM ( $^{\circ}$ ). B: Hill plots from the same, experiments, with n<sub>H</sub> values of 1.03 ( $^{\circ}$ ), 0.92 ( $^{\circ}$ ) and 0.93 ( $^{\circ}$ ). Data shown are from the same experiments in figure 3.21.

Tabl	e	3	•	5

The effect of thyroid state on hepatic alpha<sub>l</sub> receptors labelled by [<sup>3</sup>H]prazosin.

	, Normal.	Tx	<sup>Tx</sup> + <sup>T</sup> Z
b <sub>max</sub> (fmol/mg protein)	567 ± 25 * (5)	326 ± 51 * (8)	498 ± 23 (5)
K <sub>d</sub> [ <sup>3</sup> H]prazosin (pM)	69.8±6.1 (5)	53.7 ± 11.9 (8)	61.6 ± 10.2 (5)
$K_{d}$ l-adrenaline ( $\mu$ M)	0.68 ± 0.17 (9)	0.67 ± 0.15 (7)	0.89 ± 0.11 (6)
+ 30 M Gpp (NH)p	1.06 ± 0.45 · (3)	$0.73 \pm 0.36$ (3)	0.98 ± 0.28 (3)
+ 100 M Gpp (NH) P	0.73 ± 0.23 (3)	$0.76 \pm 0.39$ (3)	$0.94 \pm 0.16$ (3)
$\kappa_d$ l-phenvlephrine (µM)	2.58 ± 0 <sup>2</sup> .15 (3)	2.56 ± 0.40 * (3)	4.05 ± 0.10 (3)
و			

Liver plasma membranes were incubated with  $[{}^{3}H]$  prazosin in the absence or presence of competing ligands as described in section 2.10. The K<sub>d</sub> and  $b_{max}$  of  $[{}^{3}H]$  prazosin was determined by Scatchard analysis of specific binding as described in the caption to figure 3.23. In competition experiments with agonists, of  $[{}^{3}H]$  prazosin was 0.2 nM and the K<sub>d</sub> of agonists was determined as described in section 3.12. Means  $\pm$  s.e.m are 'shown with the number of experiments in parentheses.' \* indicates a significant difference between adjacent values.





Binding assays were done as described in section 2.10. Assay tubes contained  $[{}^{3}H]$  prazosin (0.2 nM) in the absence (Bo) or presence (B) of various Data points represent the means of three concentrations of adrenaline. separate experiments. In each preparation, l-adrenaline suppression of binding was tested in the absence (filled symbols, solid lines) and presence (open symbols, dashed lines) of 100  $\mu$ M Gpp(NH)p. d-Adrenaline suppression in the absence of Gpp(NH)p was tested in controls (D). The mean  $K_{d}$  values in the absence and presence of Gpp(NH)p (calculated as described in section 3.12), respectively, were 0.88 + 0.14 and 0.73 + 0.23  $\mu$ M in controls, 0.60 + 0.12 and 0.73  $\pm$  0.39  $\mu\text{M}$  in Tx, and 1.02  $\pm$  0.25 and 0.94  $\pm$  0.16  $\mu\text{M}$  in The n<sub>H</sub> values of corresponding Hill plots (95% confidence limits)  $Tx + T_{2}$ . were 0.77-0.87 and 0.68-0.82 (controls); 0.70-0.84 and 0.75-0.89 (Tx); 0.65-0.75 and 0.82-0.98 (Tx + T<sub>3</sub>).

Ð

( :

## SECTION FOUR

١.

## DISCUSSION

4.1 Géneral Considerations: The effect of altered thyroid state on hepatic adrenergic receptors.

The observations presented here indicate a change in the adrenoceptormediated activation of hepatic glycogen phosphorylase from an alpha adrenergic response in hepatocytes from normal rats to a predominantly beta adrenergic effect in hepatocytes from hypothyroid rats. This was shown by an increase in both the potency and efficacy of the selective beta receptor agonist, isoproterenol, and by the ability of the beta receptor antagonist propranolol to reduce activation of glycogen phosphorylase by adrenaline in hepatocytes from hypothyroid, but not normal rats. Alpha receptor antagonists, which were effective in euthyroid rats, were without effect in hepatocyte preparations from thyroidectomized rats. As well, there was a decrease in the efficacy of adrenaline and phenylephrine, a relatively selective alpha receptor agonist, in the activation of phosphorylase in cells prepared from hypothyroid rats.

Changes in thyroid state are known to influence the activity of enzymes involved in catecholamine metabolism and altered disposition of agonists could conceivably be involved in the observed changes in their effectiveness. Results of studies in which this question has been addressed are variable and often contradictory (see section 1.2.2). The effect of thyroid state on the activities of these enzymes specifically in rat liver has been extensively studied in numerous preparations, but no one to date has examined this problem using collagenase dispersed hepatocytes. Zimon et al. (1976) examined the metabolism of <sup>14</sup>C-labelled noradrenaline in isolated perfused livers from hypothyroid, euthyroid and hyperthyroid rats. They inferred from their results that both COMT and MAO activities are slightly decreased in hypothyroid rat livers, while in hyperthyroidism increased metabolism is observed. Sourkes et al. (1977) examined the metabolism of a series of  $^{14}$ C-labelled monoamines in whole liver homogenates and reported reduced MAO activity in

hyperthyroidism with no change after thyroidectomy. Note that the observations of neither of these studies may be directly applicable to experiments with isolated parenchymal cells; it is not clear what the relative importance of other tissue elements in catecholamine metabolism may be. Even so, it is unlikely that altered metabolism of drugs used in this study could account for the apparent differences in adrenoceptor activity between groups of rats for the following reasons.

(1) The amount of tissue relative to extracellular medium was lower in the present experiments than in those described above making it unlikely that concentrations of agonists were significantly depleted, particularly since phosphorylase activity was determined after only 3 min of stimulation.

(2) Adrenergic agonists and antagonists are metabolized by very different pathways (Weiner 1980). That enzyme activities would be altered in such a manner as to produce compatible changes in agonist and antagonist actions would be an unlikely coincidence.

(3) When beta receptor activity was evaluated by comparing cAMP accumulation in normal and thyroidectomized rats (section 3.4), the magnitude of the response was found to vary directly with the amount of cell material present.

(4) The major change in the hepatic adrenoceptor response pattern was observed in the hypothyroid state where either no change in enzyme activity or a similar small decrease in both MAO and COMT was noted (see above). These observations cannot explain the opposite changes in the effectiveness of phenylephrine, which is a substrate for MAO, and isoproterenol, which is preferentially metabolized by COMT (Weiner 1980).

The increase in beta receptor activity in hepatocytes from hypothyroid rats has been confirmed by others (Malbon et al. 1978a; Malbon et al. 1980; Malbon 1980a). Malbon (1980a) has extended these studies by examining both isoproterenol-stimulated adenylate cyclase activity and specific binding of

 $[^{125}]$  iodohydroxybenzylpindolol ( $[^{125}I]$  IHBP), a selective beta receptor antagonist, to plasma membranes purified from hepatocytes isolated from euthyroid, hypothyroid and hypothyroid,  $T_3$ -treated rats. As expected on the basis of the enhanced beta receptor-mediated stimulation of phosphorylase and cAMP accumulation in isolated hepatocytes, beta receptor activation of plasma membrane adenylate cyclase was increased in hypothyroid rats (Malbon 1980a). The specific binding of  $[^{125}I]$  IHBP was also increased in preparations from hypothyroid rats and Malbon (1980a) suggested that this increase in the number of putative beta receptor sites was the major factor responsible for enhanced beta adrenergic activity.

Malbon and Fain (1978) also described findings of a marked reduction in the ability of adrenaline and phenylephrine to activate glycogen phosphorylase in hypothyroid rat hepatocytes suggestive of a reduced alpha adrenergic activity. However, in subsequent reports (Malbon et al. 1978a, 1980), this group did not confirm the decreased alpha receptor contribution in hypothyroid hepatocytes described here and in their earlier work (Malbon and Fain 1978). Some possible reasons for this discrepancy will be discussed later (section 4.3).

Hepatic gluconeogenesis is regulated by adrenergic receptors in a manner analogous to the glycogenolytic response since both beta and alpha receptors are involved, the latter being more important in the rat (Tolbert et al. 1973; Kneer et al. 1974; Hutson et al. 1976). The effects of adrenergic agonists and antagonists on <u>gluconeogenesis</u> from lactate have been examined in perfused hypothyroid rat livers (Hagino and Nakashima 1973, 1974). The findings of those studies suggested that gluconeogenesis is mediated by both alpha and beta receptors in normal rats and that the contribution of beta receptors to this response becomes less important in hypothyroidism. Those results appear to contradict the present fundings in isolated hepatocytes as well as studies that have clearly shown catecholamine-mediated gluconeogenesis in the normal rat to be predominantly an alpha receptor response (see above). The fundings of

Hagino and Nakashima (1973, 1974) are difficult to evaluate; agonists were used at only single doses and neither alpha nor bota antagonists were effective in normal rat livers even at very high concentrations (0.4 mM). Moreover, in a later study, Hagino and Shigei (1976) found that cAMP accumulation in response to in vivo injection of catecholamine was enhanced in livers of hypothyroid rats. This latter finding contradicts their conclusions and is more compatible with the interpretation of the present discussion: an increased beta receptor activity in hypothyroidism. However, it is important to bear in mind that conclusions regarding adrenergic receptors made on the basis of the glycogen phosphorylase activation response alone may not be extrapolated to other adrenergie responses in the liver without caution (see section 1.3.3 for a list of these other responses). This is particularly true of the gluconeogenic response, which probably involves more than one enzymatic step under adrenergic control. This point is discussed in more detail in section 4.5.

G

The effect of hypothyroidism on hepatic adrenoceptors developed slowly and was not apparent at two weeks after surgical thyroidectomy even though serum thyroxine levels were as low as those seen at 10 to 14 wk. Malbon and coworkers identified similar effects on hepatic beta adrenoceptors after 18 to 25 days of iodine-deficiency and propylthiouracil administration (Malbon et al. 1978a; Malbon 1980a); however, effects on alpha adrenergic receptors were not apparent after this shorter period of thyroid deficiency (Malbon et al. 1978a). Although there may be other reasons why these latter investigators could not identify changes in the alpha receptor response (secton 4.3), these findings suggest that beta receptor function may be more sensitive to serum thyroid hormone levels than alpha. On the other hand, the effects of thyroid deficiency were almost completely reversed by only four days of T<sub>3</sub> treatment in the present study and two days in that of Malbon (1980a). Taken together, these results suggest that both alpha and beta receptor functions have a relatively high sensitivity to regulation by thyroid hormones; detectable changes

in these receptors are likely to occur at low hormone levels.

The effect of thyroid hormone excess was also examined in the present study. Hyperthyroidism, induced by the administration of thyroxine, produced a similar decrease in the efficacy of adrenaline and phenylephrine to that seen in hypothyroid rat hepatocytes. Basal phosphorylase activity was unaltered and the slight stimulatory effect of a high concentration of isoproterenol (probably an alpha receptor-mediated effect) was not changed. These findings could suggest that hyperthyroidism in the rat results in a decrease in alpha adrenoceptor activity, without a concomitant change in the beta receptor contribution to the phosphorylase response in the hyperthyroid rat.

Similar findings were reported by Malbon and LoPresti (1981) except that those authors found a 70% reduction in basal phosphorylase activity in hepatocytes from hyperthyroid rats. In agreement with the present findings, adrenaline stimulation of phosphorylase was suppressed in hyperthyroid rats. Malbon and LoPresti (1981) extended their study by demonstrating an increased phosphorylase phosphatase activity and decreased number of [<sup>3</sup>H] DHE binding sites in plasma membranes from hyperthyroid rats. They suggested that these effects could account for the attenuated phosphorylase response. In the preparations studied by Malbon and LoPresti (1981), there was a significant beta receptor contribution to phosphorylase activation even in the euthyroid rat and this was also suppressed in hyperthyroidism. That this change was not detected in the present study may be be related to the fact that the beta receptor component of adrenergic activation of phosphorylase is more important in female as compared to male rats (Studer and Borle 1982); Malbon and LoPresti (1981) used female rats, whereas only males were used in the present investigations. In a second study, Malbon and Greenberg (1982) found that the reduced beta receptor activity inhyperthyroid hepatocytes could be localized to the hormone-sensitive adenylate cyclase. The properties of the beta receptor-adenylate cyclase complex were altered at several loci by thyroid

hormone excess, including a 35% reduction in the number of  $[^{125}I]$  IHBP binding sites. The effect of hyperthyroidism on hepatic adrenergic receptors was not further investigated in the present study.

## 4.2 The effect of hypothyroidism on hepatic beta adrenergic receptors. Comparison with glucagon.

The emergent beta adrenergic contribution to glycogen phosphorylase activation in hypothyroid rat hepatocytes was found to be associated with an increased accumulation of cAMP (section 3.4). Thus, the difference between normal and thyroidectomized rats is apparent in production of the second messenger, an early step in the glycogenolytic cascade. However, decreased metabolism of cAMP in hepatocytes from hypothyroid rats could also contribute @ to this enhanced response. As described in section 1.2.3, beta receptor-mediated stimulation of lipolysis and cAMP accumulation in adipose tissue is suppressed in hypothyroidism and at least part of this suppression is thought by some workers to be due to an increased activity of the particulate high affinity form of phosphodiesterase in fat cells from hypothyroid rats (Armstrong et al. 1974). Accordingly, theophylline, a phosphodiesterase inhibitor, enhanced adrenaline-stimulated lipolysis and cAMP accumulation to a greater extent in adipocytes from hypothyroid than normal rats (Goswami and Rosenberg 1978; Van Inwegen et al. 1975). In the present experiments on isolated hepatocytes, the difference in isoproterenol-stimulated cAMP accumulation between thyroidectomized and normal rats became even more pronounced in the presence of a phosphodiesterase inhibitor, isobuty Imethylxanthine (section 3.4). Thus, unlike the situation in adipocyte preparations, the differences described could not be attributed to changes in phosphodiesterase activity. Furthermore, Gumma et. al. (1977) have directly examined the effect of thyroidectomy on cyclic nucleotide phosphodiesterases

in rat liver. They found that the activities of both the high and low affinity phoshodiesterases were increased after thynoidectomy. Hence, on this basis one would "have expected a decreased cAMP accumulation in response to hormonal stimulation. However, the present findings\show a clearly increased response Gumma et al. (1977) point out that these changes in after thyroidectomy. hepatic cAMP phosphodiesterase activity parallel those found in adipose tissue from 'hypothyroid rats and also suggest that the observed changes in phosphodiesterase activity in adipose tissue account (at least in part) for the decreased cAMP and lipolytic response of these animals. The results presented here show that this is not a general property of adenylate cyclase/cAMPdependent systems. Furthermore, one must keep in mind the fact that findings in unstimulated tissues may not reflect the true situation during hormonal activation: there is some evidence that cAMP phosphodiesterase activity may itself be subject to hormonal influences (Francis and Kono 1982). In addition, because of the possible compartmentalization of the cAMP rise and cAMP metabolism that has been postulated to exist in some tissues (Brunton et al. 1981), homogenates of whole tissue may not reflect accurately physiologic mechanisms at work in the cell.

The enhanced isoproterenol-stimulated cAMP accumulation in hepatocytes from hypothyroid rats was reduced towards normal levels by in vivoadministration of  $T_3$  (section 3.6). Although four days of  $T_3$  treatment resulted in a cAMP response that was indistinguishable from that observed in normal rat hepatocytes, isoproterenol-stimulated phosphorylase activity, while less than that seen in hypothyroid hepatocytes, was still enhanced above normal (section 3.6). A similar dissociation of the cAMP response and phosphorylase activation has been noted under other circumstances (Birnbaum and Fain 1977) and can be explained by the fact that very small, often inapparent, changes in cAMP are capable of producing marked increases in phosphorylase activity (see below).

Very similar results to those presented here have been simultaneously

reported by Malbon and co-workers<sup>1</sup>. Beta receptor-mediated activation of glycogen phosphorylase and stimulation of cAMP accumulation was enhanced in hepatocytes from hypothyroid rats (Malbon et al. 1978a). This difference was not abolished in the presence of IBMX (Malbon et al. 1978a). A later study showed that beta adrenergic activation of adenylate cyclase was enhanced in homogenates and purified plasma membranes prepared from hypothyroid rat hepatocytes when compared to euthyroid controls (Malbon 1980a). Basal and fluoride-stimulated cyclase activities were unchanged. Specific binding of [<sup>125</sup>I] IHBP, a potent beta adrenergic antagonist, was increased in those Both the increased [<sup>125</sup>I]IHBP binding and adenylate cyclase preparations. activity were reversed by two days of in vivo T<sub>3</sub>-pretreatment of hypothyroid rats,

Conversely, Jones et al. (1972) had earlier reported no change in basal and adrenaline-stimulated adenylate cyclase activity in hypothyroid rat liver homogenates, but an increased fluoride-stimulated activity in these preparations. Reasons for this discrepancy are not entirely clear. Differences cannot be accounted for by the type of preparation used in each study; Malbon (1980a) also examined fluoride-stimulated adenylate cyclase in whole liver homogenates and still found the same activity in hypothyroid and euthyroid rats. One possible reason for the similarity of agonist-induced adenylate cyclase activation in these two groups observed by Jones et al. (1972) could be due to the use of adrenaline, a mixed alpha/beta receptor agonist, rather than isoproterenol, a pure beta receptor agonist. The presence of inhibitory alpha activity (Jard et al. 1981; Assimacopoulos-Jeannet et al. 1982) could have masked an increase in beta adrenergic activation of adenylate cyclase.

<sup>1</sup>The report by Malbon et al. (1978a) appeared on Dec. 25, 1978 and our own study (Preiksaitis and Kunos 1979) on Jan. 1, 1979.

Although in the present study both adrenaline and isoproterenol stimulation resulted in greatly enhanced cAMP accumulation in hypothyroid compared to normal hepatocytes, isoproterenol consistently produced greater maximal increases than adrenaline in hypothyroid hepatocytes (section 3.6). A similar difference in response to adrenaline and isoproterenol could not be observed in the much smaller cAMP response of the euthyroid rat. The alpha receptor antagonist, phenoxybenzamine, enhanced the adrenaline-induced cAMP response in the hypothyroid, but not in the normal rat. These observations may suggest that not only the beta adrenergic stimulatory influence on adenylate cyclase is greater in hypothyroidism, but an alpha receptor-mediated inhibitory effect may be enhanced as well.

'Remarkably similar results were obtained in hepatocytes from adrenalectomized rats (Chan et al. 1979a): enhanced adrenaline-stimulated cAMP accumulation was even greater after phenoxybenzamine blockade, which had no effect on this response in hepatocytes from normal control rats. Evidence for the existence of alpha receptor-mediated inhibition of adenylate cyclase has also been reported in liver preparations from carcinogen-treated rats (Boyd and Martin 1976), as well as a number of other tissues (for references see Fain and García-Sáinz 1980). Jard et al. (1981) have provided evidence that stimulation of angiotensin II or alpha, receptors inhibits normal rat liver adenylate cyclase However, the conditions required to in plasma membrane preparations. demonstrate this alpha,-mediated inhibition of adenylate cyclase were rather precise: membranes were washed in 5 mM EDTA and activity had to be assayed in the présence of GTP and monovalent cations, which raises some doubt about its role under more physiologic conditions. Exton and coworkers Assimacopoulos-Jeannet et al. 1982) characterized a similar effect in intact hepatocytes, but concluded that it was mediated predominantly by alpha<sub>1</sub> receptors. This point has been discussed in more detail in an earlier part of this thesis (section 1.3.3).

Glucagon and beta adrenergic agonists share a common mechanism of action (Exton 1982). However, in contrast to the effect of beta receptor stimulation, no difference in glucagon activation of glycogen phophorylase could be found in hypothyroid and normal rat hepatocytes (section 3.5). On the other hand, cAMP accumulation was enhanced in hypothyroid hepatocytes, but only at glucagon concentrations above that required for maximal phosphorylase activation (10 nM; section 3.5). Just as for isoproterenol, IBMX enhanced glucagon-stimulated cAMP accumulation in both normal and hypothyroid rats, but the difference at higher glucagon concentrations was maintained (section 3.5).

Both isoproterenol (section 3.4) and glucagon (section 3.5) were more potent in stimulating phosphorylase activation than cAMP accumulation. Α similar disparity in cAMP-mediated effects and cAMP accumulation in other systems has been noted. This finding in hepatocytes led Fain et al. (1977) to propose the existence of a cAMP-independent mechanism for phosphorylase activation by low glucagon concentrations. Cherrington et al. (1977) were able to identify small but reproducible increases in intracellular cAMP at low glucagon concentrations that resulted in the most sensitive changes in glycogenolysis and phosphorylase. It would appear therefore, that these apparent dissociations of cAMP response and phosphorylase activation are the result of the great amplification between the primary receptor signal (production of cAMP) and the final response, resulting as a consequence of an efficient enzyme cascade (figure 1.1). Morevover, careful measurement of intracellular cAMP levels is required to detect changes at threshold drug concentrations. In the present study, as in many others, cAMP was determined in the incubation medium plus cells homogenized together. Exton et al. (1972) have found that cAMP released into the medium from perfused liver correlated better with glucose output than did tissue levels of cAMP. Intracellular cAMP alone was not assayed in the present experiments and hence it is not known

whether the increased accumulation at higher glucagon concentrations in cells from hypothyroid rats would also be observed in the physiologically important intracellular compartment. The role of the 'extra' cAMP released from cells or produced by agonist at levels greater than those required to maximally activate phosphorylase is unknown and hence, it is difficult to attribute physiological significance to this change.

Several other groups have examined the effect of thyroid state on The results are quite variable and often glucagon actions in the liver. contradictory. In agreement with results of the present experiments, Malbon et al. (1978a) found no change in glucagon-mediated phosphorylase activation in hepatocytes from hypothyroid compared with normal rats. They also found no difference in cAMP accumulation, but<sup>v</sup> glucagon concentrations above 10 nM were not tested by Malbon et al. (1978a); the difference between hypothyroid and normal hepatocytes in the<sup>®</sup>, present study was apparent only at higher glucagon levels. In a subsequent report, plasma membranes purified from hypothyroid rat hepatocytes showed an increased specific binding of [<sup>125</sup>I]glucagon when compared with euthyroid controls (Malbon 1980a). However, only a single concentration (2.0 nM) of  $[^{125}I]$  glucagon was examined and hence these results must be interpreted with caution; single point binding determinations cannot be used to differentiate changes in affinity and binding site density, and are prone to errors introduced by cooperativity 'or the existence of multiple binding sites (Sperling et al. 1980). Glucagon-stimulated adenylate cyclase activity was examined in the same preparations (Malbon 1980a). While 1  $\mu$ M glucagon produced greater cyclase activity in liver homogenates and membranes from hypothyroid rats, this difference was not statistically significant.

These latter findings agreed with those of earlier investigations that showed no change in basal, fluoride- or glucagon-stimulated adenylate cyclase in homogenates of hypothyroid rat liver compared with cuthyroid controls (Harkcom et al. 1978). Treatment of hypothyroid rats with thyroxine resulted in higher basal and glucagon-stimulated adenylate cyclase activity than that of either hypothyroid or euthyroid rat preparations (Harkcom et al. 1978).

Sperling et al. (1980) demonstrated a substantial reduction of glucagon receptor number and adenylate cyclase response in plasma membranes from hypothyroid rats without any significant difference in these parameters between hyperthyroid and euthyroid rats. Scatchard analysis of specific  $[^{125}I]$  glucagon binding data revealed evidence for high and low affinity glucagon receptors, which may account for some of the discrepancies in the above-described findings (Sperling et al. 1980).

An additional reason for these variable findings may be the fact that some of the above preparations were whole tissue homogenates, while others were from collagenase-dispersed liver cells as in the present experiments. Hanoune and co-workers have demonstrated the presence of a proteolytic contaminant in commercial preparations of crude collagenase which produced at least a two-fold increase in adenylate cyclase activity (Hanoune et al. 1977; Lacombe et al. 1977). This factor, which was shown not to be collagenase itself, enhanced basal as well as catecholamine-, glucagon-, fluoride- or GTP-stimulated activity. Activation was time-dependent and irreversible. It may be noted that some variation in maximal cAMP reponses to isoproterenol were observed in the present experiments (sections 3.4 and The most likely explanation for these differences is variation in the 3.6). amount of contaminating proteases in collagenase batches used in preparing hepatocyte suspensions. Although care was taken to prepare all cells in a given series of experiments with a single batch of collagenase, one cannot rule out the possibility that cyclase activity in hypothyroid preparations has an increased sensitivity to activation by the protease contaminant. This possibility was not investigated in the present study.

Glucocorticoid deficiency has been shown to have similar effects on

hepatic adrenoceptors as thyroidectomy in the present study (section 4.4). In contrast to the effect of hypothyroidism, glucagon-stimulated phosphorylase activity is impaired in adrenalectomized rats, while cAMP accumulation remains normal (Chan et al. 1979b). Further investigation revealed that phosphorylase <u>b</u> kinase activation by glucagon was suppressed after adrenalectomy and this offers the best explanation for the impaired phosphorylase response (Chan et al. 1979b). Thus, while adrenalectomy and thyroidectomy have similar effects on hepatic adrenergic receptors, their effects on glucagon are quite different.

In summary, the present findings indicate that beta receptor activity is unequivocally increased in preparations from hypothyroid rats when compared with normal. This increase was apparent at the level of production of the second messenger, cAMP, and could not be attributed to differential metabolism by phosphodiesterase in the two groups of rats. A concomitant increase in putative beta adrenergic receptor binding sites has been identified by others (Malbon 1980a) and is the most likely reason for the increased activity in hypothyroidism, although additional changes in the coupling of beta receptor to adenylate cyclase cannot be ruled out. These changes were reversible by <u>in</u> <u>vivo</u> administration of  $T_3$ .

The effects of hypothyroidism on the hepatic actions of glucagon are less clear. Nonetheless, it appears that glucagon-stimulated phosphorylase activation and cAMP accumulation are not influenced in the same way as beta receptor-mediated responses. The fact that glucagon activation of glycogen phosphorylase was the same in normal and hypothyroid rat hepatocytes indicates that changes in the sensitivity of the glycogenolysis cascade to cAMP did not contribute significantly to the altered beta adrenergic response.

**4.3** The effect of hypothyroidism on hepatic alpha adrenergic receptors. Comparison with vasopressin and A23187.

The observations presented in this thesis indicate that hypothyroidism in rats results in a decrease in the alpha adrenergic stimulation of glycogen phosphorylase <u>b</u> to <u>a</u> conversion and calcium release in isolated hepatocytes. These changes were accompanied by a decrease in the density of plasma membrane binding sites for  $[^{3}H]$  prazosin, a selective alpha<sub>1</sub> receptor antagonist, which may be the basis of the altered physiological responsiveness observed in the intact cells. Furthermore, the change in both binding site density and alpha adrenergic responses could be returned towards normal levels by <u>in vivo</u> administration of T<sub>3</sub> to thyroidectomized rats.

Work from several laboratories has indicated that alpha adrenergic stimulation of rat liver results in the release of intracellular calcium from mitochondrial stores and its subsequent extrusion into the extreellular space (for references see section 1.3.3). The transient increase in cytosolic calcium levels so produced is believed to activate glycogen phosphorylase kinase, which in turn phosphorylates glycogen phosphorylase to the active a form of the enzyme. Vasopressin has been shown to activate glycogen phosphorylase by a cAMP-independent mechanism (Kirk and Hems 1974) and to cause changes in calcium distribution similar to those produced by alpha receptor agonists (Blackmore et al. 1978; Chen et al. 1978). The divalent cation ionophore, A23187, can also cause release of calcium from the same intracellular pool (Chen et al. 1978; Blackmore et al. 1978; Babcock et al. 1979) which results in the activation of glycogen phosphorylase (Blackmore et al. 1978). Accordingly, the effect of thyroidectomy on phosphorylase activation and calcium release in response to A23187 and vasopressin were examined

In contrast to the effects of phenylephrine, neither vasopressin nor A23187 activation of glycogen phosphorylase and stimulation of calcium release were

greatly influenced by thyroidectomy (sections 3.8 and 3.9), although the response to vasopressin was moderately reduced. Since the source of released intracellular calcium and the subsequent post-receptor events are believed to be the same for phenylephrine, vasopressin and A23187 (section 1.3.3), these findings suggest that the locus of the impaired alpha adrenergic response in the hypothyroid hepatocyte is at a site proximal to the release of intracellular calcium.

The fact that A23187 produced identical calcium release responses in hepatocytes from normal and hypothyroid rats (section 3.9) indicates that releasable calcium stores were not affected by thyroidectomy. Rosenavist (1978) has examined the effect of thyroidectomy on the amount and distribution of exchangeable calcium in isolated rat liver cells. He identified two distinct intracellular calcium pools: a large, slow \turnover pool and a smaller, more rapidly exchangeable pool. Thyroidectomy did not significantly alter the pool with rapid turnover, but markedly reduced the slow turnover pool. Another group of workers (Barritt et al. 1981) found a similar intracellular calcium distribution. They suggested that each kinetically defined compartment is likely to be heterogeneous with respect to anatomical composition, but that the larger compartment represents calcium sequestered in organelles and the smaller, exchangeable calcium in the cytoplasm. Stimulation with alpha agonists resulted in an increase of about 100% of exchangeable calcium in the small compartment and a decrease of 20% in the large (Barritt et al. 1981). The time-course and dose-response relationship of cytosolic free calcium changes during alpha adrenergic stimulation closely follow the increase in phosphorylase Thus it would appear that it is changes in the a (Murphy et al. 1980). smaller kinetically defined calcium pool that are important in the activation of glycogen phosphorylase. Rosenquist (1978) found no change in this pool after thyroidectomy. The 20% decrease in the larger pool during stimulation (Barritt et al. 1980) most probably represents release of calciumyfrom mitochondrial

٢,

stores (Babcock et al. 1979; Blackmore et al. 1979b,c; Murphy et al. 1980). This larger compartment was decreased after thyroidectomy (Rosenqvist 1978), but unfortunately, changes in calcium pools during hormonal stimulation have not been compared in thyroidectomized and normal rats. Whatever this effect may be, the present findings that maximal stimulation of calcium efflux and phosphorylase activation by A23187 and vasopressin were not or only slightly affected by hypothyroidism indicates that a change in the hormone sensitive calcium pool cannot fully account for the marked reduction in the effect of phenylephrine.

0

Experiments in which calcium efflux was measured with a calcium-sensitive electrode were made in nominally calcium-free medium (sections 3.7, -3.8 and 3.9). Total cellular calcium can be directly related to the concentration of calcium in the medium (Borle 1975). Hence the possibility arises that the decreased phosphorylase response to vasopressin and phenylephrine observed under these conditions (section 3.10) could result from a depletion of calcium from the hormone sensitive calcium pool, which may be greater in hypothyroid than in normal rat hepatocytes. A similar decreased responsiveness under low calcium conditions has been reported by others (Keppens et 'al. 1977; Stubbs et al. 1976; Van de Werve et al. 1977). However, the net calcium loss in response to agonist stimulation in the present experiments (approximately 0.15 to 0.30 nmol of calcium pet/mg cells) compares well with that determined by measuring total caldium in isolated hepatocytes at normal medium calcium concentrations (approximately) 0.20 to 0.30 nmol per mg cells; Blackmore et al. 1978) or the estimated total alpha adrenergic agonist-mobilized pool in isolated perfused liver (approximately 0.14 nmol per mg of liver; Reinhart et al. 1982a).

Kneer et al. (1979) have found that like phosphorylase activation, stimulation of gluconeogenesis from lactate by noradrenaline was suppressed at low extracellular calcium concentrations, whereas it was enhanced when the

substrate was fructose. These authors proposed that calcium influenced the rate of transfer of reducing equivalents between cytosol and mitochondria. Thus the increased cytosolic NADH/NAD ratio under low calcium conditions would favour gluconeogenesis from oxidized substrates (fructose) but not reduced substrates (lactate). It is possible, therefore, that suppression of phosphorylase response under low calcium may be the result of secondary alterations in the metabolic state of the hepatocyte and not an effect on calcium homeostasis per se.

In order to rule out the possibility that the observed differences in calcium release were related to the low calcium conditions under which these experiments were made,  $^{45}$ Ca efflux from prelabelled hepatocytes was examined with normal calcium levels in the medium. Results of these experiments confirm those obtained with the calcium-sensitive electrode (sections 3.7 and 3.8): phenylephrine-stimulated  $^{45}$ Ca release was suppressed after thyroidlectomy and returned towards normal by T<sub>3</sub> replacement.

While both the use of the calcium sensitive electrode and  $^{45}$ Ca efflux are felt to reflect alpha adrenoceptor activity, methodological differences preclude a direct comparison of these results. Nonetheless, one observes that by using either approach, phenylephrine-induced calcium efflux is suppressed in the thyroidectomized rat. However, the results obtained with vasopressin are somewhat less straightforward. The concentration-response curve for  $^{45}$ Ca is somewhat supressed at lower vasopressin concentrations and is significantly shifted to the right in the thyroidectomized rat (figure 3.0, table 3.4). On the other hand, in the calcium electrode studies the effect of vasopressin curve in thyroidectomized rats was somewhat suppressed over the entire concentration range, but this difference is not statistically significant (section 2.8). Hence, one cannot conclude that vasopressin-mediated calcium release is unaffected by thyroidectomy. Nevertheless, it is clear that the reduction of the alpha-mediated response is substantially greater.
Using  $[{}^{3}$ II]prazosin a decrease in putative alpha<sub>1</sub> receptor binding sites in liver plasma membranes from hypothyroid rats could be demonstrated (section 3.11). Just as for the above-described effects on alpha adrenergic stimulation of phosphorylase and calcium mobilization, the decrease in  $[{}^{3}$ H]prazosin binding was reversible by T<sub>3</sub> replacement. The observed decrease in the density of putative alpha<sub>1</sub> receptor binding sites points to a decreased alpha<sub>1</sub> receptor number as the likely basis for the altered response to alpha adrenergic stimulation in hypothyroid hepatocytes. No change in the affinity of binding sites for  $[{}^{3}$ H]prazosin or antagonist could be detected.

In the absence of significant "spare" alpha receptors in rat liver (Kunos et al. 1983), a decrease in receptor number is expected to result in a decreased maximal response to agonists as observed (section 3.7). However, one cannot rule out additional effects of thyroidectomy on the coupling of receptor activation and the release of intracellular calcium, since the nature of this process is not fully understood. Indeed, the smaller decrease in the effectiveness of vasopressin in hypothyroid rats could suggest such a possibility. Alternatively, an independent effect of hypothyroidism on vasopressin receptors could also account for this. The latter effect would be analogous to the effect of hypothyroidism on adenylate cyclase-coupled receptors in rat liver, which involve a marked increase in beta adrenergic and a much smaller, but significant increase in glucagon receptor number (section 4.2; Malbon 1980a). Direct binding of  $[{}^{3}H]$  vasopressin to isolated hepatocytes and liver membranes has been examined by Cantau et al. (1980). A comparison of  $[{}^{3}H]$  vasopressin binding in euthyroid and hypothyroid rats would clarify the effect on vasopressin-mediated responses. It is of interest to note that adenylate cyclase response to vasopressin stimulation in renal tissue particulate fractions was reversibly decreased by thyroid hormone deficiency (Harkom ct al. 1978), although this response is mediated by a different subtype of vasopressin receptor (Cantau et al. 1980).

Recent studies have implicated alpha<sub>1</sub> adrenergic effects on phosphoti@ylinositol in the activation of glycogenolysis, possibly as an intermediate in the release of intracellular calcium (section 1.3.3). Gareía-Sáinz and Fain (1980) have examined the effect of hypothyroidism on the alpha adrenergic stimulation of phosphotidylinositol labelling in rat fat cells and find no difference in this parameter when compared with euthyroid controls. These observations cannot be extended to the present results since effects of altered thyroid state are highly tissue dependent. The effect of hypothyroidism on phosphotidylinositol turnover in rat liver remains to be determined.

Changes similar to those occurring after thyroidectomy have been demonstrated in the adrenalectomized rat (see section 4.4). Chan et al. (1979a) found that adrenalectomy produced a reciprocal change in alpha and beta adrenergic activation of hepatic glycogen phosphorylase, including a decreased alpha receptor-mediated mobilization of intracellular calcium. However, binding of  $[^{3}H]$  noradrenaline to liver plasma membranes showed no significant change after adrenalectomy in the initial study (Chan et al. 1979a). This was interpreted as evidence for no change in alpha receptor binding capacity or affinity. Similar results were obtained by Guellean et al. (1978) using  $[^{3}H]$  DHE as a binding probe for alpha receptors. However, after re-examining the binding of  $^{3}H$ -catecholamines the former authors (Chan et al. 1979a) have unpublished results showing a decrease in alpha receptor binding sites after adrenalectomy (Exton et al. 1981).

As described in section 1.3.3 and 1.3.4, it has recently been shown that rat liver has at least two populations of alpha receptors (Hoffman et al. 1980a,b; Hoffman et al. 1981):  $alpha_2$  receptors that can, under certain conditions, inhibit adenylate cyclase (Jard et al. 1981) and also  $alpha_1$ receptors responsible for the activation of glycogen phosphorylase (Hoffman et al. 1980b; Aggerbeck et al. 1980a). <sup>3</sup>H-Catecholamines were shown to

label  $alpha_1$  (El-Refai 1980a) as well as  $alpha_2$  receptors in the rat liver (Hoffman et al. 1980b), and [<sup>3</sup>H] DHE also labels both alpha receptor subtypes (Hoffman et al. 1980a,b; 1981). As discussed above, an inhibitory alpha component (possibly mediated by the  $alpha_2$  receptor) may be enhanced in the hypothyroid liver and this could have influenced the above results. In the present experiments this difficulty was avoided by the use of an  $alpha_1$ , receptor selective ligand: [<sup>3</sup>H] prazosin.

A question of obvious importance is the identity of  $[^{3}H]$  prazosin binding sites with functional alpha, receptors, and the evidence for this is necessarily circumstantial. While the binding  $K_d$  values of antagonists determined by their competition for  $[^{3}H]$  prazosin were identical with their  $K_{B}$  values for antagonising adrenaline activation of phosphorylase in isolated hepatocytes<sup>1</sup>, the  $EC_{50}$  of adrenaline was 20 times and that of phenylephrine 3 times lower than their respective K<sub>d</sub> values (section 3.12). This discrepancy, which is similar to that noted with other radiolabelled antagonists (El-Refai et al. 1979), cannot be attributed to spare receptors, since experiments with the irreversible alpha antagonist phenoxybenzamine indicated a lack of significant receptor reserve in the rat liver (Kunos, et al. 1983). Hanoune and coworkers (Geynet et al. 1981) recently compared the binding characteristics of  $[^{3}H]$  prazosin,  $[^{3}H]$  DHE and  $[^{3}H]$  noradrenaline in liver plasma membranes. On the basis of their findings they proposed that  $[{}^{3}H]$  noradrenaline labelled the physiologically active form of the alpha receptor and that  $[^{3}H]$  prazosin labelled either a precursor of the active receptor or an  $alpha_1$ -like binding site not coupled to the effector system. The present findings do not rule out the possibility that only a small fraction of the ['H] prazosin binding sites with high affinity for agonists is functionally relevant, the rest being precursors for such sites or sites with undefined function (Geynet et al. 1981; El-Refai and Exton 1980b).

<sup>1</sup>Results of experiments done by W.H. Kan at low hepatocyte concentrations.

However, neither the agonist displacement curves (figure 3.24) nor the kinetics of  $[^{3}H]$  prazosin binding (figure 3.22) gave any indication of binding site heterogeneity that could suggest such a possibility. An alternative explanation of the above anomaly is offered by the hypothesis put forward by Franklin (1980). He proposed that the nonproductive interaction of antagonists with receptors is fully translated into tightness of binding, which results in K<sub>d</sub> values similar to physiological K<sub>B</sub> values. However, the productive binding energy of strong agonists is largely used to cause conformational changes necessary to initiate the biological response. This may explain the apparent paradox of low binding affinity of biologically highly potent agonists as seen in the present as well as in many previous studies (for references see section 1.3.4). Finally, the binding characteristics of [<sup>3</sup>H]prazosin are compatible with seperate interacting agonist and antagonist alphal receptor sites, the latter being labelled by [<sup>3</sup>H] prazosin. The low binding potency of agonists could reflect their low affinity for the antagonist site. The possibility of distinct agonist and antagonist forms of the alpha receptor was first proposed by U'Prichard et al. (1977), although their findings were later attributed to differential binding to alpha, and alpha, receptors (U'Prichard et al. 1978). The present findings are more consistent with the original proposal, since  $[^{3}H]$  prazosin is highly selective for the alpha, receptor.

Comparison of  $[{}^{3}H]$  prazosin binding in membranes prepared from normal, thyroidectomized and thyroidectomized, T<sub>3</sub>-treated rats showed that the decreased density of binding sites in thyroidectomized rats was unaccompanied by any change in affinity for this ligand (section 3.11). Similarly, the affinity of adrenaline for these sites was unchanged even in the presence of guanine nucleotide analogue, Gpp(NH)p (section, 3.11); agonist displacement curves were shown to be shifted to lower affinity in the presence of Gpp(NH)p when binding was to alpha<sub>2</sub> receptor sites (Hoffman et al. 1980a). A similar effect on alpha<sub>1</sub> receptor sites in liver has not been demonstrated (Hoffman et al.

1980a,b) which supports the contention that the present experiments examined only binding to alpha, receptor sites.

15

Recoveries of specific marker enzymes for plasma membranes were not examined in the present study. Sperling et al. (1980) showed that basal and fluoride-stimulated adenylate cyclase activity, were the same in euthyroid, hypothyroid and hyperthyroid rat hepatic plasma membranes isolated from whole livers by the same method used in the present experiments. Malbon (1980a) also found no difference in recovery of basal and fluoride-stimulated adenylate cyclase or in 5'-nucleotidase activity in membranes prepared from hypothyroid and normal hepatocytes. However, the latter results are not directly applicable to the present experiments since in Malbon's study, membranes were prepared from isolated hepatocytes rather than whole liver.

The present experiments clearly show a change in phenylephrine-induced activation of glycogen phosphorylase (section 3.10). However, as stated in section 4.1, others were unable to identify similar effects of thyroidectomy on alpha receptor activity (Malbon et al. 1978a, 1980). A lack, of thyroid -regulation of hepatic alpha receptors was suggested (García-Sáinz and Fain 1980) on the basis of the absence of changes in the effect of phenylephrine (Malbon et al. 1978a; 1980). While classically considered an alpha receptor agonist, phenylephrine can stimulate both beta and alpha receptors in various tissues (Chahl and O'Donnell 1969; Wagner and Schumann 1979). Its EC<sub>50</sub> for relaxation of guinea pig trachea, an effect mediated by  $beta_2$  receptors is 3.1  $\mu$ M (Chahl and O'Donnell 1969), which is close to its EC<sub>50</sub> of  $0.8 \pm 0.3 \mu M$  for alpha receptor-mediated activation of phosphorylase in normal rat liver (table 3.4). Since hepatic beta receptors are also of the beta $_2$ subtype (section 3.3), it is very likely that a significant beta receptor activation by phenylephrine contributes to its effect on phosphorylase in the hypothyroid rat liver. It may be noted that in the present study the difference in the effects of phenylephrine<sup>®</sup> in cells from euthyroid and hypothyroid rats is

greater for calcium release, a pure alpha adrenergic response, than for phosphorylase activation, a mixed alpha/beta effect. The observation of Malbon et al. (1978a) that propranolol reduced the phonylephrine-induced activation of phosphorylase in hypothyroid, but not in normal hepatocytes (Table I in Malbon et al. 1978a), also indicates potentiation of the beta receptor component in the effect of phenylephrine" in the hypothyroid liver. In this group of rats, phenylephrine was more effective in hypothyroid than euthyroid controls and propranolol abolished this difference; the similar residual activity of phenylephrine in the presence of propranolol indicated that alpha receptor activity was unchanged. However, in another group of rats used in the same study, phenylephrine tested in the absence of propranolol was not more, but even less effective in hypothyroid than in normal cells (Figure 1 in Malbon Thus in these latter animals, the increased beta receptor 1978a). et al. activity must have been offset by a similar decrease in alpha receptor activity, which is consistent with the conclusions of the present study.

Some additional difference between the present study and that of Malbon et al. (1978a; Malbon 1980a) could account for the absence of an identifiable effect of hypothyroidism on alpha activation of phosphorylase in their studies. ()) In the present experiments rats were surgically thyroidectomized and used after 10 to 14 weeks while Malbon et al. (1978a; Malbon 1980a) induced hypothyroidism by feeding an iodine-deficient diet and propylthiouracil for 18 to 25 days before the experiment. A comparison of thyroid effects produced in these two different ways on<sup>a</sup> hepatic metabolism has not been made.

(2) The present study used exclusively male rats, whereas the latter used only females. Bitensky et al. (1970) have found that adrenaline-stimulated hepatic adenylate cyclase activity was greater in preparations from female rats when compared to males. More recently, Studer and Borle (1982) demonstrated that in hepatocytes isolated from female rats beta receptor-mediated stimulation of cAMP accumulation and `its contribution' to glycogen phosphorylase activation

was enhanced when compared to males. Conversely, the alpha receptor-mediated calcium efflux was greater in males than females and accounted for virtually all the activation of phosphorylase by adrenaline in the former. Since the female rats in their euthyroid state appear to possess hepatic adrenoceptor properties similar to the hypothyroid male, it is possible that a further reduction in alpha receptors in the hypothyroid female rat may be difficult to demonstrate clearly.

(3) Animals used in the present experiments were older (250-350 g) than those used by Malbon et al. (1978a; 100-150 g). Blair et al.<sup>(1)</sup>(1979a) have examined the adrenergic control of glucose output in hepatocytes from juvenile (100-150 g) and adult (greater than 300 g) rats. They found aging to be accompanied by a loss of functional beta receptors with adrenaline acting primarily through the alpha adrenergic receptor system. Hence, it is possible that the alpha response was more pronounced in the older animals used in the present study than it may have been in younger animals.

(4) Malbon et al. (1978a) express their reults for phosphorylase activation as % change over basal. In spite of the fact that basal enzyme activities were not different when hypothyroid and euthyroid rats were compared, small changes in basal activity in a given experiment could profoundly influence the calculated stimulated enzyme activity when expressed in that way. Actual enzyme activities are reported in the present experiments.

These consideration notwithstanding, one must point out that changes in the beta adrenergic activation of phosphorylase after thyroidectomy are more profound than for the alpha receptor system. The mechanism underlying these changes and how either the emergence of beta receptors or the suppression of alpha responses in the hypothyroid rat might influence overall metabolic regulation in vivo remain matters of speculation. Each of these questions will now be briefly considered.

4.4 Similar changes in hepatic adrenoceptors produced by non-thyroidal influences.

Thyroid hormone effects on adrenergic receptors in a number of other tissues have been described (section 1.2.3). Most of these studies have involved either adipose or cardiac tissue and have shown that beta adrenergic responsiveness is decreased and alpha responsiveness increased in hypothyroidism (for a more detailed discussion and references, see section 1.2.3). Thus, the effect of hypothyroidism on hepatic adrenergic receptors is opposite to that seen in these tissues indicating that the effects of thyroid hormones on these receptors are highly tissue specific and cannot be generalized. Indeed, in some preparations no effect of altered thyroid state on adrenergic receptors could be found (for references see section 1.2.3).

As already discussed in section 4.3, similar changes in hepatic adrenergic receptors have been described in adrenalectomized rats. Wolfe et al. (1976) found an increase in isoproterenol-stimulated adenylate cyclase activity and specific binding of [<sup>125</sup>I] IHBP, a potent beta antagonist. However, this latter finding is of questionable reliability because of the unusually low affinity for this antagonist observed in that study (see section 1.3.4). These studies were extended by Chan et al. (1979a) who reported a greater beta receptor-mediated cAMP accumulation and phosphorylase b to a conversion in hepatocytes from adrenalectomized rats. Alpha adrenoceptor-mediated phosphorylase activation and calcium mobilization decreased after adrenalectomy. The fact that these changes are analogous to those reported here after thyroidectomy suggests the possibility that they may have occurred as a consequence of a thyroid hormone effect on the pituitary-adrenal axis. Indeed, it has been shown that in the rat, deficiency of thyroid hormone is associated with subnormal concentrations of corticosteroids in plasma (Bray and Jacobs 1974). Also, the effects of adrenalectomy were more rapid than thyroidectomy occuring after only 3 to 8

days. However, administration of cortisol to thyroidectomized rats using a treatment regimen that was shown to reverse adrenalectomy-induced changes in hepatic adrenoceptors (Wolfe et al. 1976) had no effect on the pattern of phosphorylase response to adrenergic agonists (section 3.3). These findings argue against thyroid effects being secondary to changes in the adrenal-pituitary axis.

But what of the converse of this hypothesis: could the effects of adrenalectomy on hepatic adrenoceptors be secondary to an effect on thyroid hormone action? Thyroid hormone receptor levels and consequently tissue sensitivity to  $T_3$  and  $T_4$  can be influenced by a variety of factors (Samuels et al. 1982). It may be then, that the liver becomes "functionally hypothyroid" in the presence of normal circulating thyroid hormone levels, and hence shows changes similar to those seen following thyroidectomy. Unfortunately, there is no simple approach to testing this hypothesis, since treatment of adrenalectomized rats with excess thyroxine may not reverse the effect in the face of reduced tissue responsiveness. It would be of interest to know how thyroid hormone receptors are influenced by adrenalectomy. To the best knowledge of the author, neither of these experiments has been done.

Hepatic adrenergic receptors can also be modulated by non-hormonal influences. Hornbrook (1978) has shown that alpha-mediated activation of glycogenolysis in hepatocytes from normal rats takes on the characteristics of a mixed alpha/beta response when examined 24 h after partial hepatectomy. In support of these findings, Brønstad and Christoffersen (1980) later demonstrated that adrenaline-stimulated cAMP accumulation and adenylate cyclase activity was enhanced after partial hepatectomy. Again, there exists a relationship with thyroid hormone effects on the liver. Partial hepatectomy in the rat reduces the number of nuclear thyroid hormone receptors (Dillman et al. 1977). Plasma glucagon levels are markedly elevated after partial hepatectomy (Morley et al. 975) and glucagon administration has been reported

to lower hepatic nuclear  $T_3$  binding capacity (Dillman et al. 1978). Thus, partial hepatectomy might also produce a situation in which the liver becomes "functionally hypothyroid" and hence, manifests changes in adrenoceptor properties analogous to those observed following thyroidectomy. Starvation also produces an elevation in circulating glucagon levels (Grey et al. 1970) as well as a decrease in  $T_3$  nuclear binding sites (Burman et al. 1977). It would, therefore, be of interest to see whether either starvation or exogenous glucagon administration would produce changes in hepatic adrenergic receptors similar to those produced by thyroidectomy. Burns et al. (1979) have observed that, after only one day of fasting, the adrenergic receptor activity of human adipocytes is altered in a way similar to that seen in hypothyroid subjects.

Treatment of adult rats with the hepatic carcinogen 2-acetylaminofluorene resulted in a marked increase in hepatic adenylate cyclase response to adrenaline (Christoffersen et al. 1972). Hornbrook (1979) later found that feeding of the same carcinogen to rats' shifted the adrenergic activation of hepatic phosphorylase in isolated hepatocytes from an alpha to a mixed Increase in beta adrenoceptor activity in isolated alpha/beta response. hepatocytes has also been reported following feeding of a second hepatic carcinogen, 3'-methyl-4-dimethylaminobenzene (Christoffersen and Berg 1975). Boyd and Martin (1976), studying the action of this second carcinogen, found an increased catecholamine- and decreased glucagon-responsive adenylate cyclase activity in preneoplastic liver. These changes were seen in non-turmor as well as tumor tissue (Boyd et al. 1978) which suggest that they may be the manifestation of some reactive change in non-tumor tissue analogous to that seen following partial hepatectomy.

Of some relevance to these studies of hepatic tumors are reports comparing adrenoceptors in normal rat liver and the Zajdela ascites hepatoma cell line (Lacombe et al. 1976). The adrenoceptor mediating adenylate cyclase activation in the normal rat liver is typically of the beta<sub>2</sub> subtype (Lacombe,

et al. 1976; Vanquelin et al. 1976). However, in the Zajdela hepatoma, adenylate cyclase stimulation occurs predominantly via activation of a beta<sub>1</sub> adrenergic receptor; moreover, the response is greatly enhanced when compared with normal rat liver (Lacombe et al. 1976; Hanoune et al. 1977). The present results indicate that enhanced beta receptor activity observed after thyroidectomy is mediated by receptors of the beta, subtype since adrenaline is more potent than noradrenaline in stimulating cAMP accumulation (section 3.4). Leray et al. (1973) reported that enhanced adrenaline-stimulated adenylate cyclase in preparations from adrenalectomized rat liver was also mediated by beta, adrenoceptors. And, data shown in the study of preneoplastic liver tissue also suggests that the receptor remains the beta, subtype; noradrenaline was less potent than adrenaline (Boyd and Martin 1976). Thus, unlike the Zajdela hepatoma, where enhanced beta receptor activity is associated with a beta<sub>1</sub> adrenoceptor type, preneoplastic liver tissue, preparations from adrenalectomized rat liver and hepatocytes from thyroidectomized rats show the characteristics of beta, adrenergic receptors as in the normal rat liver. The beta receptor subtype in liver after partial hepatectomy or in young rats (see below) has not been analyzed in detail.

Extrahepatic cholestasis produced by ligation of the common bile duct in the rat has been shown to produce a 2.6-fold increase in isoproterenol-stimulated adenylate cyclase activity in isolated liver plasma membranes after two days (Schmelck et al. 1979). This increase in activity was associated with a four-fold enhancement of beta receptor binding site density as measured by  $[^{3}H]$  DHA specific binding (Schmelck et al. 1979). Although the characteristics of alpha receptor responses have not yet been reported in cholestatic livers, Guellaen et al. (1982) have found a two- to three-fold decrease in the number of  $[^{3}H]$  DHE and  $[^{3}H]$  prazosin binding sites. Apparently only single concentrations of  $[^{3}H]$  prazosin (1 nM) were examined in the latter study and detailed reports of binding data are lacking.

Another parameter that is known to influence hepatic adrenergic receptor properties is age. Hornbrook (1978) reported that catecholamine-mediated activation of glycogen phosphorylase occurred predominantly via beta adrenoceptors in weanling rats. In a later report, Blair et al. (1979a) examined adrenergic control of glucose output and cAMP levels in juvenile rats. They found that in hepatocytes from young rats (27 to 35 days old, weighing 100 to 150 g), adrenaline stimulated glucose output from endogenous glycogen stores and cAMP accumulation via predominantly beta adrenergic receptor activation. Maturation (60 to 120 days, 300 g and greater) was accompanied by a loss of functional beta receptors with adrenaline acting primarily through Butlen et al. (1980) used [<sup>3</sup>H] DHE as an alpha receptor probe alpha receptors. to study the ontogenic development of alpha receptors in rat liver. They found that in 19-day-old fetuses the density of [<sup>3</sup>H] DHE binding sites was about 1.5 times that of the adult, but decreased to one guarter in late fetal life and the first two weeks after birth. Unfortunately, beta receptors have not as yet been studied in a similar way. On the other hand, Kalish et al. (1977) found that adveraline-sensitive adenviate cyclase activity was doubled in liver membranes prepared from 24 month old rats when compared to that found at 3 or 12 months. Those findings could imply the existence of an enhanced hepatic beta receptor activity in senescent rats, although the receptor mediating adrenaline's effect was not characterized in that study. Moreover, basal, fluoride- and glucagon-stimulated activities were also enhanced, but not to the same degree as adrenaline.

Three points about these observations that relate to the present study are noteworthy.

(1) Properties of hepatic adrenoceptors are similar in normal young rats and adult rats following thyroidectomy.

(2) This suggests that thyroid hormones may exert a primary effect on hepatic adrenoceptors by influencing the developmental state of the liver (for further

discussion of this point refer to section 4.5)

(3) Normal euthyroid controls used throughout the present experiments were weight-matched, and hence hypothyroid rats were on average 10 to 14 weeks older than controls. However, the fact that in adult rats (300 g and greater) the predominant hepatic adrenoceptor becomes and remains an alpha receptor (Blair et al. 1979a) argues against the possibility that the observed difference between these groups was due to age. Kalish et al. (1977) found no difference in adrenaline-stimulated adenylate cyclase in liver plasma membranes from three month and 12 month old male rats.

A final factor that is known to influence the balance of hepatic adrenergic receptors is sex and has already been briefly discussed in section 4.3. Recall that Studer and Borle (1982) have shown that hepatic adrenoceptors in female rats resemble those described for hypothyroid males in the present study; activation of glycogen phosphorylase is by catecholamine is shared by alpha and beta receptors, alpha receptor-mediated calcium mobilization is suppressed and beta receptor-mediated cAMP accumulation is enhanced when compared to males. How these effects might be related to thyroid influences is not clear, although, sex differences in the metabolism and actions of thyroid hormones have been described (Harris et al. 1979), and one might postulate that female rats behave such that their livers are "functionally hypothyroid" as compared to males. This interesting sex-related difference and what it might mean physiologically invites further study.

Changes in the properties of hepatic adrenergic receptors have also been described in the <u>ex vivo</u> state. Okajima and Ui (1982) have examined the adrenergic regulation of glycogen phosphorylase and synthase in rat hepatocytes in primary culture. They found that beta receptor contributions become predominant over alpha after only 8 h in culture. This shift in adrenoceptor properties could be prevented by the addition of cycloheximide to the culture medium suggesting a role for protein synthesis in the observed change. It

141

Ĵ.

would be of interest to examine the effect of addition of thyroid hormone or glucocorticoid in a similar experiment.

The findings presented here as well as those of Malbon (1980a) suggest that the alterations in hepatic adrenoceptor properties in hypothyroidism are the result of changes in the density of adrenergic receptors in the plasma membrane. A similar conclusion can be drawn about changes occurring after adrenalectomy (Guellaen et al. 1978, Exton et al. 1981) or hepatic cholestasis (Schmelck et al. 1979; Guellaen et al. 1982). Unfortunately the effects of the other factors described above on alpha and beta receptor densities has not been examined directly. Thus one cannot conclude that these changes are completely analogous to the hypothyroid state. Nonetheless, it appears that hepatic adrenoceptors are influenced by a number of physiological and pathological changes of which hypothyroidism is a single example. Moreover, several of these can be hypothetically related to altered thyroid hormone function and hence, may share a common mechanism. Although evidence supporting this notion is at present only circumstantial, some aspects are worthy of further experimental consideration.

## 4.5 Physiological implications of thyroid-induced changes in hepatic adrenoceptor properties.

The physiological role of the sympathetic nervous system in the control of hepatic carbohydrate metabolism remains enigmatic. Therefore in the context of this broader question, any discussion of how changes in hepatic adrenoceptor properties of the kind described here influence in <u>vivo</u> metabolic regulation would be pure conjecture. Nonetheless, some interesting aspects of this problem are worth considering in a speculative way.

Why two different pathways whereby catecholamines stimulate hepatic glucose output should coexist is problematic. Exton and Harper (1975) have

speculated that the duality of this system may have survival value, pointing out that the liver plays a vital role in the "fight or flight" situation. However, the developmental, sex and species dependent variation in the balance of alpha and beta receptors remains unexplained; duality of the system if it has survival value should be of benefit under any circumstances. The shift in this balance observed in various pathological states such as carcinogenesis, hepatectomy, adrenal insufficiency, cholestasis and hypothyroidism (section 4.4) can be viewed in two ways. Either it is an undesirable change and contributes to the pathologic state or it represents an adaptive response that allows the 'animal to optimize its metabolic machinery in this pathological state. If alpha and beta receptors represent mere duplications that contribute survival value, one would expect these changes in the balance of receptors to gain little advantage in acute pathological states. One must, therefore, presume that alpha and beta receptor activation must have different effects on the overall metabolism of the cell.

Hormones affecting hepatic metabolism can be considered in two broad groups (see section 1.3): those which, like beta adrenoceptor agonists, produce effects that are mediated via the adenylate cyclase/cAMP-dependent protein <sup>4</sup> kinase system and others, such as alpha receptor agonists, whose effects are While most studies have stressed the similarities of the calcium-mediated. effects on cellular metabolism of these two pathways some differences have The identification of metabolic effects that are not shared by been noted. these two groups suggests possible points at which alpha and beta receptor stimulation would have different metabolic consequences. Unfortunately, there are few studies in which the metabolic effects of hepatic beta and alpha receptor activation have been compared within the same experimental However, cAMP-dependent hormonal effects (usually glucagon preparation. stimulation) and calcium-dependent hormonal effects (alpha adrenergic, vasophessin and angiotensin II) in hepatocyte preparations have been compared.

By analogy, these observations provide support for the hypothesis that alpha and beta receptor stimulation can alter cellular metabolism in very different ways, and hence probably provide the animal with a metabolic "flexibility" that under some circumstances may be essential for survival rather than a mere duplication of receptor systems.

Garrison et al. (1979) have compared the pattern of phosphorylation of rat liver proteins after stimulation with glucagon and noradrenaline. These authors were able to resolve 11 to 12 cytoplasmic proteins that were phosphorylated in response to hormonal stimulation. Protein peaks whose phosphorylation was increased by noradrenaline in the presence of 20 AM propranolol were the same as those increased by glucagon both in position and magnitude. Three of these peaks were identified as phosphorylase a, glycogen synthase and pyruvate kinase. Vasopressin, angiotensin II and A23187 gave similar results. However, one-dimensional SDS gels used in that study lack sufficient resolution to determine if the proteins phosphorylated by calcium-linked hormones were distinct from those affected by glucagon stmulation. Thus, when two-dimensional gels were analyzed in a subsequent series of experiments (Garrison and Wagner 1982), it was observed that calcium-dependent hormones and glucagon were able to affect the phosphorylation of distinct substrates. These authors found six substrates uniquely phosphorylated by glucagon, one by the calcium-linked hormones and seven by both.

Garrison and Wagner (1982) draw particular attention to some of the enzymes of the gluconeogenic pathway. In contrast to glucagon, the calcium-dependent hormones did not stimulate the phosphorylation of phosphofructokinase, fructose-6-phosphate 2 kinase or fructose-1,6bisphosphatase and had only a slight effect on pyruvate kinase. Others (Chan and Exton 1978; Garrison and Borland 1979; Hems et al. 1978b) have observed that while gluconeogenesis can be stimulated by both cAMP-dependent and

calcium-dependent pathways, hormones functioning via the latter stimulate gluconeogenesis to a lesser extent at a glycogenolytic equivilant dose.

In this context the results of one study where indirectly the effects of alpha and beta stimulation were compared is of interest. Blair et al. (1979b) have shown that adrenaline markedly inhibits pyruvate kinase and lactate production in 'juvenile rats, but is only slightly effective in adults and yet, adrenaline-stimulation of glucose output in these two groups is comparable (Blair et al. 1979a). These authors suggest that this difference is due to to the fact that catecholamine effects in younger animals are mediated predominantly by beta adrenoceptors, whereas alpha receptors are relatively more important in older rats (Blair et al. 1979b).

Others have shown that alpha-mediated inhibition of pyruvate kinase is less effective than that produced by glucagon (Chan and Exton 1978; Garrison and Borland 1979). It has been suggested that this may be the reason for the generally observed less effective stimulation of gluconeogenesis by catecholamines compared with glucagon (for references see Garrison and Borland 1979). However, Rognstad and Katz (1977) have reported that in cells from fasted rats, maximal concentrations of adrenaline stimulated glucneogenesis as effectively as glucagon, but had no effect on flux through pyruvate kinase, while glucagon inhibited flux through this step by approximately 60%. This observation led Chan and Exton (1978) to suggest that alpha adrenergic stimulation of gluconcogenesis could occur at another control point in the pathway, possibly fructose-1,6-bisphosphatase. It is interesting to note that recently Van Schaftingen et al. (1980) have shown that fructose-2,6-bisphosphate is an important regulator at this point in the--gluconeogenic pathway. This metabolite inhibits fructose-1,6-bisphosphatase and activates phosphofructokinase (Herst and Van Schaftingen 1982). Glucagon stimulation of isolated hepatocytes causes a marked decreases in levels of this metabolite and is thought to be in part responsible for the enhanced

. 145

gluconeogenesis under these conditions (Van Schaftingen and Hers 1982). However, A23187, phenylephrine and vasopressin cause an increase in fructose-2,6-bisphosphate content of the hepatocyte and yet the latter two agents also stimulate gluconeogenesis (Whitton et al. 1978). The interesting possibility that this regulator exerts effects on other metabolic pathways and control points requires further investigation.

Still other mechanisms not involving enzyme phosphorylation directly may be at work in the cell. For example, Merryfield et al. (1982) have shown that phosphénolpyruvate carboxykinase, another key enzyme in the gluconeogenic pathway, is activated by the release of  $Fe^{++}$  from rat liver mitochondria mediated by free calcium. These authors speculate that alpha receptor agonists as well as other calcium-dependent hormones may exert a regulatory influence on this enzyme via this mechanism. Whether cAMP-dependent hormones/may have a similar effect is unknown.

Another indication of profound differences in the effect of these two groups of hormones on hepatocyte metabolism comes from the study of Reinhart et al. (1982b). These authors demonstrated that phenylephrine, vasopressin and angiotensin II administered to perfused rat liver produced a rapid parallel increase in glucose output and oxygen uptake. On the other hand, glucagon's effect on glucose output was slower and occurred in the absence of any large change in respiration. These authors concluded that mitochondria play a vital role in mediating the glycogenolytic effect of the calcium-dependent hormones, but not glucagon.

Although these two groups of hormones appear to have different effects on cellular metabolism, evidence exists to suggest that these, pathways do not function independently: hormones of one group may modify or override the effects of the second. Exton and coworkers (Assimacopoulos-Jeannet et al. 1982) demonstrated that phenylephrine and adrenaline could suppress glucagon-mediated elevation of cAMP and gluconeogenesis from lactate in

isolated hepatocytes. In contrast to the studies of Jard et al. (1981) who demonstrated that adenylate cyclase inhibition in isolated plasma membranes occurred via  $alpha_2$  receptors, Exton's group (Assimacopoulos-deannet.et al. 1982) showed that the inhibitory effect on cAMP accumulation in hepatocytes suggested an  $alpha_1$  receptor mechanism was more important. The contradictory nature of these findings may be a consequence of the different preparations used. Whether this inhibitory effect is mediated by  $alpha_2$  or  $alpha_1$  receptors, the ability to also inhibit stimulation of gluconeogenesis by exogenous cAMP suggests that other loci in addition to adenylate cyclase may be involved (Assimacopoulos-Jeannet et al. 1982). The elucidation of the true nature of this interaction and its possible role <u>in vivo</u> require further study.

Assuming that alpha and beta receptor stimulation do affect the cell's metabolic machinery in different ways, how and under what circumstances does differential stimulation of these receptors occur? As can be seen from table 3.1, the sensitivity to glycogen phosphorylase stimulation by adrenaline is not markedly changed in the hypothyroid rat; half-maximal phosphorylase activation  $\delta ccurs$  at 7.73 (pD<sub>2</sub>) and 7.96 (pD<sub>2</sub>) for normal and hypothyroid rats respectively. Since at least some residual alpha receptor activity remains, one would not expect the threshold for phosphorylase activation to be greatly altered in the hypothyroid rat liver. However, hypothyroid rats possess adrenoceptors of the beta, subtype, (section 4.4) which are very much less sensitive to noradrenaline released from hepatic sympathetic nerves (section 1.2.3) than are alpha receptors. One may then speculate that in the hypothyroid rat the major pathway for adrenergic activation of liver phosphorylase is through stimulation of beta, receptors by adrenaline released from the adrenal medulla, whereas in normal rats noradrenaline released from sympathetic nerve, endings and acting on alpha, receptors would be more important. As described in section 1.4 there is some evidence that such an arrangement might exist in the rabbit and the guinea pig which normally

possesses both alpha and beta hepatic adrenoceptors.

In conclusion, one must concede that in the face of only sparse information about the role of sympathetic nervous system in control of hepatic metabolism there is little that can be said about the physiological role of thyroid-induced changes in adrenoceptor properties. However, as described above, logical reasons do exist for expecting that such changes may have important physiological consequences.

## 4.6 Possible mechanisms of thyroid induced alterations in hepatic adrenoceptor properties.

The present study documents a clear effect of thyroid state on hepatic adrenoceptor properties, but unfortunately provides little insight into the possible mechanism of such changes. However, isolated hepatocytes provide an ideal model system which might be used to reproduce and study these effects under controlled in <u>vitro</u> conditions. Thus far there is some evidence that thyroid horniones can produce changes in adrenergic receptor properties in isolated cardiac tissue (see section 1.2.3), but attempts to affect similar changes in isolated adipocytes has met with little success (see discussion in Fain 1981).

One must consider that thyroid hormones could influence adrenergic receptors other than by a direct effect on tissue for which the change is identified. For example, thyroid-induced alterations in the metabolic state that result in changes in adrenergic receptors could represent part of an overall adaptive response of the animal to its new physiologic state. There is at present little that can be said to refute or support this possibility other than to reiterate the facts that thyroid hormones have profound effects on almost every metabolic system studied and that adrenergic receptors exhibit significant "plasticity" in their ability to change in order to-accommodate new stimuli

(Hoffman and Lefkowitz 1980). Furthermore, the fact that similar changes are observed in a number of situations not directly related to thyroid hormone action argues for a generalized nature of the phenomenon (section 4.4).

Thyroid hormones can modulate the actions of a number of other hormones and may influence adrenergic receptors secondarily via one of these. This possibility has been discussed with respect to corticosteroids (section 4.4). Results presented here suggest that thyroid hormone effects on hepatic adrenoceptors do not occur because of corticosteroid deficiency in hypothyroidism. However, the possibility that part of the effect of adrenalectomy on hepatic adrenergic receptors may be due to an effect on thyroid hormone actions cannot be ruled out (see section 4.4).

One possible clue to the mechanism relates to the apparent effect of thyroid hormones on the developmental state of the tissue. Adequate thyroid function is necessary for normal growth and differentiation of many tissues (Greenberg et al. 1974) as are glucocorticoids. Hepatic beta receptors are relatively more important in young rats and there is an apparent shift in the predominant hepatic adrenoceptor from alpha to beta after partial hepatectomy, exposure to hepatocarcinogens or in hepatocytes maintained in primary culture (see section 4.4). The similarity between these situations and the hypothyroid rat model suggests that the presence of thyroid hormone may be required to maintain the state of differentiation of the tissue. When thyroid hormone is removed the hepatocyte may revert to a less well differentiated form in which the beta receptor contribution is more significant than in the normal adult animal.

An effect of thyroid hormone directly on the tissue could influence adrenergic reponsiveness by an action on the receptor site or some post-reeptor event. The loss of beta receptor activity in fat cells from hypothyroid rats provides an example of the latter (see section 1.2.3); the refractoriness of adipose tissue to adrenergic stimuli was due in part to an increase in cAMP

149

Ć

phosphodiesterase (Armstrong et al. 1974) and possibly also a loss of coupling between the beta receptor and adenylate cyclase (Fain 1981). In the present experiments the increased beta and decreased alpha receptor responsiveness are accompanied by an increase in beta and decrease in alpha receptor binding site densities. Although this is the most likely explanation for the altered tissue responsiveness the possible existence of additional thyroid effects on receptor coupling or other post-receptor events cannot be ruled out.

How then might thyroid state modulate adrenoceptor numbers in the liver? Thyroid hormone receptors have been described in a number of sub-cellular structures including the plasma membrane and the nucleus (Sterling 1979; Bernal and DeGroot 1980). There is evidence that thyroid hormone stimulates protein synthesis at both the level of translation and transcription (Bernal and DeGroot 1980). Hence, it is possible that thyroid hormones may, directly influence the production of receptor protein or some other gene product that has a regulatory effect on the adrenergic receptor. If these effects could be reproduced <u>in vitro</u> one could test the role of protein synthesis directly by the use of specific inhibitors. Recall that the emergence of beta receptor activity in hepatocytes maintained in primary culture could be prevented by the addition of cycloheximide, a protein synthesis inhibitor (Okajima and Ui 1982).

Kempson et al. (1978) reported changes in beta receptor binding in cardiac tissue incubated in vitro with thyroid hormone that appeared not to be dependent on protein synthesis (section 1.2.4). Thus one could speculate that a specific effect of thyroid hormone at the level of the cell membrane may be important. The incorporation or activation of cryptic receptors from an inactive membrane or cytoplasmic pool may be regulated by thyroid hormone through either a direct effect on such receptors or as a consequence of a more generalized change such as alteration of membrane fluidity (Mendoza et al. 1977).

Of particular interest with respect to possible direct effects of thyroid

hormones on adrenergic receptors is the proposal that alpha and beta receptors represent two forms of a single protein which can interconvert in response to certain stimuli such as temperature or thyroid state (Kunos 1977, 1978). However, in light of recent demonstrations of the existence of at least four types of adrenergic receptors, some with very different mechanism for response activation (section 1.2.3) this interconversion hypothesis has become somewhat strained. Wood et al. (1979) have solubilized from hepatic plasma membranes seperate macromolecules that have the property of binding [<sup>3</sup>H] WB 4101 and [<sup>3</sup>H] DHA respectively, and concluded that these receptors, therefore, do not reside on the same macromolecule. However, recovery of alpha receptor-like binding sites was low (1-5%) and may not be truly representative of the hepatic receptor. Furthermore, as discussed in section 1.3.4, the most appropriate ligands for identifying alpha and beta receptors in liver are unsettled issues. Kunos (1980, 1981) has broadened the definition of "interconversion" to imply a close functional coupling between receptors. Hence, a decrease in one receptor type would result in a concomitant increase in the second type to which it is coupled. Such coupled regulation may be mediated indirectly by a thyroid hormone-dependent cellular signal other than the adrenergic receptor itself. Indeed, observations in many experimental systems have shown that in many cases when either alpha or beta receptor activities change they do so in a 'reciprocal manner. The explanation for this is not known.

In conclusion, a variety of possible mechanism whereby thyroid hormones could modulate adrenergic receptor activity exist. Some of these have been described above. The elucidation of the actual mechanism(s) involved in this regulation will require a greater understanding of both thyroid hormone action and adrenergic receptor properties. Such studies would be facilitated by a reliable <u>in vitro</u> model system in which thyroid regulation of adrenergic receptor properties can be demonstrated. The isolated rat hepatocyte might provide such a model.

## SECTION FIVE

## SUMMARY AND CONCLUSIONS

• (Contributions of Original Knowledge)

\$

° Ф

\*X i

1. The effect of thyroid hormones on hepatic adrenergic receptors was examined in hepatocyte suspensions prepared from euthyroid, hyperthyroid, hypothyroid and hypothyroid, triiodothyronine-treated rats.

2. Experiments with adrenergic agonists and selective antagonists showed that 'activation of liver glycogen phosphorylase in cells prepared from euthyroid and hyperthyroid rats was mediated by alpha adrenergic receptors.

3. The induction of hypothyroidism revealed the emergence of a significant beta receptor contribution to this response, as well as an enhanced accumulation of cAMP in response to beta receptor stimulation. The presence of a phosphodiesterase inhibitor did not eliminate this difference.

4. The response to glucagon, which also activates glycogen phosphorylase via a cAMP-dependent mechanism, was not similarly affected; phosphorylase activation by glucagon was the same in hepatocytes from hypothyroid and euthyroid rats. Cyclic AMP accumulation was enhanced in hypothyroid hepatocytes at high glucagon concentrations and this difference was maintained in the presence of a phosphodiesterase inhibitor.

5. This emergence of beta receptor activity in hypothyroidism was accompanied by a decreased alpha receptor activity, as demonstrated by a suppression of phenylephrine-stimulated phosphorylase activation and calcium efflux from isolated hepatocytes.

6. Responses to vasopressin and the divalent cation ionophore, A23187, which like alpha agonists produce activation of liver glycogen phosphorylase by a release of intracellular calcium, were not similarly affected; the effectiveness of vasopressin in stimulating calcium efflux was slightly reduced and that of A23187 unaffected in hypothyroid hepatocytes.

7. Analysis of specific binding of  $[{}^{3}H]$  prazosin, a selective alpha<sub>1</sub> antagonist, to liver plasma membranes revealed a decrease in the density of putative alpha receptor sites in preparations from hypothyroid rats. Hypothyroidism did not influence the binding affinity of adrenergic agonists or  $[{}^{3}H]$  prazosin to alpha<sub>1</sub>

153- 📜

receptor sites. This change was opposite to the increase in density of beta receptors in hypothyroid rat liver reported by others.

8. All of these changes: the increased beta receptor activity, the decreased alpha receptor activity and the decreased density of alpha<sub>1</sub> receptor binding sites could be returned towards normal by prior administration of triiodothyronine to hypothyroid rats.

9. It is concluded that hypothyroidism produces a selective and reversible enhancement of beta receptor activity and supression of alpha receptor activity in rat hepatocytes. These changes are most likely the result of alterations in the density of receptor sites in the plasma membrane.

154

· .

<1.0

5

(;

(

REFERENCES

Aggerbeck M, Guellaen G and Hanoune J (1980a) Adrenergic receptor of the alpha\_subtype mediates the activation of glycogen phosphorylase in normal rat liver. Biochem Pharmacol 29:643-645.

- Aggerbeck M, Guellaen G and Hanoune J (1980b) The α-adrenergic mediated effect in rat liver. Correlation between [<sup>3</sup>H] dihydrocrgocryptine binding to plasma membranes and glycogen phosphorylase activation in isolated hepatocytes. Biochem Pharmacol 29:1653-1662.
- Ahlquist RP (1948) Study of adrenotropic receptors. Am J Physiol 153:586-600. Althaus-Saltzman M, Carafoli E and Jakob A (1980)  $Ca^{2+}$ , K<sup>+</sup> redistribution and  $\alpha$ -adrenergic activation of glycognolysis in perfused rat liver. Eur J Biochem 106:241-248.
- Armstrong KJ, Stouffer JE, Van Inwegen RG, Thompson WJ and Robison GA (1974) Effects of thyroid hormone deficiency on cyclic ad enosine 3',5'-monophosphate and control of lipolysis in fat cells. J Biol Chem 249:4226-4231.
- Assimacopoulos-Jeannet FD, Blackmore PF and Exton JH (1977) Studies on  $\alpha$ -adrenergic activation of hepatic glucose output. Studies on the role of calcium in  $\alpha$ -adrenergic activation of phosphorylase. J Biol Chem 252:2662-2669.
- Assimacopoulos-Jeannet FD, Blackmore PF and Exton JH (1982) Studies of the interaction between glucagon and  $\alpha$ -adrenergic agonists in the control of hepatic glucose output. J Biol Chem 257:3759-3765.
- Atkinson DE (1968) Citrate and the citrate cycle in the regulation of energy metabolism. In: The Metabolic Roles of Citrate. Ed: TW Goodwin Published by: Academic Press, London. pp 23-40
- Babcock DF, Chen J-LJ, Yip BP and Lardy HA (1979) Evidence for mitochondrial localization of the hormone-responsive pool of Ca<sup>2+</sup> in isolated hepatocytes. J Biol Chem 254:8117-8120.

Baquer NZ, Cacales M, McLean P and Greenbaum AL (1976) Effects of thyroid

hormone deficiency on the distribution of hepatic metabolites and control of pathways of carbohydrate metabolism in liver and adipose tissue of the rat. Eur J Biochem 68:403-413.

- Barritt AJ, Parker JC and Wadsworth JC (1981) A kinetic analysis of the effects of adrenaline on calcium distribution in isolated rat liver parenchymal cells. J Physiol 312:29-55.
- Becker J and Jakob A (1982) α-Adrenergic stimulation of glycolysis and Na<sup>+</sup>, K<sup>+</sup>-transport in perfused rat liver. Eur J Biochem 128:293-296.
- Bernal J and DeGroot LJ (1980) Mode of action of thyroid hormones. In: The Thyroid Gland pp 123-143, chapter 6 Editor: M De Visscher Published by: Raven Press, New York
- Bernard C (1849) Influence du systeme nerveux sur la production du sucre dans l'economie animale. Soc philam 49
- Berry MN and Friend DS (1962) High-yield peparation of isolated rat liver parenchymal cells. J Cell Biol 43:506-520.
- Berthelson S and Pettinger WA (1977) A functional basis for classification of alpha-adrenergic receptors. Life Sci 21:595-606.
- Berthon B, Poggioli J, Capiod T and Claret M (1981) Effect of the  $\alpha$ -agonist noradrenaline on total and  ${}^{45}Ca^{2+}$  movements in mitochondria of rat liver cells. Biochem J 200:177-180.

Billah MM and Michell RH (1978) Stimulation of the breakdown and resynthesis

- of phosphatidylinositol in rat hepatocytes by angiotensin, vasopressin and adrenaline. Biochem Soc Trans 6:1033-1035.
- Birnbaum MJ and Fain JN (1977) Activation of protein kinase and glycogen phosphorylase in isolated rat liver cells by glucagon and catecholamines. J Biol Chem 252:528-535.
- Bitensky MW, Russell V and Blanco M (1970) Independent variation of glucagon and epinephrine responsive components of hepatic adenyl cyclase as a function of age, sex and steroid hormones. Endocrinology 86:154-159.

- Blackmore PF, Assimacopoulos-Jeannet FD, Chan TM and Exton JH (1979a) Studies on α-adrenergic activation of hepatic glucose output. Insulin inhibition of α-adrenergic and glucagon actions in normal and calcium-depleted hepatocytes. J Biol Chem 254:2828-2834.
- Blackmore PF, Brumley FT, Marks JL and Exton JH (1978) Studies on  $\alpha$ -adrenergic activation of hepatic glucose output. Relationship between  $\alpha$ -adrenergic stimulation of calcium efflux and activation of phosphorylase in isolated rat liver parenchymal cells. J Biol Chem 253:4851-4858.
- Blackmore PF, Dehaye JP, Strickland WG and Exton JH (1979b) α-Adrenergic mobilization of hepatic mitochondrial calcium. FEBS Lett 100:117-120.
- Blackmore PF, Dehaye JP and Exton JH (1979c) Studies on the  $\alpha$ -adrenergic activation of hepatic glucose output. The role of mitochondrial calcium release in  $\alpha$ -adrenergic activation of phosphorylase in perfused rat liver. J Biol Chem 254:6946-6950.
- Blackmore PF, El-Refai MF and Exton JH (1979d)  $\alpha$ -Adrenergic blockade and inhibition of A23187 mediated Ca<sup>2+</sup> uptake by the calcium antagonist verapamil in rat liver cells. Mol Pharmacol 15:598-606.
- Blackmore PF, Hughes BP, Shuman EA and Exton JH (1982) α-Adrenergic activation of phosphorylase in liver cells involves mobilization of intracellular calcium without influx of extracellular calcium. J Biol Chem 257:190-197.
- Blackmore PF and Exton JH (1981)  $\alpha_1$ -Adrenergic stimulation of Ca<sup>2+</sup> mobilization without phosphorylase activation in hepatocytes from phosphorylase <u>b</u> kinase-deficient <u>gsd/gsd</u> rats. Biochem J 198:379-383.
- Blair JB, James ME and Foster JL (1979a) Adrenergic control of glucose output and adenosine 3',5'-monophosphate levels in hepatocytes from juvenile and adult rats. J Biol Chem 254:7579-7584.
- Blair JB, James ME and Foster JL (1979b) Adrenergic control of glycolysis and pyru vate kinase activity in hepatocytes from young-and old rats. J Biol Chem 254:7585-7590.

- Borle AB (1975) Methods for assessing hormone effects on calcium fluxes in vitro. In: Methods in Enzymology Vol XXXIX. Ed: JG Hardman and BW O'Malley. Academic Press. pp 513-573.
- Bottger I, Kriegel H and Weiland O (1970) Fluctuation of hepatic enzymes important in glucose metabolism in relation to thyroid hormone function. Eur J Biochem 13:253-257.
- Boyd H, McAffee DA and Rubin JJ (1978) A study of the cyclic nucleotide metabolism and the histology of rat liver during 3'-methyl-4- dimethylaminoazobenzene carcinogenesis. II. Cyclic AMP metabolism. Tissue and Cell 10:477-494.
  - Boyd H and Martin TJ (1976) Changes in catecholamine- and glucagon-responsive adenylate cyclase activity in preneoplastic rat liver. Mol Pharmacol 12:195-202.
  - Bréant B, Keppens S and DeWulf H (1981) Heterologous desensitization of the cyclic AMP-independent glycogenolytic response in rat liver cells. Biochem
    J 200:509-514.
- Bray GA and Jacobs HS (1974) Thyroid activity and other endocrine glands. In: Handbook of Physiology Ed: RO Greep and EB Astwood. American Physiological Society, Washington, D.C. pp 413-433.
- Brodde O-E, SchumannH-J and Wagner J (1980) Decreased responsiveness of the adenylate cyclase system on left atria from hypothyroid rats. Mol Pharmacol 17:180-186.
- Brønstad G and Christoffersen T (1980) Increased effect of adrenaline on cyclic AMP formation and positive β-adrenergic modulation of DNA-synthesis in regenerating hepatocytes. FEBS Lett 120:89-93.

Brunton LL, Hayes JS and Mayer SE (1981) Functional compartmentalization of cyclic AMP and protein kinase in heart. Adv Cyc Nuc Res 14:391-397.
Burgess GM, Claret M and Jenkinson DH (1979) Effects of catecholamines, ATP and ionophore A23187 on potassium and calcium movements in isolated

hepatocytes. Nature 282:415-417.

- Burman KD, Lukes Y, Wright FD and Wartofsky (1977) Reduction in hepatic triiodothyronine binding capacity induced by fasting. Endocrinology \*101:1331-1337.
- Burns TW, Boyer PA, Terry BE, Langley PE and Robison GA (1979) The effect of fasting on the adrenergic receptor activity of human adipocytes. J Lab Clin Med 94:387-394.
- Butlen D, Guillon G, Cantau B and Jard S (1980) Comparison of the developmental patterns of vasopressin, glucagon and A-adrenergic receptors from rat liver membranes. Mol Cell Endocrinol 19:275-289.
- Cardinás-Tánus RJ, Huerta-Bahena J and García-Sáinz JA (1982) Angiotensin II inhibits the accumulation of cyclic AMP produced by glucagon but not its metabolic effects. FEBS Lett 143:1-4.
- Caldwell A and Fain JN (1971) Triidothyronine stimulation of cyclic adenosine 3',5'-monophosphate accumulation in fat cells. Endocrinology 89:1195-1211.
- Campanile CP, Grane JK, Peach MJ and Garrison JC (1982) The hepatic angiotensin II receptor. II.Effect of guanine nucleotides and interaction with cyclic AMP production. J Biol Chem 257:4959-4965.
- Cantau B, Keppens S, DeWulf H and Jard S (1980) [<sup>3</sup>H] Vasopressin binding to isolated hepatocytes and liver membranes: regulation by GTP and relation to glycogen phophorylase activation. J Rec Res 1:137-168.
- Chahl IA and O'Donnell SR (1969) Actions of phenylephrine on β-adrenoceptors in guinea pig trachea. Brit J Pharmacol 37:41-51.
- Chan TM, Blackmore PF, Steiner KE and Exton JH (1979a) Effects of adrenalectomy on hormone action on hepatic glucose metabolism. Reciprocal change in  $\alpha$ - and  $\beta$ -adrenergic activation of hepatic glycogen phosphorylase and calcium mobilization in adrenalectomized rats. J Biol Chem 254:2428-2433.

Chan TM, Steiner KE, and Exton JH (1979b)-Effects of-adrenalectomy\_on

hormone action on hepatic glucose metabolism. Impaired glucagon activation of glycogen phosphorylase in hepatocytes from adrenalectomized rats. J Biol Chem 254:11374-11378.

- Chan TM and Exton JH (1977) α-Adrenergic-mediated accumulation of adenosine 3';5'-monophosphate in calcium-depleted hepatocytes. J Biol Chem 252:8645-8651.
- Chan TM and Exton JH (1978) Studies on α-adrenergic activation of hepatic glucose output. Studies on α-adrenergic inhibition of hepatic pyruvate kinase and activation of gluconeogenesis. J Biol Chem 253:6393-6400.

log s .

Chang HY and Kunos G (1981) Short term effects of triiodothyronine on rat heart adrenoceptors. Biochem Biophys Res Commun 100:313-320.

Chen J-LJ, Babcock DF and Lardy HA (1978) Norepinephrine, vasopressin, glucagon and A23187 induce effux of calcium from an exchangeable pool in isolated rat hepatocytes. Proc Natl Acad Sci USA 75:2234-2238.

- Cheng Y and Prusoff WH (1973) Relationship between the inhibition constant  $(K_{I})$  and the concentration of inhibitor which causes 50% inhibition  $(I_{50})$  of and enzymatic reaction. Biochem Pharmacol 22:3099-3108.
- Cherrington AD, Hundley RF, Dolgin S and Exton JH (1977) Studies of the role of beta-adrenergic receptors in the activation of phophorylase in rat hepatocytes by catecholamines. J Cyc Nuc Res 3:263-273.
- Chock PB, Rhee SG and Stadtman ER (1980) Interconvertible enzyme cascades in cellular regulation. Ann Rev Biochem 49:813-843.
- Chrisman TD, Jordan JE and Exton JH (1982) Purification of rat liver phosphorylase kinase. J Biol Chem 257:10798-10804.
- Christoffsen T and Berg T (1975) Altered hormonal control of cyclic AMP formation in isolated parenchymal liver cells from rats treated with 2-acetylaminoflourene. Biochem Biophys Acta 381:72-77.

Christofferson T, Morland J, Osnes J-B and Elgjo K (1972) Hepatic adenylate

carcinogen. Biochim Biophys Acta 279:363-366.

Ciaraldi T and Marinetti GV (1977) Thyroxine and propylthiouracil effects in vivo on alpha and beta adrenergic receptors in rat heart. Biochem Biophys\_ Res Commun 74:984-991.

Ciarialdi T and Marinetti GV (1978) Hormone action at the membrane level. VIII. Adrenergic receptors in rat heart and adipocytes and their modulation by thyroxine. Biochem Biophys Acta 541:334-346.

Clark MG and Jarret IG (1978) Responsiveness to glucagon by isolated rat hepatocytes controlled by the redox state of the cytosolic nicotinamide-adenine dinucleotide couple acting on adenosine 3':5'-monophosphate phosphodiesterase. Biochem J 176:805-816.

Clarke WR, Jones LR and Lefkowitz RJ (1978) Hepatic a-adrenergic receptors. Identification and subcellular localization using [<sup>3</sup>H] dihydroergocryptine. J Biol Chem 253:5975-5979.

Claus TH and Pilkis SJ (1977) Effect of dichofacetate and glucagon on the incorporation of labeled substrates into glucose and on pyruvate dehydrogenase in hepatocytes from fed and starved rats. Arch Biochem Biophys 192:52-63.

Cohen P (1978) The role of cyclic AMP-dependent protein kinase in the regulation of glycogen metabolism in mammalian skeletal muscle. Curr Top Cellu Regul 14:117-196.

Cohen, P, Burchell A, Foulkes JG, Cohen PTW, Vanaman TC and Nairn AC (1978) Identification of the Ca<sup>2+</sup>-dependent modulator protein as the fourth subunit of rabbit skeletal muscle phosphorylase kinase. FEBS Lett 92:287-293.

Colard O and Breton M (1981) Rateliver plasma membrane phospholipids ' methylation: its absence of direct relationship to adenylate cyclase activities. ' Biochem Biophys Res Commun 101:727-733.

Correze C, Auclair R and Nunez J (1976) Cyclic nucleotide phosphhodiesterases,

insulin and thyroid hormones. Mol Cell Endocrinol 5:67-69.

Correze C, Laudat MH and Nunez J (1974) Hormone-dependent lipolysis in fat cells from thyroidectomized rats. Mol Cell Endocrinol 1:309-327.

Corvera S and García-Sáinz JA (1981)  $\alpha_1$ -Adrenoceptor activation stimulates ureogenesis in rat hepatocytes. Eur J Pharmacol 72:387-390.

Côté TE and Epand RM (1979) № -Trinitrophenol glucagon. An inhibitor of glucagon-stimulated cyclic AMP production and its effects on glycogenolysis. Biochem Biophys Acta 582:295-306.

Cuatrecasas P, Tell GPE, Siea V, Parikh I and Chang K-J (1974) Noradrenaline binding and the search for catecholamine receptors. Nature 247:92-97.

Dale HH (1906) Some physiologic actions of ergot. J Physiol London 34:163-206.

Dax E-M, Parafilla JS and Gregerman RI (1981) Quantitation of beta adrenergic receptors in rat liver: confounding effect of displaceable, but nonstereoselective antagonist binding. J Rec Res 2:267-284.

Debons AF and Schwartz IL (1961) Dependence of lipolytic action of epinephrine in vitro upon thyroid hormone. J Lipid Res 2:86-89.

Dehaye J-P, Blackmore PF, Venter JC and Exton JH (1980) Studies on the  $\alpha$ -adrenergic activation of hepatic glucose output.  $\alpha$ -Adrenergic activation of phosphorylase by immobil ized epinephrine. J Biol Chem 253:3905-3910. Dehaye JP, Blackmore PF and Exton JH (1981) Studies on  $\alpha$ -adrenergic activation of hepatic glucose output. Insulin inhibition of  $\alpha$ -adrenergic mobilization of mitochondrial calcium in the perfused liver. Biochem J 194:949-956.

Dillman WH, Bonner RA and Oppenheimer JH (1978) Glucagon administration decreases hepatic nuclear triidothyronine binding capacity. Endocrinology 102:1633-1636.

Dillman WH, Schwartz HL and Oppenheimer JH (1977) Specific nuclear  $T_3$ binding sites decrease in the regenerating liver. Clin Res 25:283a.

El-Refai MF, Blackmore PF and Exton JH (1979) Evidence for two alpha-adrenergic binding sites in liver plasma membranes. Studies with [<sup>3</sup>H] epinephrine and [<sup>3</sup>H] DHE J Biol Chem 254:4375-4386.

El-Refai MF and Exton JH (1980a) Subclassification of two types of α-adrenergic binding sites in rat liver. Eur J Pharmacol 62:201-204.

El-Refai MF and Exton JH (1980b) Effects of trypsin on binding of  $[{}^{3}$ H] epinephrine and  $[{}^{3}$ H] dihydroergocryptine to rat liver plasma membranes. Evidence for interconversion of binding sites. J Biol Chem 255:5853-5858.

- Ellis S (1967) The effects of sympathomimetic amines and adrenergic blocking agents on metabolism. In: Physiological Pharmacology, vol IV. Ed: WS Root and FG Hofman. Academic Press, New York.
- Exton JH (1979) Mechanisms involved in effects of catecholamines on liver carbohydrate metabolism. Biochem Pharmacol 28:2237-2240.
- Exton JH (1980) Mechanisms involved in, α-adrenergic phenomena: role of calcium ions in actions of catecholamines in liver and other tissues. Am J Physiol 238:E3-E12.
- Exton JH (1981) Molecular, mechanisms involved in alpha-adrenergic responses. Mol Cell Endocrinol 23:233-264.
- Exton JH (1982) Regulation of carbohydrate metabolism by cyclic nucleotides. In: Handbook of Experimental Pharmacology Vol 58/II, pp 3-87.
- Exton JH, Assimacopoulos-Jeannet FD, Blackmore PF, Cherrington AD and Chan
  TM (1978) Mechanism of catecholamine actions on liver carbohydrate metabolism. Adv Cyc Nuc Res 9:441-452.
- Exton JH, Blackmore PF, El-Refai MF, Dehaye J-P, Strickland WG, Cherrington AD, Chan TM, Assimacopoulos-Jeannet FD and Chrisman TD (1981) Mechanisms of hormonal regulation of liver metabolism. Adv Cyc Nuc Res
14:491-505.

- Exton JH, Lewis BB, Ho RJ and Park CR (1972) The role of cyclic AMP in the control of hepatic glucose production by glucagon and insulin. Adv Cyc Nuc Res 1:91-101.
- Exton JH, Robison GA, Sutherland EW and Park CR (1971) Studies on the role of adenosine 3',5'-monophosphate in hepatic actions of glucagon and catecholamines. J Biol Chem 246:6166-6177.
- Exton JH and Harper SC (1975) Role of cyclic AMP in the actions of catecholamines on hepatic carbohydrate metabolism. Adv Cyc Nuc Res 5:519-532.
- Fain JN (1981) Catecholamine-thyroid interactions in liver and adipose tissue. Life Sci 28:1745-1754.
- Fain JN, Birnbaum MJ, Pointer RH and Butcher FR (1977) The role of cyclic
  adenosine and guanosine 3',5'-monophosphate in the regulation of glycogenolysis in rat liver cells. In: Proceedings of the First International
  Symposium on Hormonal Receptors in Digestive Tract Physiology, INSERM
  Symposium No 3 (S Bonfils, P Fromageot and C Rosselin Eds) pp 155-164, Elsevier, Amsterdam.
- Fain JN and García-Sáinz JA (1980) Role of phosphatidylinositol turnover in alpha-l and of adenylate cyclase inhibition in alpha-2 effects of catecholamines. Life Sci 26:1183-1194.
- Fain JN and Malbon CC (1979) Regulation of adenylate cyclase by adenosine. Mol Cellu Biochem 25:143-149.
- Fain JW (1981) Catecholamine-thyroid interactions in liver and adipose tissue. Life Sci 28:1745-1754.
- Fletterick RJ and Madsen NB (1980) The structures and related functions of phosphorylase a. Ann Rev Biochem 49:31-61.
- Foden S and Randle PJ (1978) Calcium metabolism in rat heoatocytes. Biochem J 170:615-625.

Foulkes JG and Cohen P (1979) Hormonal control of glycogen metabolism. Phosphorylation of protein phosphatase inhibitor-1 in vivo in response to adrenaline. Eur J Biochem 97:251-256.

- Francis SH and Kono T (1982) Hormone-sensitive cAMP phosphodiesterase in liver and fat cells. Mol Cell Biochem 42:109-116.
- Franklin TJ (1980) Binding energy and the activation of hormone receptors. Biochem Pharmacol 29:853-856.
- Frazer A Hess ME and Shawfeld J (1969) The effects of thyroxine on rat heart adenosine 3',5'-monophosphate, phosphorylase <u>b</u> kinase and phosphorylase <u>a</u> activity. J Pharmacol Exp Ther 170:10-16.
- Friedman N, Divakaran P, Kirkland J, Kimura S and Wood J (1979) Effects of the calcium ionophore A23187 on liver metabolism. J Pharm Exp Ther 211:127-132.
- Furchgott RF (1978) Pharmacological characterization of receptors: its relation to radioligand-binding studies. Federation Proc 37:115-120.
- García-Sánz JA, Litosch I, Hoffman BB, Lefkowitz RJ and Fain JN (1981) Effect of thyroid status on  $\alpha$ - and  $\beta$ -catecholamine responsiveness of hamster adipocytes. Biochim Biophys Acta 678:334-341.
- García-Sáinz JA and Fain JN (1980) Effect of adrenergic amines on phosphatidylinositol labeling and glycogen synthase activity in fat cells from euthyroid and hypothyroid rats. Mol Pharmacol 18:72-77.
- Garrison JC, Borland MK, Florio VA and Twible DA (1979) The role of calcium ion as a mediator of the effects of angiotensin II, catecholamines and vasopressin on the phosphorylation and activity of enzymes in isolated hepatocytes. J Biol Chem 254:7146-7156.
- Garrison JC and Borland MK (1979) Regulation of mitochondrial pyruvate
  carboxylation and gluconeogenesis in rat hepatocytes via an α-adrenergic,
  adenosine 3':5'-monophosphate-independent mechanism. J Biol Chem
  254:1129-1133.

Garrison JC and Wagner JD (1982) Glucagon and the  $Ca^{2+}$ -linked hormones andgiotensisn II, norepinephrine, and vasopressin stimulate the phosphorylation of distinct substrates in intact hepatocytes. J Biol Chem 257:13135-13143.

- Gerich JE, Charles MA and Grodsky GM (1976) Regulation of pancreatic insulin and glucagon secretion. Ann Rev Physiol 38:353-388.
- Geynet P, Borsodi A, Ferry N and Hanoune J (1980) Proteolysis of rat liver membranes cancels the guanine nucleofilde sensitivity of agonist binding to the alpha-adrenoceceptor. Biochem Biophys Res Commun 97:947-954.
- Geynet P, Ferry N, Borsodi A and Hanoune J (1981) Two distinct  $\alpha_1$ -adrenergic receptor sites in rat liver: differential binding of [<sup>3</sup>H] norepinephrine, [<sup>3</sup>H] prazosin and [<sup>3</sup>H] dihydroergocryptine. Biochem Pharmacol 30:1665-1675. Giudicelli Y (1987) Thyroid-hormone modulation of the number of  $\alpha$ -adrenergic receptors in rat fat-cell membranes. Biochem J 176:1007-1010,
- Giudicelli Y, Lacasa D and Agli B (1980) White fat cell  $\alpha$ -adrenergic receptors and responsiveness in altered thyroid status. Biochem Biophys Res Commun 94:1113-1122.
- Goldberg ND, O'Dea RF and Haddox MK (1973) Cyclic GMP. Adv Cyc Nuc Res 3:155-223.
- Goris J, Detreyn G, Vandenheede JR and Merlevede W (1978) Protein inhibitors of dog liver phosphorylase phosphatase dependent on and independent of protein kinase. Eur. J Biochem 91:457-464.
- Goswami A and Rosenberg IN (1978) Thyroid hormone modulation of epinephrine-induced lipolysis in rat adipocytes: a possible role of calcium. Endocrinology 103:2223-2233.
- Graham RM, Hess HJ and Homey CJ (1982) Biophysical characterization of the purified a -adrenergic receptor and identification of hormone binding subunit. J Biol Chem 257:15174-15181.

Greenberg AH, Najjar S and Blizzard RM (1974) Effects of thyroid hormone on growth, differentiation, and development.In: Handbook of Physiology, Section

7, Vol III, pp377-390. ed: MA Greer and DS Solomon American Physiological Society, Washington, D.C..

Grey N, McGuigan JE and Kipnis DM (1970) Neutralization of endogenous glucagon by high titer glucagon antiserum. Endocrinology 86:1383-1389.

Guellaen G, Aggerbeck M, Schmelck P-H, Barouki R and Hanoune J (1982) Physiological and physiopathological modulations of the balance between  $\alpha$ and  $\beta$ -adrenoceptors. J Cardiovasc Pharmacol 4:S46-S50.

Guellaen G, Aggerbeck M and Hanoune J (1979) Characterization and solubilization of the  $\alpha$ -adrenoccptor of rat liver plasma membranes labelled with [<sup>3</sup>H]POB J Biol Chem 254:10761-10768.

Guellaen G, Yates-Aggerbeck M, Vauguelin G, Strosberg D and Hanoune J (1978) Characterization with  $[{}^{3}H]$  DHE of the  $\alpha$ -adrenergic receptor of the hepatic plasma membrane. Comparison with the  $\alpha$ -adrenergic receptor in normal and adrenalectomized rats. J Biol Chem 253:1114-1120.

Guellean G, Goodhardt M, Barouki R and Hanoune J (1982) Subunit structure of rat liver  $\alpha_1$ -adrenergic receptors. Biochem Pharmacol 31:2817-2820.

Gumma KA, Hothersall JS, Greenbaum AL and McLean P (1977) Thyroid hormone control of cyclic nucleotide phosphodiesterases and the regulation of the sensitivity of the liver to hormones. FEBS Lett 80:45-48.

Hagino Y and Nakashima M (1973) Adrenergic receptors in rat liver. I. Effects of epinephrine and adrenergic blocking agents in perfused liver. Japan J Pharmacol 26:535-542.

Hagino Y and Nakashima M (1974) Adrenergic receptors in rat liver. II. Effects of phenylephrine, isoproterenol and adrenergic blocking agents on gluconeogenesis in perfused liver. Japan J Pharmacol 24:373-381.

Hagino Y and Shigei T (1976) Effect of adrenergic agonists on the cyclic AMP level in the liver and heart from normal and hypothyroid rats. Japan J Pharmacol 26:335-342.

Hanoune J, Pecker F, Lacombe ML and Rene E (1977) The liver cyclic AMP

system and its sensitivity to catecholamines. In: First International Symposium on Hormonal Receptors In Digestive Tract Physiology, INSERM Symposium No 3. Editors: Bonfils et al. Elsevier/North Holland Biomedical Press.

Hanoune J, Stengel D, Lacombe ML, Feldman G and Coudier E (1977)
Proteolytic activation of rat liver adenylate cyclase by a contaminant of crude collagenase from clostridium histolyticum. J Biol Chem 252:2039-2045.
Harkcom TM, Kim JK, Palumbo PJ, Hui YS and Dousa TP (1978) Modulatory
effect of thyroid function on enzymes of the vasopressin-sensitive adenosine

3',5'-monophosphate system in renal medulla. Endocrinology 102:1475-1484.

Harris ARC, Vegenakis AG and Braverman LE (1979) Sex-related differences in outer ring monodeiodination of thyroxine and 3,3',5'-triiodothyronine by rat liver homogenates. Endocrinology 104:645-652.

Harrisson TS (1964) Adrenal medullary and thyroid relationships. Physiol Rev 44:161-185.

Hashimoto H and Nakashima M (1978) Influence of thyroid hormone on the positive inotropic effects mediated by α- and β-adrenoceptors on isolated guinea pig atria and rat pappilary muscles. Eur J Pharmacol 50:337-347.
Haylett DG (1976) Effect of sympathomimetic amines on <sup>45</sup>Ca efflux from liver slices. Br J Pharmacol 57:158-160.

- Haylett DG (1979) The identification of metabolic receptors for catecholamines.
  In: Trends in Autonomic Pharmacology. Ed: S Kalsner. Urban and Schwartzenberg, Baltimore.
- Haylett DG and Jenkinson DH (1972a) Effects of noradrenaline on potassium efflux, membrane potential and electrolyte levels in tissue slices prepared from guinea-pig liver. J Physiol 225:721-750.
- Haylett DG and Jenkinson DH (1972b) The receptors concerned in the actions of catecholamines on glucose release, membrane potential and ion movements in guinea-pig liver. J Physiol 225:751-772.

Hems DA (1977) Short-term hormonal control of hepatic carbohydrate and lipid catabolism. FEBS Lett 80:237-245.

- Hems DA, Davies CJ and Siddle K (1978a) Effects of hormones on the content of purine nucleoside cyclic monophosphates in perfused rat liver. FEBS Lett 87:196-198.
- Hems' DA, McCormack JG and Denton RM (1978b) Activation of pyruvate dehydrogenase in the perfused rat liver by vasopressin. Biochem J 176:627-629.
- Hems DA, Rodrigues LM and Whitton PD (1978c) Rapid stimulation by vasopressin, oxytocin and angiotensin II of glycogen degradation in hepatocyte suspensions. Biochem J 172:311-317.
- Hems DA and Whitton PD (1980) Control of hepatic glycogenolysis. Physiol Rev 60:1-50.
- Hers H-G and Van Schaftingen E (1982) Fructose-2,6-bisphosphate 2 years after its discovery. Biochem J 206:1-12.
- Hers HG (1976) The control of glycogen metabolism in the liver. Ann Rev Biochem 45:167-189.
- Himms-Hagen J (1972) Effects of catecholamines on metabolism. In: Handbook of Experimental Pharmacology, vol XXXIII, chapt 11, pp.363-462. (H Blaschko and E Muscholl, eds). Springer-Verlag, N.Y., N.Y..
- Hirata F and Axelrod J (1980) Phospholipid methylation and biological signal transmission. Science 209:1082-1090.
- Hoch FL (1974) Metabolic effects of thyroid hormones. In: Handbook of Physiology, Section 7, Vol. III, pp 391-412 Editors: AM Greer and DH Solomon American Physiological Society, Washington, D.C.
- Hoffman BB, DeLean A, Wood CL, Schocken DD and Leftkowitz RJ (1979) Alpha-adrenergic receptor subtypes: quantitative assessement by ligand binding. Life Sci 24:1739-1746.

Hoffman BB, Dukes DF and Lefkowitz RJ (1981) Alpha-adrenergic receptors in

liver membranes: delineation with subtype selective radioligands. Life Sci 28:265-272.

- Hoffman BB, Michel T, Kilpatrick DM, Leftkowitz RJ, Tolbert MEN, Gilman H and Fain JN (1980b) Agonist versus antagonist binding to  $\alpha$ -adrenergic receptors. Proc Natl Acad Sci USA 77:4569-4573.
- Hoffman BB, Mullikin-Kilpatrick D and Leftkowitz RJ (1980a) Heterogeneity of radioligand binding to  $\alpha$ -adrenergic receptors. Analysis of guanine nucleotide regulation of agonist binding in relation to receptor subtypes. J Biol Chem 255:4645-4652.
- Hoffman BB and Lefkowitz RJ (1980) Radioligand binding studies of adrenergic receptors: new insights into molecular and physiological regulation. Ann Rev Pharmacol Toxicol 20:581-608.
- Hornbrook KR (1970) Adrenergic receptors for metabolic responses in the liver. Fed Proc 29:1381-1385.
- Hornbrook KR (1978) Alterations in the responsiveness of the adrenergic receptor in parenchynnal cells isolated from livers of weanling rats and of adult rats after partial hepatectomy. The Pharmacologist 20:166a.
- Hornbrook KR (1979) Responsiveness of the adrenergic receptor in parenchymal cells isolated from livers of rats fed 2-acetylaminofluorene (AAF). Fed Proc 38:540a
- Hornbrook KR, Quinn PV, Siegel JH and Brody TM (1965) Thyroid hormone regulation of cardiac glycogen metabolism. Biochem Pharmacol 14:925-938.
  Hornbrook KR and Cabral A (1972) Enhancement by thyroid hormone treatment of norepinephrine-induced phosphorylase activation in the rat heart. Biochem Pharmacol 21:897-907.
- Hutson NJ, Brumley FT, Assimacopoulos FD, Harper SC and Exton JH (1976) Studies on the  $\alpha$ -adrenergic activation of hepatic glucose output. I.Studies on the  $\alpha$ -adrenergic activation of phosphorylase and gluconeogenesis and inactivation of glycogen synthase in isolated rat liver parenchymal cells. J

Biol Chem 251:5200-5208.

Huttner WB, DeGennero LJ and Greengard P (1981) Differential phosphorylation of multiple sites in purified protein I by cyclic AMP-dependent and calcium-dependent protein kinases. J Biol Chem 256:1482-1488.

- Ingebritsen TS, Geelen MJH, Parker RA, Evenson KJ and Gibson DM (1979) Modulations of hydroxymethylglutaryl-CoA reductase activity, reductase kinase activity and cholesterol synthesis in rat hepatocytes in response to insulin and glucagon. J Biol Chem 254:9986-9989.
- Jackowski MM, Johnson RA and Exton JH (1982) Regulation of adrenergic stimulation of hepatic adenylate cyclase by divalent cations. Biochim Biophys Acta 714:74-83.
- Jakob A and Diem S (1975) Metabolic responses of perfused rat livers to alphaand beta-adrenergic agonists, glucagon and cyclic AMP. Biochim Biophys Acta 404:57-66.
- Jard S, Cantau B and Jakobs KH (1981) Angiotensin II and q-adrenergic agonists inhibit rat liver adenylate cyclase. J Biol Chem 256:2603-2606.
- Jenkinson DH, Haylett DG, Koller K and Burgess GM (1978) Classification and actions of liver cell adrenoceptors. In: Recent Advances in the Pharmacology of Adrenoceptors. Ed: E Szabadi, CM Bradshaw and P Beaven Elsevier/North Holland Biomedical Press.
- Jones JK, Ismail-Beigi F and Edelman IS (1972) Rat liver adenylate cyclase activity in various thyroid states. J Clin Invest 51:2498-2501.
- Kalish MI, Katz MS, Pineyro MA and Gregerman RI (1977) Epinephrine and glucagon-sensitive adenylate cyclases of rat liver during aging. Evidence for membrane instability associated increased enzymatic activity. Biochim Biophys Acta 483:452-466.
- Kan WH, Farsang C, Preiksaitis HG and Kunos G (1979) Affinity labelling of alpha-adrenpceptors in intact liver cells by [<sup>3</sup>H]POB Biochem Biophys Res Commun 91:303-311

13

Katz S, Hamilton D, Tenner T and McNeill JH (1977) Cyclic AMP-dependent
protein kinase activation in hearts from euthyroid and hyperthyroid rats. Res
Com Chem Path Pharmacol 18:777-780.

- Kawai Y and Arinze IJ (1981) Activation of glycogenolysis in neonatal liver. J Biol Chem 256:853-858.
- Kempson S, Marinetti GV and Shaw A (1978) Hormone action at the membrane level. VII Stimulation of dihydroalprenolol binding to beta-adrenergic receptors in isolated rat heart ventricle slices by triidothyronine and thyroxine. Biochem Biophys Acta 540:320-329.
- Keppens S, Vandenheede JR and DeWulf H (1977) On the role of calcium as a second messenger in liver for the hormonally induced activation of glycogen phosphorylase. Biochem Biophys Acta 496:448-457.
- Keppens S and DeWulf H (1982) Characteristics of the desensitization and resensitization of the cyclic AMP-independent glycogenolytic response in rat liver cells. Biochem J 208:317-322.

Khoo JC and Steinberg D (1975) Stimulation of rat liver phosphorylase kinase by micromolar concentrations of  $Ca^{2+}$  FEBS Lett 57:68-72.

Kirk CJ, Verrinder TR and Hems DA (1977) Rapid stimulation, by vasopressin and adrenaline, of inorganic phosphate incorporation into phosphatidylinositol in isolated hepatocytes. FEBS Lett 83:267-271.

Kirk CJ and Hems DA (1974) Hepatic action of vasopressin: lack of a role for adenosine-3',5'-cyclic monophosphate. FEBS Lett 47:128-130.

- Kneer NM, Bosch AL, Clark MG and Lardy HA (1974) Glucose inhibition of epinephrine stimulation of hepatic gluconeogenesis by blockade of the α-receptor function. Proc Nat Acad Sci USA 71:4523-4527.
- Kneer NM, Wagner MJ and Lardy HA (1979) Regulation by calcium of hormonal effects on gluconeogenesis. J Biol Chem 254:12160-12168.
- Krebs EG and Beavo JA (1979) Phoshorylation-dephosphorylation of enzymes. Ann Rev Biochem 48:923-959.

Krebs HA and Pfenseleit K (1932) Unter-suchungen uber die Harnstoffbildung im Tierkorper. Z Physiol Chem 201:33-66.

Kunos G (1977) Thyroid hormone-dependent interconversion of myocardial alphaand beta-adrenoceptors in the rat. Br J Pharmacol 59:177-189.

Kunos G (1978) Adrenoceptors. Ann Rev Pharmacol Toxicol 18:291-301.

Kunos G (1981) Modulation of adrenergic reactivity and adrenoceptors by thyroid hormones. In: Adrenoceptors and Catecholamine Action, Part A. Ed: G Kunos, John Wiley and Sons, Toronto. chpt 10, pp 297-334.

Kunos G, Kan WH, Greguski R and Venter JC (1983) Selective affinity labeling and molecular characterization of of hepatic  $\alpha_1$ -adrenergic receptors with [<sup>3</sup>H]phenoxybenzamine. J Biol Chem 258:326-332.

Kunos G, Mucci L and Jeager V (1976) Interconversion of myocardial adrenoceptors: its relation to adenylate cyclase activation. Life Sci 19:1597-1602.

Kunos G, Mucci L and O'Regan S (1980) The influence of hormonal and neuronal factors on rat heart adrenoceptors. Br J Pharmacol 71:371-386.

Kuo S-H, Kamoba JK and Lum BKB (1977) Adrenergic receptor mechanisms involved in the hyperglycemia and hyperlacticacidemia produced by sympathetic amines in the cat. J. Pharmacol Exp Ther 202:301-309.

Lacombe ML, Rene E, Guellaen G and Hanoune J (1976) Transformation of the  $\beta_2$ -adrenoceptor in normal rat liver into a  $\beta_1$ -type in Zajdela hepatoma. Nature 262:70-72.

Lacombe ML, Stengel D and Hanoune J (1977) Proteolytic activation of adenylate cyclase from rat-liver plasma membranes. FEBS Lett 77:159-163.
Laker ME and Mayes PA (1981) Effect of hyperthyroidism and hypothyroidism on lipid and carbohydrate metabolism of the perfused rat liver. Biochem J 196:247-255.

Lands AM, Arnold A, McAuliff JP, Luduena FP and Brown TC (1967) Differentiation of receptor systems activated by sympathomimetic amines.

Nature 214:597-598.

(Q)

Landsberg L (1977) Catecholamines and hyperthyroidism. Clin Endocrinol Metab 6:697-702.

Landsberg L and Axelrod J (1968) Influence of pituitary, thyroid and adrenal hormones on norepinephrine turnover and metabolism in the rat heart. Circ Res 22:559-571.

Langer SZ (1974) Presynaptic regulation of catecholamine release. Biochem. Pharmacol 23:1793-1800.

Lautt W and Cote MG (1977) The effect of 6-hydroxydopamine-induced hepatic sympathectomy on the early hyperglycemic response to surgical trauma under anesthesia. J Trauma 17:270-274.

Lautt WW (1979) Neural activation of  $\alpha$ -adrenoccptors in glucose mobilization from liver. Can J Physiol Pharmacol 57:1037-1039

Lautt WW (1980) Hepatic nerves: a review of their functions and effects. Can J Physicl Pharmacol 58:105-123.

Lawrence JC and Larner L (1977) Evidence for alpha-adrenergic activation of phoaphorylase and inactivation of glycogen synthase in rat adipocytes. Mol Pharmacol 13:1060-1075.

Leak D (1970) The Thyroid and the Autonomic Nervous System. William Heinemann, Medical Books, Ltd. London.

LeCam A and Freychet P (1978) Effect of catecholamines on amino acid transport in isolated rat hepatocytes. Endocrinology 102:379-385.

Lee EYC, Silberman SR, Ganapathi MK, Petrovic S and Paris H (1980) The phosphoprotein phosphatases: properties of the enzymes involved in the regulation of glycogen metabolism. Adv Cyc Nuc Res 13:95-131.

Leray F, Chambaut AM, Perrenoud ML and Hanoune J (1973) Adenylate cyclase activity of rat liver plasma membranes. Hormonal stimulation and effect of adrenalectomy. Eur J Biochem 39:185-192.

Levey GS, Skelton CL and Eipstein SE (1969) Decreased myocardial adenylate

cyclase activity in hypothyroidism. J Clin Invest 48:2244-2250.

Li H-C (1982) Phosphoprotein phosphatases. Curr Top Cellu Regul 21:129-174. Limbird LE (1981) Activation and attenuation of adenylate cyclase. Biochem J 195:1-13.

Llobera M, Seibel J and Herrera E (1978) Metabolic response to short periods of starvation in hypo- and hyperthyroid rats. Horm Metab Res 10:319-324.

- Llobera M and Herrara E (1980) Effects of starvation on in vivo gluconeogenesis in hypo- and hyperthyroid rats. Endocrinology 106:1628-1633.
  Loeb JN (1978a) Metabolic Changes, Chapter 38. In: The Thyroid, Section C: Hyperthyroidism, pp 707-708, Fourth Edition. Editors: SC Werner and SH Ingbar Published by: Harper and Row Publishers, Hagerstown, Maryland, USA.
  Loeb JN (1978b) Metabolic Changes, Chapter 46, Review of Body Systems, I In: The Thyroid, Section D: Hypothyroidism, pp 873-874. Fourth Edition. Editors: SC Werner and SH Ingbar Published by: Harper and Row Publishers, Hagerstown, Maryland, USA.
- Lotersztajn S, Hanoune J and Pecker F (1981) A high affinity calcium-stimulated, magnesium-dependent ATPase in rat liver plasma membranes. J Biol Chem 256:11209-11215.
- Lowry OH, Rosenbrough NJ, Farr AL and Randall RJ (1951) Protein measurement with the folin phenol reagent. J Biol Chem 193:256-275.
- Malbon CC (1980a) Liver cell adenylate cyclase and beta-adrenergic receptors. Increased beta-adrenergic receptor number and responsiveness in the hypothyroid rat. J Biol Chem 255:8692-8699.
- Malbon CC (1980b) The effects of thyroid status on the modulation of fat cell  $\beta$ -adrenergic receptor affinity by guanine nucleotides. Mol Pharmacol 18:193-198.
- Malbon CC, Gilman HR and Fain JN (1980) Hormonal stimulation of cyclic AMP accumulation and glycogen phosphorylase activity in calcium-depleted hepatocytes from euthyroid and hypothyroid rats. Biochem J 188:593-599.

Malbon CC, Li S and Fain JN (1978a) Hormonal activation of glycogen phosphorylase in hepatocytes from hypothyroid rats. J Biol Chem 253:8820-8825.

- Malbon CC, Moreno FJ, Cabelli RJ and Fain JN (1978b) Fat cell adenylate cyclase and beta-adrenergic receptors in altered thyroid states. J Biol Chem 253:671-687.
- Malbon CC and Cabelli RJ (1978) Evaluation of the negative cooperativity model for fat cell beta-adrenergic receptors. Biochem Biophys Acta .544:93-101.
- Malbon CC and Fain JN (1978) Glycogen phosphorylase activity in hepatocytes from hypothyroid rats: reduced alpha- and enhaced beta-adrenergic activation. Fed Proc 37:1825a.
- Malbon CC and Greeberg ML (1982) 3,3',5-Triiodothyronine administration in vivo modulates the hormone sensitive adenylate cyclase system of rat hepatocytes. J Clin Invest 69:414-426.
- Malbon CC and LoPresti JJ (1981) Hyperthyroidism impairs the activation of glycogen phosphorylase by epinephrine in rat hepatocytes. J Biol Chem 256:12199-12204.
- Martinez JM, Llobera M, Cornella M and Herrera E (1977) Effects of hypo and hyperthyroidism on the response to glucose loading of blood glucose, ketones and insulin and liver glycogen as studied in the fasted rat. Revista Espanola de Fisiologia 33:323-330.
- Massague J and Guinovart JJ (1978) Insulin counteraction of α-adrenergic effects on liver glycogen metabolism. Biochim Biophys Acta 543:269-272.

Mayer SE (1980) Neurohumoral transmission and the autonomic nervous system.

 In: Goodman and Gilman's The Pharmacolgical Basis of Therapeutics, Sixth Edition. Ed: AG Gilman, LS Goodman and A Gilman. McMillan Publishing Co., Inc. New York. chpt 4, pp 56-90.

McConnaughey MM, Jones LR, Watanabe AM, Besch HR Jr, Williams LT and

Lefkowitz RJ (1979) Thyroxine and propylthiouracil effects on alpha- and beta-adrenergic receptor number, ATP-ase activitys and sialic, acid content of rat cardiac membrane vesicles. J Cardiovasc Pharmacol 1:609-616.

McGarry JD and Foster DN (1980) Regulation of hepatic fatty acid oxidation and ketone body production. Am Rev Biochem 49:395-420.

McNeill JH (1977) Enhancement of the cardiac phosphorylase activating effect of dibutyryl cyclic AMP by thyroid hormone. Res Com Chem Path Pharmacol 16:735-743.

- McNeill JH, Muschek LD and Brody TM (1969) The effect of triiodothyronine on cyclic AMP, phosphorylase and adenyl cyclase in rat heart. Can J Physiol Pharmacol 47:913-916.
- McNeill JH and Brody TM (1968) The effect of triiodothyronine pretreatment on amine-induced rat cardiac phosphorylase activation. J Pharmacol Exp Ther 161: 40-46.

Menahan LA and Weiland O (1969) The role of thyroid function in the metabolism of perfused rat liver with particular reference to gluconeogenesis. Eur J Biochem 10:188-194.

Mendoza D, Moreno H, Massa EM, Morero RD and Farias RN (1977) Thyroid hormone actions and membrane fluidity. FEBS Lett 84:199-203.

Merrifield ML, Kramp DC and Lardy HA (1982) Purification and characterization of a rat liver ferroactivator with catalase activity. J Biol Chem 257:4646-4654.

Michell RH (1975) Inositol phospholipids and cell surface receptor function, Biochim Biophys Acta 415:81-147.

Michell RH and Kirk CJ (1981) Why is phosphatidylinositol degraded in response to stimulation of certain receptors? TIPS 2:86-89.

Mikhail Y and Salek AL (1961) Intrinsic nerve fibers in the liver parenchyma. Anat Rec 141:317-323.

Miller TB Jr, Garnache A and Vicalvi JJ Jr (1981) Hormonal regulation of

hepatic glycogen synthase phosphorylase. J Biol Chem 256:2851-2855.

Minneman KP, Dibner MD, Wolfe BB and Molinoff PB (1979)  $\beta_1$ - and  $\beta_2$ -Adrenergic receptors in rat cerebral cortex are independently regulated. Science 204:866-868.

Moncany ML and Plas C (1980) Interaction of glucagon and epinephrine in the regulation of adenosine 3',5'-monophosphate-dependent glycogenolysis in the cultured fetal hepatocyte. Endocrinology 107:1667-1675.

Mondon CE and Burton SD (1971) Factors modifying carbohydrate metabolism and effect of insulin in perfused rat liver. Am J Physiol 220:724-734.

Morgan NG, Shuman EA, Exton JH and Blackmore PF (1982) Stimulation of hepatic glycogenolysis by  $\alpha_1^-$  and  $\beta_2^-$ adrenergic agonists. J Biol Chem 257:13907-13910.

Morley CGD, Kuku S, Rubenstein AG and Boyer JL (1975) Serum hormone levels following partial hepatectomy in the rat. Biochem Biophys Res Commun 67:653-656.

Munnich A, Geynet P, Schmelck <u>P-H and</u> Hanoune J (1981) Rat liver beta-adrenergic receptors: identification and characterization with [<sup>3</sup>H]DHA Horm Metab Res 13:18-21.

Murphy E, Coll K, Rich TL and Williamson JR (1980) Hormonal effects on calcium homeostosis in isolated hepatocytes. J Biol Chem 255:6600-6608.
Nakashima M, Tsuru H and Shigei T (1973) Stimulant action of methoxamine in the isolated atria of normal and 6-propyl-2-thiouracil-fed rats. Japan J Pharmacol 23:307-312.

Neville DN (1968) Isolation of <u>an</u> organ specific protein antigen from cell-surface membrane of rat liver. Biochem Biophys ta 154:540-552.
Nimmo HG and Cohen P (1977) Hormonal control of protein phosphorylation.
Adv Cyć Nuc Res 8:145-266.

Noguchi A and Whitsett JA (1983) Ontogeny of  $\alpha_1$ -adrenergic receptors in the rat heart myocardium: effects of hypothyroidism. Eur J Pharmacol 86:43-50.

Ohisalo JJ and Stouffer JE (1979) Adenosine, thyroid status and regulation of lipolysis. Biochem J 187:249-251.

- Okajima F and Ui M (1978) Adrenergic modulation of insulin secretion in vivo dependent on thyroid states. Am J Physiol 234:E106-E111.
- Okajima F and Ui M (1979) Metabolism of glucose in hyper- and hypo-thyroid rats in vivo. Biochem J 182:565-575.
- Okajima F and Ui M (1982) Conversion of adrenergic regulation of glycogen phosphorylase and synthase from an  $\alpha$ -to a  $\beta$  type during primary culture of rat hepatocytes. Arch Biochem Biophys 213:658-668.
- Okamura N and Terayama H (1976) Comparison of the epinephrine-mediated activation of adenylate cyclase in plasma membranes from liver and ascites hepatomas of rats. Biochem Biophys Acta 455:297-314.
- Okamura N and Terayama H (1978) Change of coupling system of receptor-adenylate cyclase induced by epinephrine and GTP in plasma membranes of rat liver. Biochem Biophys Acta 544:113-127.
- Oppenheimer JH, Coulombe P, Gutfeld N and Schwartz HL (1978) Nonlinear (amplified) relationship between nuclear occupancy by triiodothyronine and the appearance rate of hepatic alpha-glycerphosphate dehydrogenase and malic enzyme in the rat. J Clin Invest 61:987-997.
- Osborn D (1975) Comparison of the effects of selective  $\alpha$  and  $\beta$ -receptor agonists on intracellular cyclic AMP levels and glycogen phosphorylase activity in guinea pig liver. Br J Pharmacol 55:286-287P.
- Osborn D (1978) The alpha adrenergic receptor mediates increase in guinea-pig liver glycogenolysis. Biochem Pharm 27:1315-1320.
- Payne ME and Soderling TR (1980) Calmodulin-dependent glycogen synthase kinase. J Biol Chem 255:8054-8056.
- Pilkis SJ, El-Maghrabi MR, Pilkis J, Claus TH, and Cumming DA (1981a) Fructose 2,6-biphosphate. A new activator of phosphofructokinase. J Biol Chem 256:3171-3174.

Pilkis SJ, El-Maghrabi MR, Pilkis J,and Claus TH (1981b) Inhibition of fructose-1,6-bisphosphotase by fructose 2,6-bisphosphate. J Biol Chem 256:3619-3622.

(+

Ð

- Pilkis SJ and Park CR (1974) Mechanism of action of insulin. Ann Rev Pharmacol 14:365-388.
- Poggioli J, Berthon B and Claret M (1980) Calcium movements in in situ mitochondria following activation of α-adrenergic receptors in rat liver cells. FEBS Lett 115:243-246.
- Pointer RH, Butcher FR and Fain JN (1976) Studies on the role of cyclic guanosine 3':5'-monophosphate and extracellular  $Ca^{2+}$  in the regulation of glycogenolysis in rat liver cells. J Biol Chem 251:2987-2992.
- Popovic. WJ, Brown JE and Adamson JW (1979) Modulation of in vitro erythropoiesis. Studies with euthyroid and hypothyroid dogs. J Clin Invest 64:56-61.
- Preiksaitis HG and Kunos G (1979) Adrenoceptor-mediated activation of liver glycogen phosphorylase: effects of thyroid state. Life Sci 24:35-42.
- Preiksaitis HG, Kan WH and Kunos G (1982) Decreased α<sub>1</sub>-adrenoceptor responsiveness and density in liver cells of thyroidectomized rats. J Biol Chem 257:4321-4327.
- Proost C. Carton H and DeWulf H (1979) The α-adrenergic control of rabbit liver glycogenolysis. Biochem Pharmacol 28:2187-2191.
- Prpić V, Blackmore PF and Exton JH (1982) Phosphatidylinositol breakdown induced by vasopressin and epinephrine in hepatocytes is calcium-dependent. J Biol Chem 257:11323-11331.
- Reinhart PH, Taylor WM and Bygrave FL (1982a) Calcium ion fluxes induced by the action of  $\alpha$ -adrenergic agonists perfused rat liver. Biochem J 208:619-530.
- Reinhart PH, Taylor WM and Bygrave FL (1982b) Studies on α-adrenergicinduced respiration and glycogenolysis in perfused rat liver. J Biol Chem

257:1906-1912.

- Rizza RA, Cryer PE, Haymond MW and Gerich JE (1980) Adrenergic mechanisms of catecholamine action on glucose homeostasis in man. Metabolism 29:suppl.1, 1155-1163.
- Robison GA, Butcher RW and Sutherland EW (1971) Cyclic AMP. Academic Press, New York.
- Rognstad R and Katz J (1977) Role of pyruvate kinase in the regulation of gluconeogenesis from L-lactate. J Biol Chem 252:1831-1833.
- Rosenquist U (1978) Thyroidectomy on intracellular amount and distribution of exchangeable Ca<sup>2+</sup> Mol Cell Endocrinol 12:111-117.
- Rosenqvist U, Efendic S, Jereb B and Ostman J (1971) Influence of thyroid state on lipolysis in human adipose tissue in vitro. Acta Med Scand (189:381-384.
- Ross EM and Gilman AG (1980) Biochemical properties of hormone-sensitive adenylate cyclase. Any Rev Biochem 49:533-564.
- Samuels HH, Perlman AJ, Raaka BM and Stanley F (1982) Organization of the thyroid hormone receptor in chromatin. Rec Prog Hormone Res 38:557-600.
  Scatchard G (1949) The attractions of proteins for small molecules and ions.

Ann NY Acad Sci 51:660-672.

Schanche J-S, Ogreid D, Doskeland SO, Refsnes M, Sand TE, Ueland PM and Christofferson T (1982). Evidence against a requirement for phospholipid methylation in adenylate cyclase activation by hormones. FEBS Lett 138:167-172.

Schmelck P-H, Billan M-C, Munnich A, Genet P, Houssin D and Hanoune J (1979) The effect of common bile duct ligation upon the rat liver  $\beta$ -adrenergic receptor-adenylate cyclase system. FEBS Lett 107:259-263. V Seydoux J, Brunsman MJA, Jeanreneaud B and Girardier L (1979).  $\alpha$ -Sympathetic control of glucose output of mouse liver perfused in situ. Am J Physiol 236:E323-E327. Sharma VK and Banerjee SP (1978)  $\alpha$ -Adrenergic receptor in rat heart. Effects of thyroidectomy. J Biol Chem 253:5277-5279.

- Sherline P, Lynch A and Glinsman WH (1972) Cyclic AMP and adrenergic receptor control of rat liver glycogen metabolism. Endocrinology 91:680-690. Shimazu T (1967) Glycogen synthetase activity in liver: regulation by the autonomic nerves. Science 156:1256-1257.
- Shimazu T (1968) Glycogen synthase activity in liver: regulation by the autonomic nerves. Science 156:1256-1257.
- Shimazu T (1971) Regulation of glycogen metabolism in liver, by the autonomic nervous system. V. Activation of glycogen synthetase by vagal stimulation. Biochim Biophys Acta 252:28-38.
- Shimazu T and Amakawa A (1968a) Regulation of glycogen metabolism in liver by the autonomic nervous system. II. Neural control of glycogenolytic enzymes. Biochim Biophys Acta 164:335-348.
- Shimazu T and Amakawa A (1968b) Regulation of glycogen metabolism in liver by the autonomic nervous system. III. Differential effects of sympathetic nerve stimulation and of catecholamines on liver phosphorylase. Biochim Biophys Acta 165:349-356.
- Shimazu T and Amakawa A (1975) Regulation of glucogen metabolism in liver by the autonomic nervous system. VI. Possible mechanism of phosphorylase activation by the splanchnic nerve. Biochim Biophys Acta 385:242-256.
- Shimazu T and Fujimoto T (1971) Regulation of glycogen metabolism in liver by the autonomic nervous system. IV. Neural control of glycogen biosynthesis. Biochim Biophys Acta

252:18-27.

Shimazu T and Ogasawara S (1975) Effects of hypothalamic stimulation of gluconecgenesis and glycolysis in rat liver. Am J Physiol 228:1787-1793.
Siegel E and Tobias CA (1966) Actions of the thyroid hormones on cultured human cells. Nature 212:1318-1321.

Soderling TR, Srivastava AK, Bass MA and Khatra BS (1979) Phosphorylation and inactivation of glycogen synthase by phosphorylase kinase. Proc Nat Acad Sci USA 76:2536-2540.

- Soderling TR and Park CR (1974) Recent advances in glycogen metabolism. Adv Cyc Nuc Res 4:283-333.
- Sokal RR and Rohlf FJ (1969) Biometry: the Principles and Practice of Statistics in Biological Research. WH Freedman and Co., San Francisco.
  Soskin S (1940) The liver and carbohydrate metabolism. Endocrinology 26:279-308.
- Sourkes TL, Missala K, Bastomsky CH and Fang TY (1977) Metabolism of monoamines and diamines in hyperthyroid and hypothyroid rats. Can J Biochem 55:789-795.
- Spaulding SW and Noth (1975) Thyroid-catecholamine interactions. Med Clin N Am 59:1123-1131.
- Sperling MA, Ganguli S, Voina S, Kaptein E and Nicoloff JT (1980) Modulation by thyroid status of the glucagon receptor adenylate cyclase system in rat liver plasma membranes. Endocrinology 107:684-690.
- Stalmans W and Hers H-G (1975) The stimulation of liver phosphorylase b by AMP, fluoride and sulphate. A technical note on the specific determination of the a and b forms of liver glycogen phosphorylase. Eur J Biochem 54:341-350.
- Sterling K (1979) Thyroid hormone action at the cell level. N Eng J Med 300:117-123, 173-177.
- Stubbs M, Kirk CJ and Hems DA (1976) Role of extracellular calcium in the action of vasopressin on hepatic glycogenolysis. FEBS Lett 69:199-202

Studer RK and Borle AB (1982) Differences between male and female rats in the regulation of hepatic glycogenolysis. J Biol Chem 257:7987-7993.

Takahashi'T and Suzuki M (1975) Influence of thyroid hormone on glycogen metabolism in rat liver. Endocrinology Japan 22:187-194.

184 ·

- Thomas JA, Schlender KK and Larner J (1968) A rapid filter paper assay for UDPglucose-glycogen glucosyltransferase, including an improved biosynthesis of UDP-<sup>14</sup>C-glucose. Anal Biochem 25:486-499.
- Tolbert MEM, Butcher FR and Fain JN' (1973) Lack of correlation between catecholamine effects on cyclic adenosine 3':5'-monophosphate and gluconeogenesis in isolated rat liver cells. J Biol Chem 248:5686-5692.
- Tolbert MEM, White AC, Aspry K, Cutts J and Fain JN (1980) Stimulation by vasopressin and  $\alpha$ -catecholamines, of liver phosphatidylinositol formation in isolated rat liver parenchymal cells. J Biol Chem 255:1938-1944.
- Tovey KC, Oldham KG and Whelan JAM (1974) A simple direct assay for cyclic AMP in plasma and other biological samples using and improved competative protein binding technique. Clinica Chimica Acta 56:221-234.
- Tsai JS, and Chen A (1978) Effect of L-triiodothyronine on [<sup>3</sup>,H] DHA binding and cyclic AMP response to (-)adrenaline in cultured heart cells. Nature 275:138-140.
- Tse J, Wrenn RW and Kuo JF (1980) Thyroxine-induced changes in characteristics and activities of β-adrenergic receptors and adenosine 3',5'-monophosphate and guanosine 3',5'-monophosphate systems in heart may be related to reputed catecholamine supersensitivity in hyperthyroidism. Endocrinology 107:6-16.
- U'Prichard DC (1981) Direct binding studies of α-adrenoceptors. In: Adrenoceptors and Catecholamine action, Part A. Ed: G Kunos, John Wiley and Sons. Toronto. chpt 5, pp 131-180.
- U'Prichard DC, Charness ME, Robertson D and Snyder SH (1978) Prazosin: differential affinities for two populations of  $\alpha$ -noradrenergic receptor binding sites. Eur J pharmacol 50:87-89.
- U'Prichard DC, Greeberg DA and Snyder SII (1977) Binding characteristics of a radiolabeled agonist and antagonist at central nervous system alpha 'noradrenergic receptors. Mol<sup>1</sup> Pharmacol 13:454-473.

Van de Werve G, Hue L and Hers H-G (1977) Hormonal and ionic control of the glycogenolytic cascade in rat liver. Biochem J 162:135-142.

- Van Inwegen RG, Robison GA, Thompson WJ, Armstrong KJ and Stouffer JE (1975) Cyclic nucleotide phosphodiesterases and thyroid hormones. J Biol Chem 250:2452-2456.
- Van Schaftingen E, Hue L and HErs HG (1980) Fructose 2,6-bisphosphate, the probable structure of the glucose- and glucagon - sensitive stimulation of phosphofructokinase. Biochem J 192:897-901.
  - Vandenheede JR, DeWulf H and Merlevede W (1979) Liver phosphorylase b kinase. Cyclic AMP-mediated activation and properties of the partially purified rat liver-enzyme. Eur J Biochem 101:51-58.
  - Vauquelin G, Lacombe ML, Guellaen G, Strosberg AD and Hanoune J (1976)
     Tazolol (l-isopropylamino-3-(2-thiazoxy))- 2-propranolol as a β-adrenergic
     blocker. Biochem Pharmacol 25:2605-2608.
  - Wagner J and Schumann H-J (1979)<sup>c</sup> Differential mechanisms underlying the stimulation of myocardial  $\alpha$  and  $\beta$ -adrenoceptors. Life Sci 24:2045-2052.
  - Waldstein SS (1966) Thyroid-catecholamine interrelations. Ann Rev Med 17:123-132.
- Walsh DA (1978) Role of the cAMP-dependent protein kinase as the transducer of cAMP action. Biochem Pharmacol 27:1801-1804.
- Weiner N (1980) Drugs that inhibit adrenergic nerves and block adrenergic receptors. In: Goodman and Gilman's The Pharmacolgical Basis of Therapeutics, Sixth Edition. Ed: AG Gilman, LS Goodman and A Gilman. McMillan Publishing Co., Inc. New York. chpt 9, pp 176-210.
- Weiss SJ and Putney JW Jr (1978) Does calcium mediate the increase in potassium permeability due to phenylephrine or angiotensin II in the liver. J Pharmacol Exp Ther 207:669-676.
- Whiting JA and Barritt GJ (1982) On the mechanism by which hormones induce the release of  $Ca^{2+}$  from mitochondria in the liver cell. Biochem J

206:121-129.

- Whitton PD, Rodrigues LM and Hems DA (1978) Stimulation by vasopressin, angiotensin and oxytocin of gluconeogenesis in hepatocyte suspensions. Bioghem (J 176:893-898.
- Williams LT and Lefkowitz RJ (1978) Receptor Binding Studies in Adrenergic Pharmacology. Raven Press, New York.
- Williams RS, Guthrow CE and Lefkowitz RJ (1979) β-Adrenergic receptors of human lymphocytes are unaltered by hyperthyroidism. J Clin Endocrinol Metab 48:503-505.
- Williams RS and Leftkowitz RJ (1979) Thyroid hormone regulation of alpha-adrenergic receptors: studies in rat myocardium. J Cardiovasc Pharmacol 1:181-189.
- Wolfe BB, Harden TK and Molinoff PB (1976) β-Adrenergic receptors in rat
  liver: effects of adrenalectomy. Proc Natl Acad Sci USA 73:1343-1347.
  Wood CL, Caron MG and Lefkowitz RJ (1979) Separation of solubilized alpha and beta adrenergic receptors by affinity chromatography. Biochem Biophys Res Commun 88:1-8.
- Wurtman RJ, Kopin IJ and Axelrod J (1963) Thyroid function and the cardiac disposition of catecholamines. Endocrinology 73:63-74.
- Yamaguchi I and Kopin IJ (1980) Differential inhibition of alpha-l and alpha-2 adrenoceptor-mediated pressor responses in pithed rats. J Pharmacol Exp Ther 214:275-281.
- Yamaguchi N (1982) Evidence supporting the existence of presynaptic a-adrenoceptors in the regulation of endogenous noradrenaline release upon hepatic sympathetic nerve stimulation in the dog diver in vivo. Naunyn-Schmiedeberg's Arch Pharmacol 321:177-184.
- Youdim MBH and Holzbauer M (1976) Physiological and pathological in tissue monoamine oxidase activity. J Neural Transm 38:193-229.

Young BA and McNeill JH (1974) The effect of noradrenaline and tyramine on

cardiac contractility, cyclic AMP and phosphorylase a in normal and <sup>a</sup>hyperthyroid rats. Can J Physiol Pharmacol 52:375-383.

Young JB and Landsberg L (1977) Catecholamines and intermediary metabolism. <sup>°</sup> Clin Endicrinol Metab 6:599-631.

Zimon RP, Flock EV, Tyce GM, Sheps SG and Owen CA Jr (1967) Effects of thyroid hormones on metabolism of d, I-noradrenaline by isolated rat liver. Endocrin°ology 80:808-814.